

## Using Affine GARCH Models in Portfolio Optimization

Ben Spies

Vollständiger Abdruck der von der TUM School of Computation, Information and Technology der Technischen Universität München zur Erlangung eines

Doktors der Naturwissenschaften (Dr. rer. nat.)

genehmigten Dissertation.

Vorsitz: Prof. Dr. Andreas Wiese

Prüfende der Dissertation:

- 1. Prof. Dr. Rudi Zagst
- 2. Prof. Dr. Marcos Escobar-Anel
- 3. Prof. Dr. Gaurav Khemka

Die Dissertation wurde am 21. Dezember 2023 bei der Technischen Universität München eingereicht und durch die TUM School of Computation, Information and Technology am 4. März 2024 angenommen.

## Abstract

This dissertation studies different aspects of a dynamic portfolio allocation problem in a discretetime environment where the underlying price process of the single risky asset follows the dynamics of an affine GARCH model. While prior to the work contained in this dissertation, a closed-form solution was known only in the case of a simple Gaussian GARCH model and an investor maximizing utility from terminal wealth via a constant relative risk aversion (CRRA) utility function, we extend this approach in three different directions. Firstly, we modify the setup to allow for consumption at intermediate time points before the end of the investment period. In a second step, we generalize the result for utility from terminal wealth to the class of hyperbolic absolute risk aversion (HARA) utility functions and relax an assumption on the investor's risk preferences. Finally, the model complexity is increased towards Lévy GARCH models, which, besides a Gaussian innovation, include an additional jump component. The optimal solution is available in approximate closed form in all these settings, enabling practitioners to implement the corresponding strategies efficiently. In the realm of discrete-time models, once renowned for the challenge of obtaining convenient solutions in the context of financial applications, this dissertation thus makes valuable contributions by addressing broadly formulated problems. We also investigate various facets of our topic numerically, discovering that accommodating heteroscedasticity in the underlying asset price model is highly beneficial for medium-term investors. In a study based on S&P 500 index data, a comparison of an investor implementing the strategy of a Gaussian GARCH model and another one following a homoscedastic variant shows a significant difference in the performance related to both consumption at intermediate time points and terminal wealth, in favor of the GARCH investor. Similarly, we use the generalizations with respect to HARA utilities and the investor's risk preferences to calculate the efficient frontier for an investor in the Gaussian GARCH model as well as in the homoscedastic variant and find that, given the same expected portfolio return, the heteroscedastic investor is exposed to significantly less portfolio variance. On the other hand, a comparison of a Gaussian and a non-Gaussian GARCH model within the framework of consumption and investment suggests that an extension to the latter model, which incorporates conditional skewness and excess kurtosis of log returns, has a relatively small impact. A different type of analysis in the context of Lévy GARCH models with jumps, based on the concept of wealth-equivalent loss (WEL), shows that correctly calibrated jump-free GARCH models can mimic very well the strategies suggested by models incorporating jumps.

# Acknowledgements

I express my deepest gratitude to my supervisor, Rudi Zagst. He accepted me as a mentee within the TopMath program during a challenging phase and provided invaluable support throughout the process of working on my dissertation topic. This support includes funding for my doctorate and various related activities such as conferences and research stays, which he consistently encouraged me to pursue. Beyond the financial support, his contributions are exemplified by his commitment, often dedicating late evenings and early mornings to review portions of this dissertation. Without him, this work would not have been possible, and I am incredibly thankful for his guidance and support.

The same heartfelt appreciation extends to Marcos Escobar-Anel, who has been an integral part of the project from the very beginning. Marcos provided incredible support at every stage of the process, from idea development to the publication of the final papers. Additionally, I would like to thank Marcos for his incredible hospitality; my research stays at Western University will be remembered fondly, and I truly appreciate the time Marcos and his family spent with me.

A special thank you also goes to Katja Kröss and Carl-Friedrich Kreiner, who, as coordinators of the TopMath program, have supported me with valuable advice.

The everyday work at the Technical University of Munich has been a source of joy, thanks to my colleagues in the Research Group Financial and Actuarial Mathematics. I am very grateful for the opportunities to discuss mathematical topics and many other unrelated subjects, for the research seminars, for the talks and the feedback, and, of course, for all the social activities. A special thanks goes to Michel Kschonnek and Dominik de Witte, who have enriched my life both within and beyond the university grounds.

I want to express my sincere gratitude to my parents, who have accompanied me throughout my entire journey, offering unwavering support in all aspects of my life. I am endlessly thankful for the time and effort they have invested in me, my upbringing, and my education.

Lastly, my profound thanks go to Elisa for her enduring patience and constant backing throughout my doctorate. Her presence has been a source of strength, and I am truly fortunate to have her by my side.

> Ben Spies December 2023

# Contents

Ab	stract	iii				
Ac	knowledgements	v				
Co	ontents vii st of Abbreviations ix					
Lis						
1	Introduction	1				
2	Mathematical Preliminaries         2.1       Measure and Probability         2.2       Stochastic Control Models         2.2.1       Model Formulation         2.2.2       Value Iteration         2.2.3       Concave Value Functions         2.3       Maximum-Likelihood Estimation	<b>5</b> 7 7 9 10 12				
3	Expected Utility Theory on General Affine GARCH Models         3.1       Introduction         3.2       Outline of the Approach         3.3       Solution to the Portfolio Optimization Problem         3.4       Application to IG-GARCH Model         3.5       Conclusion	<b>15</b> 16 19 21 24				
4	<b>Optimal Consumption and Investment</b> 4.1       Introduction         4.2       Problem Description         4.2.1       Self-Financing Condition         4.2.2       Affine GARCH Models and Transition Function         4.2.3       Optimization Problem         4.3       Solution to the Portfolio Optimization Problem         4.3.1       Main Results         4.3.2       Examples of Applications to Affine GARCH Models         4.4       Numerical Analysis         4.4.1       Optimal Solution Analysis         4.4.2       Data-based Performance Analysis	<b>25</b> 25 27 28 29 31 31 34 36 36 40 43				
5	HARA Utilities and Mean-Variance Optimization         5.1       Introduction         5.2       Formulation of the EUT Problem         5.2.1       Taylor Approach         5.2.2       CPPI Approach         5.3       Optimal Investment Maximizing HARA Utility	<b>45</b> 45 47 48 49 50				

	$5.4 \\ 5.5 \\ 5.6$	Connection to Mean-Variance Theory	52 54 57
6	Expe 6.1 6.2 6.3 6.4	ected Utility Theory within Lévy GARCH ModelsIntroduction	<ul> <li><b>59</b></li> <li>61</li> <li>63</li> <li>63</li> <li>65</li> <li>67</li> <li>68</li> <li>69</li> <li>73</li> <li>75</li> <li>78</li> </ul>
7	Con	clusion	81
Bi	bliogr	aphy	83
Α	<b>App</b> A.1	endix for Chapter 2 Proof of Theorem 2.15	<b>87</b> 87
В	<b>App</b> B.1 B.2	endix for Affine GARCH Models MGF for the HN-GARCH Model	<b>89</b> 89 91
С	<b>App</b> C.1 C.2	endix for Chapter 4         Proofs from Section 4.3         Complementary Material         C.2.1         Additional Plots for Section 4.4.1: Sensitivity Analysis         C.2.2         Alternative Parametric Choice         C.2.3         Wealth-Equivalent Loss	<b>95</b> 95 104 104 105 107
D	<b>App</b> D.1	endix for Chapter 6	<b>113</b>

# List of Abbreviations

ARCH	Autoregressive Conditional Heteroscedasticity
CPPI	Constant Proportion Portfolio Insurance
CRRA	Constant Relative Risk Aversion
DP	Dynamic Programming
EUT	Expected Utility Theory
GARCH	Generalized Autoregressive Conditional Heteroscedasticity
GBM	Geometric Brownian Motion
HARA	Hyperbolic Absolute Risk Aversion
HN-GARCH	Affine Gaussian GARCH model introduced by Heston and Nandi (2000)
HS	Homoscedastic model with constant conditional variance
IG-GARCH	Inverse Gaussian GARCH model
MDP	Markovian Decision Process
MGF	Moment Generating Function
MJ	Merton Jump, referring to a possible choice for a Lévy jump innovation
MLE	Maximum Likelihood Estimate
MPR	Market Price of Risk
MV	Mean-Variance
NIG	Normal-Inverse Gaussian, referring to a possible choice for a Lévy jump
	innovation
PDF	Probability Density Function
SCM	Stochastic Control Model
$\mathrm{SCM}_{T-t}$	(T-t)-stage optimization problem of maximizing the expected reward in an
	SCM
SFC	Self-Financing Condition
VG	Variance Gamma, referring to a possible choice for a Lévy jump innovation
VIX	Volatility index, based on S&P 500 index options, calculated and published
	by the Chicago Board Options Exchange
WEL	Wealth-Equivalent Loss

# 1 Introduction

In the realm of portfolio optimization and investment decision-making, the expected utility hypothesis is a fundamental assumption on the behavior of rational agents, stating that market participants maximize their expected subjective utility. In an uncertain environment involving, e.g., stochastic processes for the asset price processes, expected utility theory (EUT) provides a framework for developing portfolio strategies and evaluating their performance. Within this framework, the problem of finding the optimal portfolio selection has many facets, consisting of various ways to measure utility and different model choices concerning the underlying price processes.

In modern financial literature, two large model streams can be identified, interpreting the asset price process either as a continuous or a discrete function of time. Mossin (1968) and Samuelson (1969) made seminal contributions to EUT in a discrete-time setting, with their work also offering insights into the abovementioned question of how to measure an agent's utility. The former approach considers an investor that may re-allocate the investments at intermediate time points, but the strategy performance is evaluated solely at the end of the investment period, called the investor's (finite) time horizon. This formulation was extended by Samuelson (1969), who included the possibility of consuming parts of the allocated wealth at each intermediate time point. In continuous time, Merton (1969, 1971) pioneered the corresponding work in terms of optimizing consumption and portfolio allocation, delivering explicit formulas for both parts of the optimal solution. Merton's generalization from the earlier to the later paper shows yet another dimension of the problem formulation: To measure utility, he used the larger class of hyperbolic absolute risk aversion (HARA) utility functions instead of the nested constant relative risk aversion (CRRA) family. As a referral to this groundbreaking work, the problem of maximizing expected utility derived from terminal wealth and consumption is also known as Merton's portfolio problem.

Concerning the underlying asset price process model, the two streams identified above divide the research in the field into two parts. In his continuous-time approach, Merton (1969) relied on a geometric Brownian motion (GBM), assuming that asset prices are log-normally distributed. More advanced ways of modeling the underlying processes were brought up later, e.g., in the form of stochastic volatility models. One of the most famous approaches in this regard was published by Heston (1993), and the corresponding portfolio problem was solved a few years later by Kraft (2005). The major advantage of continuous-time models was rooted in their analytical tractability via Itô calculus and the abovementioned availability of closed-form solutions to pricing and portfolio allocation problems. On the other hand, practitioners always have to work with discrete financial data because of physical limitations, forming an inherent need for discrete-time models that can be estimated from historical data more easily than their continuous counterparts. On the side of discrete-time models, however, the famous modeling approaches, particularly the celebrated ARCH and GARCH family introduced by Engle (1982) and Bollerslev (1986), respectively, did not lead to convenient formulas for similar applications.

This challenge was overcome many years later when Heston and Nandi (2000) presented the first member of a class that would henceforth be called the affine GARCH models – the setup by Heston and Nandi (2000) is referred to as the HN-GARCH model. The crucial and characteristic property ensures that the joint moment generating function (MGF) of the log-asset return

### 1 Introduction

and its conditional variance is exponentially affine in these two variables. Initially invented for pricing purposes, models with this property bring about the advantage that closed-form solutions are available for values of various financial derivatives. Within the world of affine GARCH models, the research focused on extending the setup in the subsequent years. Christoffersen et al. (2006) introduced a new model and used inverse Gaussian innovations instead of the normal distribution in the HN-GARCH, and the corresponding model was named IG-GARCH after this choice. A few years later, Christoffersen et al. (2012) incorporated jump components in both the asset return and its volatility process. This feature, however, led to the loss of the crucial affine structure of the MGF and thus came at a great cost. Two years later, Ornthanalai (2014) found a formulation, based on the original HN-GARCH model, that allows for an additional (Lévy) jump component in the asset price process and still preserves the affine form of the MGF – the class of Lévy GARCH models was invented. Badescu et al. (2019) finally formulated a common framework that includes the Gaussian HN-GARCH as well as the IG-GARCH as special cases – the Lévy GARCH models, however, are not included. This setup has been generalized to a multi-factor model very recently (Augustyniak et al., 2023).

As mentioned previously, affine GARCH models were mainly used for pricing upon their invention. Heston and Nandi (2000) and Christoffersen et al. (2006) considered S&P 500 index options, and Badescu et al. (2019) found closed-form expressions for prices of variance swaps. Escobar-Anel et al. (2022a) were the first to exploit the availability of the MGF in a specific affine GARCH model in the context of portfolio optimization, solving an EUT problem with the investor deriving utility from terminal wealth via a CRRA function, with the underlying asset price following HN-GARCH dynamics. Figure 1.1 illustrates the state of the art in the



Figure 1.1: This Figure displays the state of the literature before the publications related to this dissertation, with only the solution to the EUT problem for an investor with CRRA utility in the HN-GARCH model available. This figure is extended as we proceed through the chapters of this dissertation, see Figures 3.1, 4.1, 5.1, and 6.1

area of optimal investment (and consumption) within affine GARCH models at that time, with-

out the contributions to the field presented in this dissertation.<sup>1</sup> While at the intersection of the HN-GARCH model and CRRA utility from terminal wealth without any consumption at intermediate time points, the solution by Escobar-Anel et al. (2022a) existed, many connected research questions were still unanswered. This specifically involves the different problem dimensions described above and visualized in Figure 1.1:

- Is it possible to obtain closed-form solutions also allowing for consumption at intermediate time points before the end of the investment horizon?
- Can the assumption of CRRA utility from terminal wealth be relaxed and generalized to a larger class of utilities?
- Is it possible to solve the problem in a larger and more advanced class of models?

The present dissertation addresses all three of these questions, offering affirmative responses to each. Figure 1.1 will be extended successively as we move through the chapters and tackle the extensions, see Figures 3.1, 4.1, 5.1 and 6.1.

In the first step, we solved the EUT problem with a CRRA utility from terminal wealth for the class of general affine GARCH models in the formulation of Badescu et al. (2019). As specified above, this includes the non-Gaussian IG-GARCH model. Together with a numerical study about the impact of non-Gaussianity in conditional log-asset returns on optimal portfolio strategies, this work was published in Escobar-Anel et al. (2021) and accepted as the author's Master's thesis within the elite doctoral program TopMath. Since the theoretical results are fundamental for the subsequent chapters, we include the essential parts of this project in Chapter 3 in this dissertation. Chapter 3 does not contribute to the evaluation of this dissertation.

In Chapter 4, we solve a general problem including the possibility of consumption at intermediate time points while avoiding the undesirable scenario of consuming the entire wealth before the end of the time horizon. There are three types of numerical analysis connected to the theoretical contributions in Chapter 4: We first study the sensitivity of the optimal solution – referring to consumption as well as risky allocation – to important model parameters and focus on the impact of non-Gaussianity, leading to the eye-opening finding that conditional heteroscedasticity matters more to investors than non-Gaussianity of log-asset returns. In another numerical investigation, we use S&P 500 index data to show that investors following a homoscedastic instead of a Gaussian GARCH solution may face substantial losses. A complementary analysis of losses in the context of consumption on a theoretical level and results for different parameter sets can be found in Appendix C.2.

Chapter 5 targets another extension of the previous setup, focusing on utility from terminal wealth. While earlier results used CRRA utility functions to measure utility at the end of the time horizon, we prove an extension for the more general class of HARA utilities. Furthermore, we show that a restriction on the investor's risk aversion can be relaxed, making a connection to problem formulations in mean-variance (MV) theory and portfolio optimization based on higher return moments possible. Our corresponding numerical study focuses on the efficient frontier in the special case of MV optimization and shows, again based on S&P 500 index data, that an investor implementing an HN-GARCH strategy, given the same expected portfolio return, faces significantly less portfolio variance than another investor following a homoscedastic variant.

Chapter 6 extends the underlying asset price process type to Lévy GARCH models. We show that in the presence of an additional Lévy jump component, after imposing a condition on the model parameters, we can solve the corresponding EUT problem in closed form. This

<sup>&</sup>lt;sup>1</sup>The increase in along the three axes is not meant to be continuous. We will have different sections in each dimension that correspond to model features, e.g., there are only going to be the two values NO and YES on the vertical axis, see Figure 4.1.

major theoretical contribution allows for a numerical investigation of the impact of jumps on the optimal solution and strategy performance. We simulate data from several Lévy GARCH models involving jumps and ask the question whether the jump-free Gaussian HN-GARCH model, with the parameters estimated based on the simulated dataset, is capable of producing comparable results. The awakening answer is that correctly calibrated jump-free models can produce outstanding results and suffer only minimal losses with respect to the optimal jump models.

By further developing the diagram in Figure 1.1 in the corresponding chapters, we visualize the main theoretical contributions of this dissertation. In summary, our contributions are the following:

• We present the first analytical solution to a consumption and investment problem in an affine GARCH environment in Chapter 4, based on the following article, which has been published as:

M. Escobar-Anel et al. (2024b). "Optimal consumption and investment in general affine GARCH models." In: OR Spectrum. DOI: 10.1007/s00291-024-00749-z

• In Chapter 5, we extend existing results for the EUT problem to the class of HARA functions to measure utility from terminal wealth and consider previously excluded ranges for the investor's risk aversion, establishing a connection to MV optimization. A short version of this chapter is published in:

M. Escobar-Anel et al. (2024a). "Mean–variance optimization under affine GARCH: A utility-based solution." In: *Finance Research Letters* 59.104749. DOI: 10.1016/j.frl.2023.104749

• We introduce a closed-form solution to the optimal investment problem in a Lévy GARCH environment in Chapter 6, allowing for jumps of different types in the asset price process. This chapter is based on the following article:

M. Escobar-Anel et al. (2023). "Do Jumps Matter in Discrete-Time Portfolio Optimization?" Working paper submitted for publication

Chapter 7 concludes this dissertation, providing a concise summary of the main findings and contributions, and presenting prospects for potential future research.

## 2 Mathematical Preliminaries

## 2.1 Measure and Probability

The portfolio optimization problems in this dissertation are based on discrete-time financial market models and stochastic control. This section introduces basic notations for measurability and probability spaces to adequately describe the corresponding environments. All of the following results can be found in standard textbooks. Particularly concerning the part relevant in the context of stochastic control models, we refer to Hinderer et al. (2016, Appendix B).

**Definition 2.1** ( $\sigma$ -algebra). A  $\sigma$ -algebra  $\mathcal{F}$  on a non-empty set  $\Omega$  is a system of subsets of  $\Omega$  such that

- (i)  $\Omega \in \mathcal{F}$ ,
- (ii)  $G \in \mathcal{F}$  implies  $\Omega \setminus G \in \mathcal{F}$ ,
- (iii) The countable union of sets in  $\mathcal{F}$  still belongs to  $\mathcal{F}$ .

The pair  $(\Omega, \mathcal{F})$  is called a measurable space, and the sets in  $\mathcal{F}$  are said to be  $(\mathcal{F})$ -measurable.

For a non-empty system C of subsets of  $\Omega$ , there exists a smallest  $\sigma$ -algebra  $\sigma(C)$  that contains C. We say that  $\sigma(C)$  is generated by C and call C the generator.

**Definition 2.2** (Trace of a  $\sigma$ -algebra). For a measurable space  $(\Omega, \mathcal{F})$  and a non-empty set  $B \in \mathcal{F}$ , we call

$$B \cap \mathcal{F} \coloneqq \{B \cap G \mid G \in \mathcal{F}\} = \{G \in \mathcal{F} \mid G \subseteq B\}$$

the trace  $\sigma$ -algebra of  $\mathcal{F}$  on B.

In our applications, the underlying set usually is the set of real numbers  $\mathbb{R}$ . On this set, and in general, on all metric spaces, the most natural  $\sigma$ -algebra is the one generated by the open sets, i.e., by the generator  $\{(-\infty, \alpha) \mid \alpha \in \mathbb{R}\}$ .

**Definition 2.3** (Product  $\sigma$ -algebra). For systems  $C_i$  of subsets of a set  $\Omega_i$ , for i = 1, ..., d, we define the Cartesian product as  $\times_1^d C_i := \{ \times_1^d B_i \mid B_i \in C_i, 1 \le i \le d \}$ . For measurable spaces  $(\Omega_i, \mathcal{F}_i), i = 1, ..., d$ , the product  $\sigma$ -algebra of the  $\sigma$ -algebras  $\mathcal{F}_i$  is defined as  $\otimes_1^d \mathcal{F}_i := \sigma (\times_1^d \mathcal{F}_i)$ on  $\Omega := \times_1^d \Omega_i$ .

**Definition 2.4** (Measurable mapping). Consider a mapping f from a non-empty set  $\Omega$  into a measurable space  $(\Omega', \mathcal{F}')$ . The system of preimages  $f^{-1}(\mathcal{F}') \coloneqq \{f^{-1}(G') \mid G' \in \mathcal{F}'\}$  is a  $\sigma$ algebra on  $\Omega$ . For a  $\sigma$ -algebra  $\mathcal{F}$  on  $\Omega$ , the mapping f is called  $\mathcal{F}$ - $\mathcal{F}'$ -measurable if  $f^{-1}(\mathcal{F}') \subseteq \mathcal{F}$ . An  $\mathcal{F}$ - $\mathcal{F}'$ -measurable mapping is called a random variable on  $(\Omega, \mathcal{F})$  with values in  $(\Omega', \mathcal{F}')$ . If the use of the corresponding  $\sigma$ -algebras is clear from the context, we will simply call f measurable. A collection of random variables  $\{X_i\}_{i=1,\dots,d}$  generates a  $\sigma$ -algebra via  $\sigma\left(\bigcup_{i=1}^d \sigma(X_i)\right)$ , where  $\sigma(X_i)$  is the  $\sigma$ -algebra generated by the preimages of  $X_i$ .

**Definition 2.5** (Probability space). On a measurable space  $(\Omega, \mathcal{F})$ , we will call a mapping  $\mathbb{P} : \mathcal{F} \to [0, \infty)$  a probability measure if it satisfies the following conditions:

- (i)  $\mathbb{P}(\Omega) = 1$ ,
- (ii)  $\mathbb{P}$  is countably additive, i.e.,  $\mathbb{P}(\emptyset) = 0$  and for a collection of pairwise disjoint sets  $A_1, A_2, \ldots \in \mathcal{F}, \mathbb{P}$  satisfies

$$\mathbb{P}\left(\bigcup_{i=1}^{\infty} A_i\right) = \sum_{i=1}^{\infty} \mathbb{P}\left(A_i\right).$$

The triplet  $(\Omega, \mathcal{F}, \mathbb{P})$  forms a probability space.

In financial applications, market participants gain more and more information over time by observing stock and option prices. This kind of evolution is modeled via filtrations and filtered probability spaces.

**Definition 2.6** (Filtered probability space). A collection  $\{\mathcal{F}_i\}_{i=1,...,d}$  of  $\sigma$ -algebras that satisfy  $\mathcal{F}_i \subseteq \mathcal{F}$  for all i and  $\mathcal{F}_k \subseteq \mathcal{F}_l$  for  $1 \leq k \leq l \leq d$  is called a filtration. The quadruple  $(\Omega, \mathcal{F}, \{\mathcal{F}_i\}_i, \mathbb{P})$  is called a filtered probability space.

In the sequel, we will consider real-valued random variables on a probability space  $(\Omega, \mathcal{F}, \mathbb{P})$ , i.e., the random variables are mappings into the real numbers  $\mathbb{R}$ .

**Definition 2.7** (Expectation). Let  $(\Omega, \mathcal{F}, \mathbb{P})$  be a probability space and X a real-valued random variable. The expectation of X, denoted by  $\mathbb{E}[X]$  is defined as

$$\mathbb{E}[X] = \int_{\Omega} X(\omega) \, d\mathbb{P}(\omega),$$

where the integral is interpreted as the Lebesgue integral.

A random variable X is said to be (absolutely) integrable if  $\mathbb{E}[|X|]$  is finite. The expectation is also referred to as the first moment of a random variable. Higher moments are defined below.

**Definition 2.8** (Higher moments). Assuming that X is a random variable on  $(\Omega, \mathcal{F}, \mathbb{P})$  such that the expressions below are well-defined, the variance, skewness, and kurtosis, respectively, of X, are given by:

$$Var [X] \coloneqq \sigma^{2} \coloneqq \mathbb{E} \left[ (X - \mathbb{E} [X])^{2} \right],$$
$$\mathcal{S} [X] \coloneqq \mathbb{E} \left[ \left( \frac{X - \mathbb{E} [X]}{\sigma} \right)^{3} \right],$$
$$\mathcal{K} [X] \coloneqq \mathbb{E} \left[ \left( \frac{X - \mathbb{E} [X]}{\sigma} \right)^{4} \right].$$

**Definition 2.9** (Conditional expectation). For an integrable random variable X on a probability space  $(\Omega, \mathcal{F}, \mathbb{P})$ , we define the conditional expectation  $\mathbb{E}[X \mid \mathcal{G}]$  of X given  $\mathcal{G}$ , where  $\mathcal{G} \subseteq \mathcal{F}$  is a  $\sigma$ -algebra, as the random variable Y such that

- (i) Y is G-measurable,
- (*ii*)  $\mathbb{E}[|Y|] < \infty$ ,
- (iii)  $\mathbb{E}[\mathbb{1}_C Y] = \mathbb{E}[\mathbb{1}_C X]$  for every  $C \in \mathcal{G}$ ,

with  $\mathbb{1}_C(\omega)$  the indicator function of C yielding 1 if and only if  $\omega \in C$  and 0 otherwise.

For random variables X and Y on a probability space  $(\Omega, \mathcal{F}, \mathbb{P})$ , we say that X = Y holds  $\mathbb{P}$ -almost surely if

$$\mathbb{P}\left(\left\{\omega \in \Omega \mid X(\omega) = Y(\omega)\right\}\right) = 1.$$

The following properties of conditional expectation will be helpful throughout this dissertation. the proofs are ommitted and can be found in many standard textbooks.

**Proposition 2.10** (Properties of conditional expectation, cf. Bingham and Kiesel (2004), Proposition 2.5.1). Assume that X and Y are integrable random variables on a probability space  $(\Omega, \mathcal{F}, \mathbb{P})$ , and assume that  $\mathcal{G}, \mathcal{H} \subseteq \mathcal{F}$  are  $\sigma$ -algebras.

- (a)  $\mathbb{E}[X \mid \{\emptyset, \Omega\}] = \mathbb{E}[X].$
- (b) If X is  $\mathcal{G}$ -measurable, then  $\mathbb{E}[X \mid \mathcal{G}] = X$ ,  $\mathbb{P}$ -almost surely.
- (c) Given constants  $a_1, a_2 \in \mathbb{R}$ ,  $\mathbb{E}[a_1X + a_2Y \mid \mathcal{G}] = a_1\mathbb{E}[X \mid \mathcal{G}] + a_2\mathbb{E}[Y \mid \mathcal{G}]$ ,  $\mathbb{P}$ -almost surely.
- (d) If Y is  $\mathcal{G}$ -measurable, then  $\mathbb{E}[YX \mid \mathcal{G}] = Y\mathbb{E}[X \mid \mathcal{G}]$ ,  $\mathbb{P}$ -almost surely.
- (e) If  $\mathcal{H} \subseteq \mathcal{G}$ , then  $\mathbb{E}\left[\mathbb{E}\left[X \mid \mathcal{G}\right] \mid \mathcal{H}\right] = \mathbb{E}\left[\mathbb{E}\left[X \mid \mathcal{H}\right] \mid \mathcal{G}\right] = \mathbb{E}\left[X \mid \mathcal{H}\right]$ ,  $\mathbb{P}$ -almost surely.

## 2.2 Stochastic Control Models

To solve this dissertation's discrete-time portfolio optimization problems, we first introduce the basic notation and some relevant results concerning dynamic programming and stochastic control models. This section is based on the comprehensive book by Hinderer et al. (2016) on various types of control models and decision processes.

#### 2.2.1 Model Formulation

An investor in a typical setting considered in this dissertation starts at some time  $t \ge 0$  with a certain amount of initial wealth. At this point, the investor has to decide how to allocate the available resources, choosing from the different assets available in the market, which in our problems consist of a riskless bank account and a risky asset following the dynamics of an affine GARCH model. According to this decision, and based on the random evolution of the asset price, the system then moves on to time t + 1, characterized by a new value for the investor's wealth. This procedure continues to a predefined finite time horizon T > t.

For simplicity, we usually use t = 0 as the initial time point. In terms of a stochastic control model (SCM), we speak of an initial state  $s_0 \in S$ , where S denotes the space of all possible

states, and, together with the  $\sigma$ -algebra  $\mathcal{S}$ , forms a measurable space  $(\mathbb{S}, \mathcal{S})$ . The investor's allocation choice is taken from a space of actions  $\mathbb{A}$ , endowed with the  $\sigma$ -algebra  $\mathcal{A}$ . The so-called constraint set  $\mathbb{D}$  contains all possible combinations of states and actions and is part of the product  $\sigma$ -algebra  $\mathcal{S} \otimes \mathcal{A}$ , equipped with the trace  $\sigma$ -algebra  $\mathcal{D}$  of  $\mathbb{D}$  in  $\mathcal{S} \otimes \mathcal{A}$ . For a specific  $s \in \mathbb{S}$ , we let  $D(s) \coloneqq \{a \in \mathbb{A} \mid (s, a) \in \mathbb{D}\}$ .

While in a simpler form of a control model, the transition from one state to the next is fully deterministic and given via a mapping from  $\mathbb{D}$  to  $\mathbb{S}$ , this does not apply to our investment problem. Instead, we let  $(\mathbb{Y}, \mathcal{Y})$  be another measurable space, where  $\mathbb{Y} \neq \emptyset$  is the space of so-called disturbances, in financial market models often named *innovations*. The disturbances are commonly assumed to be i.i.d., but in our applications, the distribution need not be the same at all time points – that is, we work with a slight relaxation of the mentioned assumption. The transition to the next state is now given by a measurable mapping  $\mathbb{T}: \mathbb{D} \times \mathbb{Y} \to \mathbb{S}$ , called the transition function. The combination of a disturbance distribution and a transition function can also be considered as a direct transition probability from  $\mathbb D$  into  $\mathbb S$ . The central feature of this model is the fact that, even after the choice of an action at time point t, the state one time step ahead is still random and subject to the realization of the disturbance. The random state at time t+1 is therefore denoted by  $S_{t+1}$ , its realization is  $s_{t+1}$ , and similarly  $Y_t$  denotes a random variable with values in  $(\mathbb{Y}, \mathcal{Y})$ , while  $y_t$  is its realization. Furthermore, a decision rule  $\pi_t$  at time  $t = 0, \ldots, T - 1$  is a measurable mapping from S into A such that  $(s_t, \pi_t(s_t)) \in \mathbb{D}$ , i.e., a decision rule thus yields an action dependent on the realization of the state. A collection of decision rules  $\pi = {\pi_t}_{t=0}^{T-1}$  is called *policy*, and we denote the set of all policies by  $\mathbb{F}$ . In the presence of a policy  $\pi$ , the random state  $S_t$  at some time point  $t = 1, \ldots, T$  usually depends on  $\pi$ , but we omit this dependence for readability.

To properly define the transitions between the different states, we need the following formal definition of a transition probability:

**Definition 2.11** (Transition probability, cf. Hinderer et al., 2016, Def. 16.1.1). A function  $\mathbb{P}: \mathbb{D} \times S \to \mathbb{R}_+$  is called a transition probability from  $\mathbb{D}$  into  $\mathbb{S}$  if

- $(s,a) \mapsto \mathbb{P}(s,a,B)$  is  $\mathcal{D}$ -measurable for all  $B \in \mathcal{S}, (s,a) \in \mathbb{D}$ .
- $B \mapsto \mathbb{P}(s, a, B)$  is a probability measure on S for all  $(s, a) \in \mathbb{D}$ .

Given a decision rule  $\pi$  and using the fact that the composition of measurable mappings is again measurable, we can consider transition probabilities from S into S via

$$s \mapsto \mathbb{P}_{\pi}(s, B) \coloneqq \mathbb{P}(s, \pi(s), B).$$

The performance of the investor's strategy is evaluated by a real-valued reward function  $\phi_T$  at the end of the investment horizon, i.e., at time T, and in addition by a one-stage reward function<sup>1</sup>  $r_w(s, a)$  from  $\mathbb{D}$  into the real numbers  $\mathbb{R}$ . Both functions are assumed to be measurable. Finally,  $\beta \in \mathbb{R}^+$  denotes a discount factor. We summarize the setup in the following definition:

**Definition 2.12** (Stochastic control model, cf. Hinderer et al., 2016, Definitions 16.1.2 and 16.1.13). A stochastic control model (SCM) is a tuple

$$(\mathbb{S}, \mathbb{A}, \mathbb{D}, \mathbb{Y}, \mathbb{T}, \mathbb{Q}, r_w, \phi_T, \beta)$$

with the following properties:

<sup>&</sup>lt;sup>1</sup>The reward function is denoted by  $r_w$  with a w in the subscript to distinguish it from the riskless rate r.

- $(\mathbb{S}, \mathcal{S})$  and  $(\mathbb{A}, \mathcal{A})$  are measurable spaces.
- D ∈ S ⊗ A contains the graph of a measurable mapping D : S → A, such that D(s) describes the actions that are admissible in state s ∈ S. D is equipped with the trace D of D in S ⊗ A.
- $(\mathbb{Y}, \mathcal{Y})$  is a measurable space, with  $\mathbb{Y} \neq \emptyset$  the disturbance space.
- $\mathbb{T}: \mathbb{D} \times \mathbb{Y} \to \mathbb{S}$  is a measurable mapping.
- $\mathbb{Q}$  satisfies the conditions of a transition probability from  $\mathbb{D}$  into  $\mathbb{Y}$  according to Definition 2.11, i.e., the event  $\{Y \in C\}$  for  $C \in \mathcal{Y}$  occurs with probability  $\mathbb{Q}(s, a, C)$ . The resulting transition probability  $\mathbb{P}$  from  $\mathbb{D}$  into  $\mathbb{S}$  is given by

$$\mathbb{P}\left(s,a,B\right) = \mathbb{Q}\left(s,a,\{y \in \mathbb{Y} \mid \mathbb{T}\left(s,a,y\right) \in B\}\right), \qquad B \in \mathcal{S}, (s,a) \in \mathbb{D}.$$
(2.1)

- $r_w : \mathbb{D} \to \mathbb{R}$  is measurable.
- $\phi_T : \mathbb{S} \to \mathbb{R}$  is measurable.
- $\beta \in \mathbb{R}^+$  is the discount factor.

Instead of the combination of a disturbance and a transition function, one can define the model via the transition law  $\mathbb{P}$  in (2.1) directly. This variant is called the adjoint Markovian decision process (MDP) to the SCM.

**Remark 2.13** (Disturbance distributions). We note that there exist extensions towards timedependent transition probabilities  $\{\mathbb{Q}_t\}_{t=0,...,T-1}$  and time-dependent one-step reward functions  $\{(r_w)_t\}_{t=0,...,T-1}$  (Hinderer et al., 2016, Def. 16.1.26).

#### 2.2.2 Value Iteration

In our application of an SCM, the investor is interested in maximizing the (expected) portfolio return via optimally choosing the investment strategy. For a policy  $\pi$ , define the (T - t)-stage random gain as

$$G_{t,\pi}(s_t) \coloneqq r_w(s_t, \pi_t(s_t)) + \sum_{i=t+1}^{T-1} \beta^{i-t} \cdot r_w(S_i, \pi_i(S_i)) + \beta^{T-t} \phi_T(S_T).$$
(2.2)

Based on the filtered probability space  $(\Omega, \mathcal{F}, \{\mathcal{F}_t\}_{t \in \{0,1,\dots\}}, \mathbb{P})$ , the (T - t)-stage expected reward is given by

$$\phi_{t,\pi}\left(s_{t}\right) \coloneqq r_{w}\left(s_{t},\pi_{t}\left(s_{t}\right)\right) + \mathbb{E}\left[\sum_{i=t+1}^{T-1}\beta^{i-t}\cdot r_{w}\left(S_{i},\pi_{i}\left(S_{i}\right)\right) + \beta^{T-t}\phi_{T}\left(S_{T}\right) \mid \mathcal{F}_{t}\right].$$
(2.3)

Given our finance application, assuming that we are at some time point  $t \ge 0$ , we intend to solve the problem  $\operatorname{SCM}_{T-t}$ , which, given  $s_t \in \mathbb{S}$ , is defined as  $\max \{\phi_{t,\pi}(s_t) \mid \pi \in \mathbb{F}\}$ . That is, the objective is to maximize the expected (T-t)-stage reward for policy  $\pi$  and a given initial state  $s_t$ . We define the concept of optimal policies and solutions for  $\operatorname{SCM}_{T-t}$  as follows:

Definition 2.14 (Optimal policies and maximizers, Hinderer et al., 2016, Def. 11.3).

(a) A (T-t)-stage policy  $\pi^*$  is optimal for SCM<sub>T-t</sub> if it maximizes  $\pi \mapsto \phi_{t,\pi}(s_t)$ .

#### 2 Mathematical Preliminaries

- (b) A solution of the (T-t)-stage problem consists of the (T-t)-stage optimal policy (if it exists) and the (T-t)-stage value function  $\phi_t : \mathbb{S} \to \mathbb{R}, \phi_t(s) \coloneqq \sup \{\phi_{t,\pi}(s) \mid \pi \in \mathbb{F}\}.$
- (c) A decision rule  $\pi_t$  is called maximizer at time t if for all  $s \in S$ , the action  $\pi_t(s)$  is a maximum point of the mapping

$$a \mapsto r_w(s, a) + \beta \cdot \mathbb{E}\left[\phi_{t+1}\left(\mathbb{T}\left(s, a, Y\right)\right)\right].$$

Note that, without any further assumptions, we are not sure whether  $\text{SCM}_{T-t}$  makes sense since, e.g.,  $\phi_t(s)$  as a supremum of an uncountable set of measurable functions is not necessarily measurable. To express the corresponding conditions adequately, certain operators turn out to be helpful. Assume that there exists a subset  $\mathbb{M}$  of all integrable value functions  $\phi : \mathbb{S} \to \mathbb{R}$ such that  $\phi_T \in \mathbb{M}$ . We define the following operators on  $\mathbb{M}$ :

$$\mathbb{L}\phi(s,a) \coloneqq r_w(s,a) + \beta \cdot \mathbb{E}\left[\phi\left(\mathbb{T}\left(s,a,Y\right)\right)\right],\tag{2.4a}$$

$$\mathbb{U}\phi\left(s\right) \coloneqq \sup_{a \in \mathbb{D}(s)} \mathbb{L}\phi\left(s, a\right).$$
(2.4b)

The dependencies on s and a may be suppressed in the sequel. We require the operators in (2.4) to be well-defined, and  $\mathbb{U}$  to be an endomorphism on  $\mathbb{M}$ . Furthermore, for any  $\phi \in \mathbb{M}$  and  $t \in \{0, \ldots, T-1\}$ , we assume the existence of some decision rule  $\pi_t^*$  maximizing  $\pi_t \mapsto \mathbb{L}\phi(s, \pi_t(s))$  on  $\mathbb{A}$  for all  $s \in \mathbb{S}$ . Under these assumptions, we obtain the following important result:

**Theorem 2.15** (Value iteration for  $\text{SCM}_{T-t}$ , Hinderer et al., 2016, Theorem 16.1.12). Let  $\mathbb{M}$  be a set of integrable value functions  $\phi : \mathbb{S} \to \mathbb{R}$ , and let  $\phi_T \in \mathbb{M}$ . Assume that the operators in (2.4) exist and are well-defined at all time points  $0, \ldots, T-1$ , and furthermore require the following:

- (i) There exists a decision rule  $\pi_{\phi}^*$  maximizing  $\pi_{\phi} \mapsto \mathbb{L}\phi(s, \pi_{\phi}(s))$  on  $\mathbb{A}$  for any  $\phi \in \mathbb{M}$ ,  $s \in \mathbb{S}$ .
- (ii)  $\mathbb{U}$  is an endomorphism on  $\mathbb{M}$ .

Then, for each t = 0, ..., T - 1, the decision rule  $\pi_t^* \coloneqq \pi_{\phi_{t+1}}^*$  is a maximizer at time t, and  $\pi^* \coloneqq \{\pi_t^*\}_{t=0,...,T-1}$  is an optimal policy. Furthermore,  $\phi_t \in \mathbb{M}$  for all t = 0, ..., T and the so-called value iteration holds:

$$\phi_t = \mathbb{U}\phi_{t+1}, \qquad t = 0, \dots, T-1.$$
 (2.5)

*Proof.* See Appendix A.1.

#### 2.2.3 Concave Value Functions

The results in this section, again primarily collected from Hinderer et al. (2016), will be helpful in our optimization techniques. The proofs of standard results are omitted and can be looked up in Roberts and Varberg (1973).

**Definition 2.16** (Convex set). A set  $K \subseteq \mathbb{R}^k$  is convex if, for any  $x, y \in K$  and  $\alpha \in (0, 1)$ ,

$$\alpha x + (1 - \alpha) \, y \in K.$$

In particular, an interval  $I \subset \mathbb{R}$  and the entire space  $\mathbb{R}^k$  are convex.

	_

**Definition 2.17** (Convex and concave functions). Let  $K \subseteq \mathbb{R}^k$  be a convex set. The function  $f: K \to \mathbb{R}$  is called convex if, for  $x, y \in K$ , where  $x \neq y$ , and  $\alpha \in (0, 1)$ ,

$$f(\alpha x + (1 - \alpha)y) \le \alpha f(x) + (1 - \alpha)f(y).$$

$$(2.6)$$

If (2.6) holds with strict inequality, f is said to be strictly convex. f is called (strictly) concave if -f is (strictly) convex.

Given the above relation between convexity and concavity, most results in the literature are stated for convex functions but can be translated straightforwardly.

**Proposition 2.18** (Derivatives and convexity/concavity, cf. Roberts and Varberg (1973), Theorems I.12.B and I.12.C). Let  $K \subseteq \mathbb{R}$  be a one-dimensional convex set, and consider  $f : K \to \mathbb{R}$ . If the first derivative f' exists, then

- f is (strictly) convex if and only if f' is (strictly) increasing.
- f is (strictly) concave if and only if f' is (strictly) decreasing.

If f'' exists, then

- f is convex if and only if  $f'' \ge 0$ . If f'' > 0, then f is strictly convex.
- f is concave if and only if  $f'' \leq 0$ . If f'' < 0, then f is strictly concave.

The following result is useful for combinations of multiple convex (or concave) functions and for functions of more than one variable. If not specified otherwise, we assume that the functions involved are real-valued.

**Proposition 2.19** (Sum of convex/concave functions, Roberts and Varberg (1973), Theorem I.13.A). If both  $f_1$  and  $f_2$  are (strictly) convex/concave on  $K_1 \subseteq \mathbb{R}^k$  and on  $K_2 \subseteq \mathbb{R}^k$ , respectively, then  $(x, y) \mapsto f_1(x) + f_2(y)$  is (strictly) convex/concave on  $K_1 \times K_2$ .

In our optimization context, the next result for maximum points of concave functions, usually formulated for minimum points of convex functions, is particularly interesting.

**Proposition 2.20** (Maximum points of concave functions, Roberts and Varberg (1973), Theorem V.51.A). Assume that f is concave on a convex set  $K \subseteq \mathbb{R}^k$ . Then, the set of maximum points of f, denoted by  $K^*$ , is convex. Furthermore, if f is strictly concave, it has at most one maximum point.

Lastly, we prove an essential proposition needed in the extended portfolio problem in Chapter 4, where parts of the wealth may be consumed at intermediate time points. The result is based on Hinderer et al. (2016, Prop. 7.1.7), where it was formulated for deterministic decision processes. The present extension targets an SCM as of Section 2.2.1. Note that the state space is two-dimensional here.

**Proposition 2.21** (Argumentwise concavity, cf. Hinderer et al., 2016, Prop. 7.1.7). Assume a stochastic control model with arbitrary disturbances from a set  $\mathbb{Y}$ , having state space  $\mathbb{S} = \mathbb{S}_1 \times \mathbb{S}_2$ , action space  $\mathbb{A}$ , and constraint set  $\mathbb{D} = \mathbb{S} \times \mathbb{A}$ . With the transition function denoted by  $\mathbb{T} : \mathbb{D} \times \mathbb{Y} \to \mathbb{S}$ , the finite time horizon T, and  $\phi_t$  the value function at time  $t \in \{0, \ldots, T\}$ , assume that the model satisfies the following properties:

(i)  $S_1$ ,  $S_2$  and A are convex sets.

- (ii)  $\mathbb{T} = (\mathbb{T}_1, \mathbb{T}_2)$ , where  $\mathbb{T}_2(s_1, s_2, a, y) = \mathbb{T}_2(s_2, y)$  is independent of  $(s_1, a)$ , and additionally,  $\mathbb{T}_1(s_1, s_2, a, y)$  is increasing in  $s_1$  and concave in  $(s_1, a)$  on  $\mathbb{S}_1 \times \mathbb{A}$  for all  $s_2 \in \mathbb{S}_2$  and  $y \in \mathbb{Y}$ ,
- (iii) The one-step return  $r_w(s_1, s_2, a)$  is increasing in  $s_1$  for all  $s_2 \in S_2$  and  $a \in A$  and concave in  $(s_1, a)$  for all  $s_2 \in S_2$ .
- (iv) The supremum

$$\sup_{a \in \mathbb{D}(s)} \left\{ r_w(s_1, s_2, a) + \beta \cdot \mathbb{E} \left[ \phi_{t+1} \left( \mathbb{T} \left( s_1, s_2, a, Y \right) \right) \right] \right\}$$

is finite for all  $s_1$ ,  $s_2$ , and for all t.

(v) The terminal value function  $\phi_T(s)$  is increasing and concave in  $s_1$ .

Then, all value functions  $s \mapsto \phi_t(s), t \in \{0, \ldots, T\}$ , are increasing and concave in  $s_1$ .

Proof of Proposition 2.21. We prove by induction on T - t that  $\phi_t$  is increasing and concave in  $s_1$ . For t = T, this is trivially satisfied by (v). Now assume that the assertion is true for some  $\phi_{t+1}$ . We will prove that it also holds true for  $\phi_t$ . Fixing  $\hat{s}_2 \in \mathbb{S}_2$  and  $\hat{y} \in \mathbb{Y}$ , we investigate  $\phi_{t+1}(\mathbb{T}_1(s_1, \hat{s}_2, a, \hat{y}), \mathbb{T}_2(\hat{s}_2, \hat{y}))$ . Using (ii) and the inductive hypothesis, we have that  $\mathbb{T}_1$  is increasing in  $s_1$  and concave in  $(s_1, a)$ , and that  $\phi_{t+1}$  is increasing and concave in  $s_1$ . Therefore,  $\phi_{t+1}(\mathbb{T}_1(s_1, \hat{s}_2, a, \hat{y}), \mathbb{T}_2(\hat{s}_2, \hat{y}))$  is increasing in  $s_1$  and concave in  $(s_1, a)$ . Note that together with (iii), this implies in particular that the function

$$\mathbb{L}\phi_{t+1}(s_1, \hat{s}_2, a) = r_w(s_1, \hat{s}_2, a) + \bar{\beta}\mathbb{E}\left[\phi_{t+1}\left(\mathbb{T}\left(s_1, \hat{s}_2, a, Y\right)\right)\right]$$

is increasing in  $s_1$  and concave in  $(s_1, a)$ . Now define

$$s_1 \mapsto \mathbb{U}\phi_{t+1}(s_1) = \sup_{a \in \mathbb{A}} \mathbb{L}\phi_{t+1}(s_1, \hat{s}_2, a).$$

The concavity of  $\mathbb{L}\phi_{t+1}$  in  $(s_1, a)$  combined with (i) implies that for  $\alpha \in (0, 1)$ ,  $s_1, s'_1 \in \mathbb{S}_1$  and arbitrary  $a, a' \in \mathbb{A}$ ,

$$\mathbb{U}\phi_{t+1}\left(\alpha s_{1}+(1-\alpha)s_{1}'\right) \geq \mathbb{L}\phi_{t+1}\left(\alpha s_{1}+(1-\alpha)s_{1}',\hat{s}_{2},\alpha a+(1-\alpha)a'\right) \\ \geq \alpha \mathbb{L}\phi_{t+1}\left(s_{1},\hat{s}_{2},a\right)+(1-\alpha)\mathbb{L}\phi_{t+1}\left(s_{1}',\hat{s}_{2},a'\right).$$

Since  $\alpha$ , a and a' were arbitrary, this yields

$$\mathbb{U}\phi_{t+1}\left(\alpha s_{1} + (1-\alpha)s_{1}'\right) \geq \alpha \mathbb{U}\phi_{t+1}\left(s_{1}\right) + (1-\alpha)\mathbb{U}\phi_{t+1}\left(s_{1}'\right).$$

This reasoning is true for  $\mathbb{P}$ -almost all  $\hat{s}_2 \in \mathbb{S}_2$ . We can thus deduce that

$$\phi_t\left(s_1, s_2\right) = \mathbb{U}\phi_{t+1}\left(s_1, s_2\right)$$

again is increasing and concave in  $s_1$ , which concludes the proof.

### 2.3 Maximum-Likelihood Estimation

The models considered in this dissertation are supposed to reflect the current market situation the investor faces in an actual application. To this end, the corresponding model parameters need to be estimated based on market data. One of the most common methods in this regard is maximum likelihood estimation. A short introduction of essential definitions and results, based on the extensive work by Bickel and Doksum (2006) and DeGroot and Schervish (2012), is given in this section.

The general idea of maximum likelihood estimation is to choose the set of values for the model parameters that is the "most likely", based on the given dataset. The method avoids assumptions on prior distributions and the use of loss functions. We will consider a set of random variables  $X_1, \ldots, X_n$ ,  $n \in \mathbb{N}$ , forming a random sample from a distribution with parameter (vector)  $\theta$  in a parameter space  $\Theta$ . The central instrument for the maximum likelihood estimation is the likelihood function:

**Definition 2.22** (Likelihood function). For an observed vector  $x = (x_1, \ldots, x_n)$  of values, the likelihood function is the joint probability density function (PDF)  $f_n(x \mid \theta)$  of the observations in a random sample, seen as a function of  $\theta$ .

Based on this notion of a likelihood function, we introduce the terminology of a maximum likelihood estimator:

**Definition 2.23** (Maximum likelihood estimator). For each possible observed vector x, let  $\theta(x) \in \Theta$  denote a value of  $\theta \in \Theta$  for which the likelihood function  $f_n(x \mid \theta)$  is a maximum. The estimator  $\hat{\theta} = \theta(X)$  defined in this way is called a maximum likelihood estimator of  $\theta$ . After X = x is observed, the value  $\delta(x)$  is called a maximum likelihood estimate (MLE) of  $\theta$ .

Note that the parameter value maximizing  $f_n(x \mid \theta)$  is the same as the one that maximizes  $\log f_n(x \mid \theta)$ , which often turns out to be more convenient to handle in our applications. The abbreviation MLE is often used inconsistently for both the estimator and the estimate.

**Definition 2.24** (Fisher information matrix, DeGroot and Schervish, 2012, Def. 8.8.4). Suppose that  $X = (X_1, \ldots, X_n)$  form a random sample from a distribution with joint PDF  $f_n(x \mid \theta)$ , where the value of  $\theta$  lies in an open subset  $\Theta \subseteq \mathbb{R}^k$ . Let  $l_n(x \mid \theta) \coloneqq \log f_n(x \mid \theta)$ . Assume that the mapping  $\theta \mapsto \{x \in \mathbb{R}^n \mid f_n(x \mid \theta) > 0\}$  is constant for all  $\theta$  and that  $l_n(x \mid \theta)$  is twice differentiable with respect to  $\theta$ . The Fisher information matrix  $I(\theta)$  in the random sample X is defined as the  $k \times k$  matrix

$$I(\theta) = Cov_{\theta} \left[ \nabla_{\theta} l_n \left( X \mid \theta \right) \right], \qquad (2.7)$$

where  $Cov_{\theta}$  indicates the covariance given the paramter vector  $\theta$ , and  $\nabla_{\theta}$  is the gradient with respect to  $\theta$ .

Assuming sufficient regularity of the underlying densities so that we can change the order of integration and differentiation, the Fisher information matrix also admits the representation (cf. DeGroot and Schervish, 2012, Theorem 8.8.1)

$$I(\theta) = -\mathbb{E}_{\theta} \left[ \nabla_{\theta}^{2} l_{n} \left( X \mid \theta \right) \right]$$
(2.8)

Under certain regularity conditions, implying that the MLE  $\hat{\theta}$  is consistent and asymptotically normal, as well as the assumptions of Definition 2.24, one can derive that (cf. Bickel and Doksum, 2006, Theorem 6.2.2)

$$\sqrt{n}\left(\hat{\theta}_n - \theta\right) \xrightarrow{n \to \infty} Y \sim \mathcal{N}\left(0, I\left(\theta\right)^{-1}\right),$$
(2.9)

where n is the sample size and the convergence is meant in distribution. In particular, the standard errors of the estimates can be derived via the Fisher information matrix.

# 3 Expected Utility Theory on General Affine GARCH Models

## 3.1 Introduction

This chapter presents a first result concerning expected utility theory (EUT) in the environment of general affine GARCH models. As visualized in Figure 3.1, the main contribution consists of tackling the entire class of general affine GARCH models instead of the Gaussian model version only. An extended version of this chapter was published in Escobar-Anel et al. (2021) and accepted as the author's M.Sc. thesis in the TopMath program at the Technical University of Munich. Therefore, this part does not contribute to the evaluation of the present doctoral dissertation.



Figure 3.1: Overview of the contributions of this dissertation with regard to portfolio optimization under affine GARCH models, increasing the model complexity as well as the complexity of terminal and intermediate utility, see also Figures 1.1, 4.1, 5.1 and 6.1. The orange box coresponds to Chapter 3.

As described in the context of the general motivation for this dissertation in Chapter 1, the idea of approaching optimal resource allocation problems in multiple discrete time steps measuring expected utility derived from terminal wealth originated from Mossin (1968) and was extended by Samuelson (1969). In a continuous-time setting, Merton (1969) presented closed-form solutions for constant relative risk aversion (CRRA) utility functions, considering a geometric

### 3 Expected Utility Theory on General Affine GARCH Models

Brownian motion (GBM) as the underlying asset dynamics. In the subsequent decades, more advanced asset price models were introduced, for example, the popular stochastic volatility model by Heston (1993), which Kraft (2005) solved in the context of EUT. However, just as in the abovementioned example, most considered models were continuous-time, still bearing difficulties regarding implementation and testing. These issues can be addressed using discretetime GARCH models, first introduced by Engle (1982) and Bollerslev (1986). Unfortunately, these usually do not yield closed-form solutions in the context of derivative pricing and portfolio optimization.

This lack of closed-form expressions for derivative pricing was overcome in Heston and Nandi (2000), who introduced the so-called HN-GARCH model. The key feature, the availability of the moment generating function (MGF) in exponentially affine form, allowed for closed-form solutions to asset pricing problems. The same model property can also be exploited in the context of portfolio optimization, as demonstrated by Escobar-Anel et al. (2022a), who solved the EUT problem for an investor maximizing a CRRA utility function from terminal wealth, where the underlying risky asset follows the HN-GARCH dynamics, obtaining an approximate closed-form optimal solution.

While the innovations in the HN-GARCH model are i.i.d. normally distributed, Christoffersen et al. (2006) introduced an extension with time-dependent non-Gaussian innovations a few years later. Badescu et al. (2019) recently formulated general conditions for the affine structure of GARCH models, explicitly requiring the characteristic exponentially affine form of the MGF. The extended paper version (Escobar-Anel et al., 2021) of this chapter solves the EUT problem for general affine GARCH dynamics as of Badescu et al. (2019) and focuses on the implications of the non-Gaussian setting as compared to the Gaussian model or a homoscedasctic variant.

By considering the CRRA class of utility functions and allowing for utility derived from terminal wealth only, the problem setup considered here forms a simple base case of the several extensions considered in Chapters 4, 5 and 6. The theoretical contributions presented in this chapter are twofold:

- Assuming a CRRA utility function for the decision maker and the implied EUT setting of Campbell and Viceira (1999), we demonstrate that the optimal portfolio strategy in a general affine GARCH setting is available in recursive form, and the optimal wealth process is again an affine GARCH process.
- We demonstrate that a slight relaxation of the conditions on general affine GARCH models suffices to include the IG-GARCH model as a special case.

This chapter is organized as follows: In Section 3.2, we define the portfolio optimization problem and give an outline of the approach. Section 3.3 presents the main result for a general affine GARCH model, and Section 3.4 specifically targets the necessary extensions to include the non-Gaussian IG-GARCH model in the setup.

## 3.2 Outline of the Approach

This section presents the general optimization framework for our investment problem and embeds general affine GARCH models in this context. We assume that the log-price process  $X_t = \log P_t^{(1)}$ , the price at time t being  $P_t^{(1)}$ , follows an affine GARCH model (as defined by Badescu et al., 2019), where for simplicity of exposition we assume  $\Delta = 1$  for the length of the time step. Let  $\theta$  be a vector of parameters and  $\{\epsilon_t\}_{t=1,2,\dots}$  a sequence of  $\mathcal{F}_t$ -measurable and  $\mathcal{F}_{t-1}$ -conditional i.i.d. random variables with zero mean and a finite moment generating function. Based on the associated filtered probability space  $(\Omega, \mathcal{F}, \{\mathcal{F}_t\}_{t=0,1,\dots}, \mathbb{P})$  with physical measure  $\mathbb{P}$ , the model dynamics is:

$$Y_t \coloneqq X_t - X_{t-1} = f_1(h_t, \theta) + \sqrt{h_t} \epsilon_t, \qquad (3.1a)$$

$$h_{t+1} = f_2(h_t, \theta) + f_3(h_t, \epsilon_t, \theta), \qquad (3.1b)$$

where  $f_1, f_2$  are affine in  $h_t$ , while  $f_3$  is such that the following representation for the conditional MGF holds:

$$\Psi_{(X_{t},h_{t+1})}(u,v \mid \mathcal{F}_{t-1}) \coloneqq \mathbb{E}\left[\exp\left\{u \cdot X_{t} + v \cdot h_{t+1}\right\} \mid \mathcal{F}_{t-1}\right] \\ = \exp\left\{uX_{t-1} + A(u,v;t-1,t) + B(u,v;t-1,t) \cdot h_{t}\right\}.$$
(3.2)

The conditional expectation w.r.t.  $\mathcal{F}_t$  will also be denoted via the subindex t, e.g., with  $\mathbb{E}_t$ , in the sequel. Note that the moment generating function in (3.2) can also be formulated for the log return instead of the log price as the first element, making the term  $uX_{t-1}$  in the exponential function vanish. The choice of  $f_1$ ,  $f_2$ , and  $f_3$  for a particular application shapes the coefficients A and B in Equation (3.2) according to this requirement. The form of said coefficients will also play a significant role in our framework. Since it is clear from the context that we use the one-step conditional MGF, the last two time arguments of A and B are usually obsolete, and we will omit them where possible.

We consider a portfolio optimization problem with a finite time horizon T in a setting with one risky asset and the cash account  $P_t^{(0)}$ . For  $t \in \{0, \ldots, T-1\}$ , let  $\pi_t$  denote the fraction of wealth invested in the risky asset, let  $V_t$  be the corresponding wealth and  $w_0 = \log v_0$  the log of the initial wealth. Following the reasoning in Escobar-Anel et al. (2022a), we approximate returns in the self-financing condition (SFC) by log prices. That is, we start with the true equation

$$\frac{V_t}{V_{t-1}} = \pi_{t-1} \frac{P_t^{(1)}}{P_{t-1}^{(1)}} + (1 - \pi_{t-1}) \frac{P_t^{(0)}}{P_{t-1}^{(0)}} = \pi_{t-1} \frac{P_t^{(1)}}{P_{t-1}^{(1)}} + (1 - \pi_{t-1}) e^r,$$
(3.3)

where r is the continuously compounded riskless rate, and use the following Taylor approach: For the log return of the risky asset, we write

$$X_{t} - X_{t-1} = \log\left(\frac{P_{t}^{(1)}}{P_{t-1}^{(1)}}\right) \approx \frac{P_{t}^{(1)} - P_{t-1}^{(1)}}{P_{t-1}^{(1)}} - \frac{1}{2}\left(\frac{P_{t}^{(1)} - P_{t-1}^{(1)}}{P_{t-1}^{(1)}}\right)^{2}$$
$$\approx \frac{P_{t}^{(1)} - P_{t-1}^{(1)}}{P_{t-1}^{(1)}} - \frac{1}{2}Var\left[\frac{P_{t}^{(1)} - P_{t-1}^{(1)}}{P_{t-1}^{(1)}} \mid \mathcal{F}_{t-1}\right], \quad (3.4)$$

where we approximated the squared return by its conditional variance in the last step. Observing from (3.4) that  $Var\left[\frac{P_t^{(1)}-P_{t-1}^{(1)}}{P_{t-1}^{(1)}} \mid \mathcal{F}_{t-1}\right] \approx Var\left[X_t - X_{t-1} \mid \mathcal{F}_{t-1}\right] = h_t$ , we receive

$$\frac{P_t^{(1)} - P_{t-1}^{(1)}}{P_{t-1}^{(1)}} \approx X_t - X_{t-1} + \frac{1}{2}h_t.$$
(3.5)

For the cash account, the linear approximation  $\log x \approx x - 1$  for x around 1 yields

$$\frac{P_t^{(0)} - P_{t-1}^{(0)}}{P_{t-1}^{(0)}} \approx \log\left(\frac{P_t^{(0)}}{P_{t-1}^{(0)}}\right) = r.$$
(3.6)

Now, looking at the return on wealth and performing the same steps as for the return of the risky asset above (with the variance of the return on wealth being  $\pi_{t-1}^2 h_t$  according to (3.3)), we arrive at

$$\log\left(\frac{V_t}{V_{t-1}}\right) \approx \left(\frac{V_t}{V_{t-1}} - 1\right) - \frac{1}{2} \left(\frac{V_t}{V_{t-1}} - 1\right)^2$$
$$\approx \pi_{t-1} \frac{P_t^{(1)} - P_{t-1}^{(1)}}{P_{t-1}^{(1)}} + (1 - \pi_{t-1}) \frac{P_t^{(0)} - P_{t-1}^{(0)}}{P_{t-1}^{(0)}} - \frac{1}{2} \pi_{t-1}^2 h_t$$
$$\approx \pi_{t-1} \left(X_t - X_{t-1} + \frac{1}{2} h_t\right) + (1 - \pi_{t-1}) r - \frac{1}{2} \pi_{t-1}^2 h_t$$

All in all, for t = 1, ..., T, this leads to an approximation  $W_t$  of the log wealth log  $V_t$  that reads

$$W_t = W_{t-1} + \pi_{t-1}(X_t - X_{t-1}) + \frac{1}{2} \left(\pi_{t-1} - \pi_{t-1}^2\right) h_t + (1 - \pi_{t-1})r, \qquad (3.7)$$

with  $W_0 = w_0$ . This second-order approximation is well-studied – in particular, it corresponds to Equation (16) in Campbell and Viceira (1999). Equation (3.7) exactly equals (3.3) in the limit for  $\Delta \to 0$ , i.e., as the length of the trading interval approaches zero. We refer to the fulllength paper Escobar-Anel et al. (2021) for an investigation of the quality of this approximation in this framework.

The goal is to maximize the expected utility from terminal wealth over the set of admissible, self-financing, relative portfolio strategies  $\{\pi_t\}_{t=0}^{T-1}$ . To this end, assume wealth is assessed by the decision maker according to a CRRA utility function of the form  $U(v) = \frac{1}{\gamma}v^{\gamma}$  for some parameter  $\gamma < 1$ , implying a relative risk aversion of  $1 - \gamma$ . Using the approximation of the log wealth via  $W_T$  as described above, the stochastic control problem can be respresented as follows:

$$\sup_{\{\pi_t\}_{t=0}^{T-1}} \mathbb{E}\left[\frac{1}{\gamma} (V_T)^{\gamma} \mid \mathcal{F}_0\right] \approx \max_{\{\pi_t\}_{t=0}^{T-1}} \mathbb{E}\left[\frac{\exp\left\{\gamma W_T\right\}}{\gamma} \mid \mathcal{F}_0\right] \eqqcolon \phi_0 (w_0, h_1), \quad (3.8)$$

where h denotes the conditional variance process of the log return of the underlying asset.

In the spirit of Section 2.2, let  $\mathbb{A}$  be the set of admissible portfolios,  $\mathbb{H} = (0, \infty)$  and  $\mathbb{Y} = (0, \infty)$ be the domains of the conditional variance and the log return of the stock prices, respectively. Let  $\mathbb{W} \subset \mathbb{R}$  be the set of possible values for the log wealth, and set  $\mathbb{S} = \mathbb{W} \times \mathbb{H}$ . Then, the transition function is given by  $\mathbb{T} : \mathbb{S} \times \mathbb{A} \times \mathbb{Y} \to \mathbb{S}$ ,

$$\mathbb{T}(s,a,Y) = \mathbb{T}(W,h,a,Y) \coloneqq \left(W + aY + \frac{1}{2}(a-a^2)h + (1-a)r, \\ f_2(h,\theta) + f_3\left(h,\frac{Y - f_1(h,\theta)}{\sqrt{h}},\theta\right)\right).$$
(3.9)

Now assume that there exists a subset  $\mathbb{M}$  of all integrable value functions  $\phi : \mathbb{S} \to \mathbb{R}$  such that  $\phi_T(W_T) = U(\exp\{W_T\}) \in \mathbb{M}$ . With the operators  $\mathbb{L}$  and  $\mathbb{U}$ ,

$$\mathbb{L} \phi (W, h, a) \coloneqq \mathbb{E} \left[ \phi \left( \mathbb{T} \left( W, h, a, Y \right) \right) \right],$$
$$\mathbb{U} \phi (W, h) \coloneqq \sup_{a \in \mathbb{A}} \mathbb{L} \phi \left( W, h, a \right),$$

assumed to be well-defined for all  $\phi \in \mathbb{M}$ , we moreover require for any  $\phi \in \mathbb{M}$  the existence of some  $a : \mathbb{W} \times \mathbb{H} \to \mathbb{A}$  with a(W, h) maximizing  $a \mapsto \mathbb{L} \phi(W, h, a)$  on  $\mathbb{A}$  for all  $W \in \mathbb{W}$  and  $h \in \mathbb{H}$ , and  $\mathbb{U} : \mathbb{M} \to \mathbb{M}$ .

With the above requirements, we ensure that the conditions of Theorem 2.15 are satisfied and that we can use the value iteration. We thus maximize recursively step by step in order to obtain the optimal solution  $\{\pi_t^*\}_{t=0}^{T-1}$ , defined as  $\pi_t^*(W_t, h_{t+1}) \coloneqq \arg \max_{a \in \mathbb{A}} \mathbb{L}\phi_{t+1}(W_t, h_{t+1}, a)$ , and the value iteration

$$\phi_t (W_t, h_{t+1}) = \mathbb{U} \phi_{t+1} (W_t, h_{t+1})$$

$$= \max_{a \in \mathbb{A}} \mathbb{E}_t \left[ \phi_{t+1} \left( \mathbb{T} (W_t, h_{t+1}, a, Y_t) \right) \right], \quad t \in \{0, \dots, T-1\},$$
(3.10)

with the obvious terminal condition  $\phi_T(W_T) = U(\exp\{W_T\}).$ 

## 3.3 Solution to the Portfolio Optimization Problem

This section presents the solution to the portfolio optimization problem outlined in Section 3.2. The optimal solution to (3.8) is provided in the next theorem, which is followed by a corollary on the affine GARCH nature of the optimal wealth process.

**Theorem 3.1** (Maximum expected utility representation). Assume  $\gamma < 0$ , and let the log price of the risky asset follow an affine GARCH model, where for all  $t \in \{0, \ldots, T-1\}$ , and all admissible  $\pi_t$  and v, the coefficients in (3.2) satisfy

$$\frac{\partial^2}{\partial u^2} A\left(\gamma \pi_t, v\right) \ge 0 \quad and \quad \frac{\partial^2}{\partial u^2} B\left(\gamma \pi_t, v\right) > 0. \tag{3.11}$$

Then at time t, the maximum expected utility from terminal wealth can be written as

$$\phi_t (W_t, h_{t+1}) = \frac{1}{\gamma} \exp \left\{ \gamma W_t + D_{t,T} (\pi_t^*) + E_{t,T} (\pi_t^*) \cdot h_{t+1} \right\},$$
(3.12)

with  $D_{t,T}$  and  $E_{t,T}$  given by the recursive representations

$$D_{t,T}(\pi_t^*) = D_{t+1,T}(\pi_{t+1}^*) + (1 - \pi_t^*) \gamma r + A\left(\gamma \pi_t^*, E_{t+1,T}(\pi_{t+1}^*)\right), \qquad (3.13a)$$

$$E_{t,T}(\pi_t^*) = B\left(\gamma \pi_t^*, E_{t+1,T}(\pi_{t+1}^*)\right) + \frac{\gamma}{2} \left(\pi_t^* - (\pi_t^*)^2\right).$$
(3.13b)

with  $E_{T,T} = D_{T,T} = 0$ . The optimal fraction of wealth invested in the risky asset at time t is given as a solution  $\pi_t^*$  to the equations

$$\frac{\partial}{\partial u} A\left(\gamma \pi_t, E_{t+1,T}(\pi_{t+1}^*)\right) = r, \qquad (3.14a)$$

$$\frac{\partial}{\partial u} B\left(\gamma \pi_t, E_{t+1,T}(\pi_{t+1}^*)\right) = \pi_t - \frac{1}{2}.$$
(3.14b)

Moreover, a solution  $\pi_t^*$  satisfying (3.11), (3.14a), and (3.14b) is a maximum.

Note that an optimal solution is required to satisfy both equations in (3.14), which in general might not be possible. However, in the special cases of HN-GARCH (c.f. Heston and Nandi, 2000) and IG-GARCH models (c.f. Christoffersen et al., 2006), (3.14a) is trivially satisfied and there remains only (3.14b) to solve.

Proof of Theorem 3.1. As outlined above, we use Bellman's value iteration in order to maximize via backwards induction and the optimal solution at time t + 1 for our approach at time t. In

general, this results in the calculations below, where for the terminal step at time t = T - 1 we just work with

$$\phi_T(W_T, h_{T+1}) \coloneqq \phi_T(W_T) = U(\exp\{W_T\}) = \frac{1}{\gamma} \exp\{\gamma W_T\},$$

resulting in  $D_{T,T} = E_{T,T} = 0$ . We obtain:

$$\begin{split} \phi_{t}\left(W_{t},h_{t+1}\right) &= \max_{\pi_{t}} \mathbb{E}_{t}\left[\phi_{t+1}\left(W_{t+1},h_{t+2}\right)\right] \\ &= \max_{\pi_{t}} \mathbb{E}_{t}\left[\frac{1}{\gamma}\exp\left\{D_{t+1,T}(\pi_{t+1}^{*}) + \gamma W_{t+1} + E_{t+1,T}(\pi_{t+1}^{*}) \cdot h_{t+2}\right\}\right] \quad (3.15) \\ &= \max_{\pi_{t}} \frac{1}{\gamma}\exp\left\{D_{t+1,T}(\pi_{t+1}^{*})\right\} \\ &\times \mathbb{E}_{t}\left[\exp\left\{\gamma \cdot \left[W_{t} + \pi_{t}\left(X_{t+1} - X_{t}\right) + \frac{1}{2}\left(\pi_{t} - \pi_{t}^{2}\right)h_{t+1}\right. \right. \right. \right. \\ &\left. + \left(1 - \pi_{t}\right)r\right] + E_{t+1,T}(\pi_{t+1}^{*}) \cdot h_{t+2}\right\}\right] \\ &= \max_{\pi_{t}} \frac{1}{\gamma}\exp\left\{D_{t+1,T}(\pi_{t+1}^{*}) + \left(1 - \pi_{t}\right)\gamma r + \gamma W_{t} + \frac{\gamma}{2}\left(\pi_{t} - \pi_{t}^{2}\right)h_{t+1}\right\} \\ &\times \mathbb{E}_{t}\left[\exp\left\{\gamma\pi_{t}\left(X_{t+1} - X_{t}\right) + E_{t+1,T}(\pi_{t+1}^{*}) \cdot h_{t+2}\right\}\right] \\ &= \max_{\pi_{t}} \frac{1}{\gamma}\exp\left\{D_{t+1,T}(\pi_{t+1}^{*}) + \left(1 - \pi_{t}\right)\gamma r + \gamma W_{t} + \frac{\gamma}{2}\left(\pi_{t} - \pi_{t}^{2}\right)h_{t+1}\right\} \\ &\times \Psi_{(X_{t+1} - X_{t},h_{t+2})}\left(\gamma\pi_{t}, E_{t+1,T}(\pi_{t+1}^{*}) \mid \mathcal{F}_{t}\right) \\ &= \max_{\pi_{t}} \frac{1}{\gamma}\exp\left\{D_{t,T}(\pi_{t}) + \gamma W_{t} + E_{t,T}(\pi_{t}) \cdot h_{t+1}\right\}, \end{aligned}$$

where we used the availability of the MGF in the affine GARCH setting according to (3.2) and define D and E recursively according to (3.13). Note that  $\gamma < 0$  together with the secondorder conditions in (3.11) are sufficient for the function in (3.16) to be concave in  $\pi_t$ . The first-order conditions in (3.14) thus characterize the optimal solution to our portfolio problem. In particular, a solution satisfying (3.11), (3.14a), and (3.14b) is a maximum.

**Corollary 3.2** (Affine GARCH process for optimal log wealth). The optimal log-wealth process  $\{W_t\}_t$  follows an affine GARCH process. Furthermore, the optimal solution  $\pi_t^*$  does not depend on the conditional variance or the wealth.

*Proof.* We start with the self-financing condition in Equation (3.7). The equation shows that the conditional variance of  $W_t$  is given by  $\pi_{t-1}^2 h_t$ . The affine GARCH representation for the optimal log-wealth process can now be deduced from the proof of Theorem 3.1. That is, starting in line (3.15), disregarding the maximum operator and unnecessary factors, we can proceed to line (3.16) to solve the expectation below for the conditional bivariate moment generating function of  $W_t$  and  $h_{t+1}$ . Adopting also the definitions in (3.13) results in:

$$\Psi_{(W_{t},h_{t+1})}(u,v \mid \mathcal{F}_{t-1}) = \mathbb{E}_{t-1} \left[ \exp \left\{ uW_{t} + vh_{t+1} \right\} \right]$$
  
=  $\exp \left\{ uW_{t-1} + (1 - \pi_{t-1})ur + A(u\pi_{t-1},v) + \left( B(u\pi_{t-1},v) + \frac{u}{2} \left( \pi_{t-1} - (\pi_{t-1})^{2} \right) \right) \cdot h_{t} \right\}.$ 

This satisfies the requirements for affine GARCH processes outlined in Equations (3.1) and (3.2). An investigation of the equations defining the optimal solution yields the second statement of the corollary.

It is not difficult to see that the HN-GARCH model (Heston and Nandi, 2000), which was already tackled in this context by Escobar-Anel et al. (2022a), is a particular case of Theorem 3.1. In particular, we introduce the set of parameters  $\theta = (\lambda, \omega, \beta, \alpha, \rho)$  and an i.i.d. sequence  $\{\epsilon_t\}_t$  of standard normal innovations. Then, choosing  $f_1(h_t, \theta) = r + \lambda h_t$ ,  $f_2(h_t, \theta) = \omega + \beta h_t$  and  $f_3(h_t, \epsilon_t, \theta) = \alpha (\epsilon_t - \rho \sqrt{h_t})^2$ , we obtain the following dynamics for the log return and its conditional variance process in the HN-GARCH model:

$$X_t - X_{t-1} = r + \lambda h_t + \sqrt{h_t} \epsilon_t, \qquad (3.17a)$$

$$h_{t+1} = \omega + \beta h_t + \alpha \left(\epsilon_t - \rho \sqrt{h_t}\right)^2.$$
(3.17b)

The proof by Heston and Nandi (2000) for the exponentially affine structure of the MGF and the derivation for the coefficients A and B can be found in Appendix B.1.

## 3.4 Application to IG-GARCH Model

In this section, we present the IG-GARCH model (c.f. Christoffersen et al., 2006) as a special case of our main result. Assume the following dynamics for the log return  $X_t - X_{t-1} = \log P_t^{(1)} - \log P_{t-1}^{(1)}$  and its conditional variance process  $h = \{h_t\}_{t=1,...,T}$ :

$$X_t - X_{t-1} = r + \nu h_t + \eta y_t, \tag{3.18a}$$

$$h_{t+1} = w + bh_t + cy_t + a\frac{h_t^2}{y_t},$$
(3.18b)

with  $\eta < 0$  and  $\{y_t\}_t$  a sequence of random variables with inverse Gaussian distribution and single parameter  $\delta_t = \delta(t) = \frac{h_t}{\eta^2}$ . In particular, translating this single-parameter form to the known representation of an inverse Gaussian distribution<sup>1</sup>, we have  $y_t \sim \text{IG}(\delta_t, \delta_t^2)$  and  $\mathbb{E}_{t-1}[y_t] = Var_{t-1}[y_t] = \delta_t$ .

Comparing Equation (3.18a) to (3.1), we first note that scaling of the inverse Gaussian random variable with some parameter  $\alpha > 0$  yields  $\alpha y_t \sim \text{IG}(\alpha \delta_t, \alpha \delta_t^2)$ . Together with the relation

$$\eta y_t = \sqrt{h_t} \cdot \frac{\eta}{\sqrt{h_t}} y_t = \sqrt{h_t} \cdot \left(-\frac{1}{\sqrt{\delta_t}} y_t\right) =: \sqrt{h_t} \cdot \xi_t, \qquad (3.19)$$

this implies that, in order to obtain the form of (3.1), we need to set  $-\xi_t \sim \text{IG}\left(\sqrt{\delta_t}, \sqrt{\delta_t}^3\right) = \text{IG}\left(\sqrt{h_t}/|\eta|, \sqrt{h_t}^3/|\eta|^3\right)$ . From the variance of a two-parameter inverse Gaussian variable, we obtain  $Var_{t-1}\left[\xi_t\right] = 1$ . The minus compensates for  $\eta < 0$ . Standardizing this random variable further w.r.t. the mean by subtracting  $-\sqrt{h_t}/\eta$  finally allows us to present the embedding of the IG-GARCH model into the general setting by Badescu et al. (2019). With a set of parameters denoted by  $\theta = (\nu, \eta, w, b, c, a)$ , we have

$$f_1(h_t,\theta) = r + \nu \cdot h_t, \quad f_2(h_t,\theta) = w + b \cdot h_t, \quad f_3(h_t,y_t,\theta) = cy_t + a\frac{h_t^2}{y_t},$$

<sup>&</sup>lt;sup>1</sup>An inverse Gaussian random variable  $\xi \sim \text{IG}(\mu, \lambda)$  with  $\mu, \lambda > 0$  (see, e.g., Seshadri, 1999) has support  $(0, \infty)$ . Furthermore,  $\mathbb{E}[\xi] = \mu$  and  $Var[\xi] = \frac{\mu^3}{\lambda}$ .

where  $\{\epsilon_t\}_t$ ,  $\epsilon_t = \xi_t + \sqrt{h_t}/\eta$ , are standardized inverse Gaussian random variables. However, we usually work with  $\tilde{f}_1 = r + \left(\nu + \frac{1}{\eta}\right)h_t$  and use  $\{\xi_t\}_t$  as innovations in (3.1). From this mean-standardized formulation, we derive the market price of risk as  $\lambda = \nu + \frac{1}{\eta}$ .<sup>2</sup> The derivation of the exponentially affine structure of the MGF in the IG-GARCH model can be found in Appendix B.2.

Remark 3.3 (Inverse Gaussian innovations).

- (1) Note that the shape of  $y_t$  still depends on  $h_t$ . Thus, the IG-GARCH model only fulfills a relaxation of the condition on the sequence of shocks in the general affine setting. In particular, instead of explicitly relying on the innovations  $\{\epsilon_t\}_{t=1,...T}$  (or  $\{\xi_t\}_{t=1,...T}$ ) in (3.1) being i.i.d., we require that the choice of distribution allows for the exponentially affine representation of the conditional bivariate generating function in (3.2). This slight adjustment accommodates the IG-GARCH model while preserving our framework's necessary and important properties.
- (2) Via sending  $\eta \to 0$ , the conditional skewness S and excess kurtosis K on the one-step distribution, which are determined by (c.f. Christoffersen et al., 2006)

$$\mathcal{S}_t \left[ X_{t+1} - X_t \right] = 3\eta \left( h_{t+1} \right)^{-1/2}, \quad \mathcal{K}_t \left[ X_{t+1} - X_t \right] = 15\eta^2 \left( h_{t+1} \right)^{-1}, \quad (3.20)$$

vanish and the standardized shock becomes standard normal in the limit. This can be used to derive the continuous-time Heston model (Heston, 1993) in the limit (cf. Christoffersen et al., 2006, pp. 258).<sup>3</sup>

The first two conditional moments of the log return are provided next, where we use the length of a time step being exactly one day, i.e.,  $\Delta = 1$ .

$$\mathbb{E}_t \left[ X_{t+1} - X_t \right] = r + \left( \nu + \frac{1}{\eta} \right) \cdot h_{t+1}, \qquad (3.21a)$$

$$Var_t[X_{t+1}] = h_{t+1}.$$
 (3.21b)

The following corollary applies our main results to the IG-GARCH setting.

**Corollary 3.4** (IG-GARCH model). Assuming existence, the optimal solution to the stochastic control problem in (3.8), with the log-price dynamics represented in the form of (3.18), is given as a real solution to the equation

$$\nu + \frac{\sqrt{\left(1 - 2E_{t+1,T}^* a\eta^4\right)}}{\eta \sqrt{\left(1 - 2\eta\gamma\pi_t - 2E_{t+1,T}^* c\right)}} = \pi_t - \frac{1}{2},\tag{3.22}$$

where  $E_{t+1,T}^* := E_{t+1,T}(\pi_{t+1}^*)$ , if the following two conditions are satisfied for all admissible values of  $\pi_t$  and  $t \in \{0, \ldots, T-1\}$ :

$$1 - 2E_{t+1,T}^* a\eta^4 > 0 \quad and \quad 1 - 2\eta\gamma\pi_t - 2E_{t+1,T}^* c > 0.$$
(3.23)

<sup>&</sup>lt;sup>2</sup>It is possible to derive a set of HN-GARCH parameters matching the first two moments of a given IG-GARCH parameter set (Christoffersen et al., 2006). For this pair of models, the market price of risk in the IG-GARCH coincides with the parameter  $\lambda$  of the HN-GARCH model in (3.17a).

<sup>&</sup>lt;sup>3</sup>It is known (Escobar-Anel et al., 2022a) that the optimal strategy in the HN-GARCH model converges to the continuous-time Heston solution as of Kraft (2005) for  $\Delta \rightarrow 0$  under certain conditions. Thus, imposing these requirements and taking the limit  $\eta \rightarrow 0$ , our optimal solution approaches the continuous-time Heston strategy.

The solutions of Equation (3.22) can be found via the roots of a cubic polynomial with real coefficients, and there is exactly one real solution if the discriminant of this polynomial is negative. A solution  $\pi_t^*$  satisfying (3.22) and (3.23) is a maximum.

*Proof.* As of Christoffersen et al., 2006, and as stated in Appendix B.2, the coefficients A and B of the conditional bivariate moment generating function in (3.2) in the IG-GARCH model are given by

$$A\left(\gamma\pi_{t}, E_{t+1,T}^{*}\right) = r\gamma\pi_{t} + E_{t+1,T}^{*} w - \frac{1}{2}\log\left(1 - 2E_{t+1,T}^{*}a\eta^{4}\right),$$
  

$$B\left(\gamma\pi_{t}, E_{t+1,T}^{*}\right) = E_{t+1,T}^{*}b + \nu\gamma\pi_{t} + \eta^{-2}$$
  

$$-\eta^{-2}\sqrt{\left(1 - 2E_{t+1,T}^{*}a\eta^{4}\right)\left(1 - 2\eta\gamma\pi_{t} - 2E_{t+1,T}^{*}c\right)}$$

Combining these formulas with Theorem 3.1, i.e., plugging the above expressions for A and B into (3.13), we directly obtain the explicit representations for  $D_{t,T}$  and  $E_{t,T}$  in terms of the parameters of the IG-GARCH model:

$$D_{t,T} = D_{t+1,T} + \gamma r + E_{t+1,T}^* w - \frac{1}{2} \log \left( 1 - 2E_{t+1,T}^* a \eta^4 \right), \qquad (3.24a)$$

$$E_{t,T}(\pi_t) = \frac{1}{2} \left( \pi_t - \pi_t^2 \right) + E_{t+1,T}^* b + \nu \gamma \pi_t + \eta^{-2} \left( 1 - \sqrt{\left( 1 - 2E_{t+1,T}^* a \eta^4 \right) \left( 1 - 2\eta \gamma \pi_t - 2E_{t+1,T}^* c \right)} \right).$$
(3.24b)

Thus, the two Equations (3.14a) and (3.14b) reduce to (3.22), which needs to be solved for  $\pi_t$  in order to obtain the optimal fraction invested in the risky asset. With the second-order conditions (3.23) fulfilled, we deduce that a solution must be a maximum.

Working towards solving Equation (3.22), we first isolate the radicals, square the equation and rearrange terms in order to obtain the following cubic equation:

$$\frac{1 - 2E_{t+1,T}^* a\eta^4}{\eta^2 \cdot \left(1 - 2\eta\gamma\pi_t - 2E_{t+1,T}^* c\right)} = \pi_t^2 - (1 + 2\nu) \cdot \pi_t + \left(\nu + \frac{1}{2}\right)^2$$

$$\Leftrightarrow \quad \frac{1}{\eta^2} \cdot \left(1 - 2E_{t+1,T}^* a\eta^4\right) = -2\eta\gamma\pi_t^3 + \left(1 - 2E_{t+1,T}^* c\right)\pi_t^2$$

$$\quad + 2\eta\gamma\left(1 + 2\nu\right) \cdot \pi_t^2 - (1 + 2\nu)\left(1 - 2E_{t+1,T}^* c\right)\pi_t$$

$$\quad - 2\eta\gamma\left(\nu + \frac{1}{2}\right)^2\pi_t + \left(1 - 2E_{t+1,T}^* c\right)\left(\nu + \frac{1}{2}\right)^2$$

$$\Leftrightarrow \quad p_3 \cdot \pi_t^3 + p_2 \cdot \pi_t^2 + p_1 \cdot \pi_t + p_0 = 0, \qquad (3.25)$$

where the coefficients of the polynomial in (3.25) are given by

$$p_3 = 2\eta\gamma, \tag{3.26a}$$

$$p_2 = -2\eta\gamma \left(2\nu + 1\right) - \left(1 - 2E_{t+1,T}^*c\right), \qquad (3.26b)$$

$$p_1 = 2\eta\gamma \left(\nu + \frac{1}{2}\right)^2 + (2\nu + 1) \cdot \left(1 - 2E_{t+1,T}^*c\right), \qquad (3.26c)$$

$$p_0 = \frac{1}{\eta^2} \left( 1 - 2E_{t+1,T}^* a \eta^4 \right) - \left( \nu + \frac{1}{2} \right)^2 \cdot \left( 1 - 2E_{t+1,T}^* c \right).$$
(3.26d)

The polynomial in (3.25) has three roots, and the number of real solutions among these can be deduced from the sign of the discriminant of the polynomial, which in the cubic case is known to be (Karpfinger and Meyberg, 2013, p. 366)

$$\mathcal{D} = 18 \, p_3 \, p_2 \, p_1 \, p_0 + p_2^2 \, p_1^2 - 4 \, p_3 \, p_1^3 - 4 \, p_2^3 \, p_0 - 27 \, p_3^2 \, p_0^2,$$

where we have a single real solution (and two non-real complex conjugate roots) if  $\mathcal{D} < 0$ .  $\Box$ 

The fact that the number of real solutions of a cubic polynomial can be deduced from the sign of its discriminant makes it worth pointing out that in Equations (3.26), the arrangement ensures that the signs of three coefficients can easily be derived. Here, we take into account Conditions (3.23) and use that both  $\eta$  and  $\gamma$  are assumed to be strictly negative, while, according to the estimates by Christoffersen et al. (2006, Table 2) and, more recently, by Babaoğlu et al. (2018, Table 2),  $\nu$  is positive with order of magnitude 3. In particular, we have  $p_3 > 0$  and  $p_1 > 0$ , while  $p_2 < 0$  and no immediate decision can be made for  $p_0$ . However, from this position, the sign of  $\mathcal{D}$  cannot be told immediately. We thus refer to our numerical analyses in Escobar-Anel et al. (2021), which show that for relevant values of  $\gamma$ , we have exactly one real solution to the optimality equation.

Note that we can make sure that Condition (3.11) is satisfied by assuming that the arguments of both square roots in (3.22) are positive, see (3.23). This implicitly imposes conditions on  $\pi_t^*$ , which need to be checked in practice. However, due to the order of magnitude of the parameter  $\eta$  being 10<sup>-4</sup> (considering the estimates from Christoffersen et al. (2006) and Babaoğlu et al. (2018) again), the conditions can be expected to be satisfied at least by the optimal terminal solution.

## 3.5 Conclusion

This chapter presents an approximate closed-form solution to a portfolio optimization problem with one risky asset whose log return follows a general affine GARCH process, allowing for non-Gaussian innovations. The investor maximizes a CRRA utility from terminal wealth. Using a second-order approximation for the self-financing condition supported in the literature, we apply the value iteration (Bellman's principle) iteratively to obtain the optimal strategy in recursive form. The optimal wealth process is shown to follow an affine GARCH process as well. Furthermore, the IG-GARCH model is developed as a particular case of the main result, which constitutes the first EUT closed-form solution available in the literature for a non-Gaussian GARCH model. The analysis shows that we have a unique real solution to the optimality equation for popular sets of parameter estimates. One outstanding advantage of the new model is that asset returns follow a leptokurtic, negatively skewed distribution, allowing for an investigation of the impact of these non-Gaussian features on portfolio allocation.

## 4 Optimal Consumption and Investment

## 4.1 Introduction

After the introduction to the expected utility theory (EUT) problem in an affine GARCH framework in Chapter 3, this chapter presents the solution to an extended problem, where the investor may also derive utility from consumption at intermediate time points. Figure 4.1 illustrates that we target the vertical dimension in the diagram. Hence, the present chapter corresponds to the green box. This chapter is published as a full-length paper, entitled "Optimal Consumption and Investment in General Affine GARCH Models" (Escobar-Anel et al., 2024b).



Figure 4.1: Overview of the contributions of this dissertation with regard to portfolio optimization under affine GARCH models, increasing the model complexity as well as the complexity of terminal and intermediate utility, see also Figures 1.1, 3.1, 5.1 and 6.1. The green box to Chapter 4.

The optimization of consumption and investment has been of great importance to insurance companies, pension funds, and individual investors for decades. We have already mentioned the seminal contributions made by Merton (1969, 1971), based on a geometric Brownian motion (GBM) model for the underlying risky asset with utility functions of the constant relative risk aversion (CRRA) and hyperbolic absolute risk aversion (HARA) type, respectively. Since then, the problem has been tackled for a variety of different models, including stochastic factors like volatility, interest rate, correlations, and different environments like complete and incomplete markets settings. We have pointed out in Chapter 1 that the vast majority of models considered

### 4 Optimal Consumption and Investment

in the literature have been continuous-time in nature, the reason being the availability of closedform, analytical solutions rooted in the convenience of Itô's processes and partial differential equations. For instance, Kim and Omberg (1996) explored the behavior of investors maximizing HARA utility functions, with the risk premium of the single risky asset following an Ornstein-Uhlenbeck process. The original solution approach was complemented by a duality-based proof of the same problem provided by Battauz et al. (2015). As mentioned earlier, Kraft (2005) solved the portfolio allocation problem for the famous Heston (1993) model, while Liu and Pan (2003) found closed-form solutions in the presence of stochastic volatility and price jumps. Various consumption and investment problems have been tackled, e.g., by Chacko and Viceira (2005) (with Duffie and Epstein (1992a,b) utility) and in a multivariate setting by Liu (2007) (for CRRA utility).

In the area of discrete-time models, specifically for the most important class in finance and economics, i.e., GARCH processes, very little has been done regarding analytical portfolio optimization solutions. This is not surprising due to the lack of analytical representations for basic functions of the process, e.g., the conditional probability density. The difficulties in approaching GARCH models in this context can also be observed in Chen (2005), who could solve their rich optimization problem only via scenario generation, i.e., via finite discrete approximations of the return distribution. As mentioned earlier in this dissertation, the picture finally changed with the crucial paper by Heston and Nandi (2000), who introduced the first affine GARCH model (HN-GARCH) for pricing purposes. New affine GARCH models were added to the family in the following years, more notably, the IG-GARCH of Christoffersen et al. (2006), who extended the innovations from Gaussian to inverse Gaussian, hence allowing for negatively skewed and leptokurtic one-period stock returns, while incorporating the HN-GARCH as a limiting case. More recently, Badescu et al. (2019) provided a general definition for the class of affine GARCH models, requiring the conditional bivariate moment generating function (MGF) of the log-asset return and its conditional variance to be exponentially affine in the latter.

The benefit of this property, in the context of an investor maximizing a CRRA power utility function on terminal wealth, was first exploited by Escobar-Anel et al. (2022a) for the HN-GARCH model, and later generalized to the class of general affine GARCHs (see Chapter 3 and Escobar-Anel et al., 2021). In both cases, the authors use a well-established approximation of the self-financing condition (SFC) to produce explicit expressions for the optimal risky allocation, optimal wealth process, and value function.

These developments have excluded the critical aspect of investors' consumption preferences. Not only is consumption a crucial decision-making component for individuals, insurers and pension funds, but its presence can also drastically disrupt optimal risky allocations and wealth behavior. This chapter tackles this problem by considering a large family of affine GARCH models while allowing for consumption and terminal wealth preferences by the investor. We assume a special HARA utility for consumption with a cap on wealth, ensuring that consumption cannot become greater than wealth, which is a drawback of existing solutions in the literature. After the standard approximation of the SFC, we obtain closed-form solutions for optimal consumption, allocation strategy, wealth process, and value function. In particular, the work in this chapter differs from Chen (2005) in that the affine nature of our model allows for an entirely different solution approach without scenario generation.

The main contributions of this chapter are the following:

• We present the first analytical solutions for optimal consumption and allocation in an expected utility setting for affine GARCH models.
- Optimal wealth and consumption processes are shown to also follow affine GARCH models.<sup>1</sup>
- A numerical application to Gaussian and non-Gaussian GARCH, as well as to the homoscedastic model, reveals a significant impact of the presence/absence of consumption on the optimal allocation level. It also shows significant differences in risky allocations compared to homoscedastic cases, with the greater impact coming from heteroscedasticity rather than non-Gaussianity.
- In a study based on real-world market data, over a five-year horizon, the total consumption of investors following a GARCH strategy can be up to 10% higher than with the optimal homoscedastic solution at the same time also achieving 8% more terminal wealth.

The remainder of this chapter is structured as follows: In Section 4.2, we describe the consumption and investment problem and derive a formulation in terms of stochastic control models, starting with the standard approximation of the SFC in 4.2.1, formally introducing the class of affine GARCH models in 4.2.2 and presenting the final optimization problem in 4.2.3. Section 4.3 contains our main results, for a general affine GARCH model in Subsection 4.3.1 and for the two special cases, HN-GARCH and IG-GARCH, in 4.3.2. The numerical analysis in Section 4.4 is divided into a part about a study of the optimal solution to the problem in an IG-GARCH environment (Section 4.4.1) and an assessment of the strategy performance of a Gaussian GARCH model based on real-world data from the S&P 500 stock index in 4.4.2. We conclude in Section 4.5.

# 4.2 Problem Description

This section derives a formulation of our investment problem in the context of affine GARCH models, adapting the approach by Campbell and Viceira (1999) in the presence of consumption.

## 4.2.1 Self-Financing Condition

Let  $V_t$  denote the wealth at time t, and let  $P_t^{(0)}$  and  $P_t^{(1)}$  be the value of the bank account and the risky asset at time t, respectively. With  $\pi_t$  the fraction of wealth invested in the risky asset and describing the consumption at time t by  $C_t$ , we start with the exact relation

$$\frac{V_{t+1}}{V_t - C_t} = \pi_t \frac{P_{t+1}^{(1)}}{P_t^{(1)}} + (1 - \pi_t) \frac{P_{t+1}^{(0)}}{P_t^{(0)}}.$$
(4.1)

Two Taylor expansions of order 2 around 1, using the notation  $\bar{C}_t = C_t/V_t$  for the relative consumption, yield

$$\log\left(\frac{P_{t+1}^{(1)}}{P_t^{(1)}}\right) \approx \frac{P_{t+1}^{(1)}}{P_t^{(1)}} - 1 - \frac{1}{2} \left(\frac{P_{t+1}^{(1)}}{P_t^{(1)}} - 1\right)^2,$$
$$\log\left(\frac{V_{t+1}}{V_t \left(1 - \bar{C}_t\right)}\right) \approx \frac{V_{t+1}}{V_t \left(1 - \bar{C}_t\right)} - 1 - \frac{1}{2} \left(\frac{V_{t+1}}{V_t \left(1 - \bar{C}_t\right)} - 1\right)^2$$

<sup>&</sup>lt;sup>1</sup>This is beneficial in terms of closed-form risk analysis and pricing of derivatives on the optimal portfolio.

#### 4 Optimal Consumption and Investment

We use the variance of the return instead of its square, denoted by  $h_{t+1} = Var\left(\log\left(\frac{P_{t+1}^{(1)}}{P_t^{(1)}}\right)\right)$ and obtain

$$\log\left(\frac{P_{t+1}^{(1)}}{P_t^{(1)}}\right) \approx \frac{P_{t+1}^{(1)}}{P_t^{(1)}} - 1 - \frac{1}{2} \operatorname{Var}\left(\frac{P_{t+1}^{(1)}}{P_t^{(1)}} - 1\right) \approx \frac{P_{t+1}^{(1)}}{P_t^{(1)}} - 1 - \frac{1}{2} h_{t+1},$$

$$\log\left(\frac{V_{t+1}}{V_t \left(1 - \bar{C}_t\right)}\right) \approx \frac{V_{t+1}}{V_t \left(1 - \bar{C}_t\right)} - 1 - \frac{1}{2} \operatorname{Var}\left(\frac{V_{t+1}}{V_t \left(1 - \bar{C}_t\right)} - 1\right)$$

$$\approx \frac{V_{t+1}}{V_t \left(1 - \bar{C}_t\right)} - 1 - \frac{1}{2} \pi_t^2 h_{t+1}.$$

Denoting the continuously compounded interest rate by r > 0, a linear approximation of the logarithm around 1 furthermore leads to

$$r = \log\left(\frac{P_{t+1}^{(0)}}{P_t^{(0)}}\right) \approx \frac{P_{t+1}^{(0)}}{P_t^{(0)}} - 1.$$

All in all, this leads to the following approximation of the SFC:

$$\log\left(\frac{V_{t+1}}{V_t\left(1-\bar{C}_t\right)}\right) + 1 + \frac{1}{2}\pi_t^2 h_{t+1} = \pi_t \left(\log\left(\frac{P_{t+1}^{(1)}}{P_t^{(1)}}\right) + 1 + \frac{1}{2}h_{t+1}\right) + (1-\pi_t)\left(1+r\right). \quad (4.2)$$

Solving for  $W_t \coloneqq \log V_t$ , and defining  $\hat{C}_t \coloneqq \log \left(\frac{V_t}{V_t - C_t}\right)$  as well as  $X_t \coloneqq \log P_t^{(1)}$ , we arrive at

$$W_{t+1} = W_t - \hat{C}_t + \pi_t \left( X_{t+1} - X_t \right) + \left( 1 - \pi_t \right) r + \frac{1}{2} \left( \pi_t - \pi_t^2 \right) h_{t+1}.$$
(4.3)

**Remark 4.1** (Relation to SFC approximation without consumption). Equation (4.2) is identical to the approach by Campbell and Viceira (1999) for a setting without consumption, used only for the reinvested part of wealth, and thus well-supported in the literature. In a homoscedastic setting, the authors combined this with a log-linearization around the mean consumption-wealth ratio, which is not necessary in the above approach. Note that there is no approximation concerning consumption in the procedure leading to (4.3). Thus, examining the quality of this approach and the error of the approximation essentially boils down to the analysis performed in Escobar-Anel et al. (2021) for the case without consumption.

### 4.2.2 Affine GARCH Models and Transition Function

Now assume a general affine GARCH setting as already presented in (3.1), with the log-asset return and its conditional variance being defined as

$$Y_t \coloneqq X_t - X_{t-1} = f_1(h_t, \theta) + \sqrt{h_t \epsilon_t}, \tag{4.4a}$$

$$h_{t+1} = f_2(h_t, \theta) + f_3(h_t, \epsilon_t, \theta), \qquad (4.4b)$$

according to Badescu et al. (2019), with the innovation denoted by  $\epsilon_t$  and having mean zero, and  $\theta$  being a vector of model parameters.<sup>2</sup> In particular, assume that the conditional bivariate

<sup>&</sup>lt;sup>2</sup>Note that we do not require the innovations to be i.i.d., but rely on the structure of the MGF in (4.5) directly to also accommodate the IG-GARCH model. This framework forms a special case of the recent affine multifactor model introduced in Augustyniak et al. (2023).

MGF of the log-asset return and its conditional variance admits the representation

$$\Psi_{(Y_t,h_{t+1})}\left(u,v \mid \mathcal{F}_{t-1}\right) \coloneqq \mathbb{E}\left[\exp\left\{uY_t + vh_{t+1}\right\} \mid \mathcal{F}_{t-1}\right]$$
  
= exp {  $A\left(u,v\right) + B\left(u,v\right) \cdot h_t$  } (4.5)

for some coefficients A and B dependent on the model parameters. On this basis, let  $\mathbb{W} \subset \mathbb{R}$ be the set of possible values for the log wealth, and let  $\mathbb{A}_{\pi}$  and  $\mathbb{A}_{\hat{C}}$  be the sets of admissible values for the portfolio strategy and consumption control  $\hat{C}$ , respectively. We set  $\mathbb{Y}$  the set of possible log return values and define  $\mathbb{H} = (0, \infty)$ . The transition function according to (4.3) thus can be written as  $\mathbb{T} : \mathbb{W} \times \mathbb{H} \times \mathbb{A}_{\pi} \times \mathbb{A}_{\hat{C}} \times \mathbb{Y} \to \mathbb{W} \times \mathbb{H}$ ,

$$\mathbb{T}\left(W,h,\pi,\hat{C},Y\right) = \left(\mathbb{T}_{1}\left(W,h,\pi,\hat{C},Y\right),\mathbb{T}_{2}\left(h,Y\right)\right)$$
$$= \left(W - \hat{C} + \pi Y + (1-\pi)r + \frac{1}{2}\left(\pi - \pi^{2}\right)h, \qquad (4.6)$$
$$f_{2}\left(h,\theta\right) + f_{3}\left(h,\frac{Y - f_{1}\left(h,\theta\right)}{\sqrt{h}},\theta\right)\right).$$

Note that  $\mathbb{W}, \mathbb{H}, \mathbb{A}_{\pi}$  and  $\mathbb{A}_{\hat{C}}$  are intervals in  $\mathbb{R}$  and thus convex sets.

### 4.2.3 Optimization Problem

We suppose that the investor derives utility from consumption at each time point  $t = 0, \ldots, T-1$ , and utility from terminal wealth at time T. In particular, for some constant parameter  $\bar{\gamma} > 1$ , we consider utility functions for consumption of the form  $u_t(C_t) = -a_t^{\bar{\gamma}} [(V_t - C_t)^+]^{\bar{\gamma}}$ , with some time-dependent  $a_t > 0$  and  $t \in \{0, \ldots, T-1\}$ . This utility is concave, increasing in  $C_t$  on the interval  $(0, V_t)$ , and flat on  $[V_t, \infty)$ , implying that the current wealth will never be exceeded by consumption. Note that an increase in consumption at the same time negatively affects potential future utilities, see (4.3). This choice of utility is popularly known as HARA, see Merton (1971), and given in general form by

$$U_H(C) = \frac{1 - \gamma_H}{\gamma_H} \left(\frac{\beta_H \cdot C}{1 - \gamma_H} + \eta_H\right)^{\gamma_H}.$$

We can match Merton's and our notation as follows:

$$\gamma_H = \bar{\gamma}, \qquad \eta_H = \beta_H \cdot \frac{V_t}{\gamma_H - 1}, \qquad \beta_H = \frac{a}{b},$$

with

$$a = a_t(\gamma_H - 1), \qquad b = \left(\frac{\gamma_H - 1}{\gamma_H}\right)^{\frac{1}{\gamma_H}}$$

In the sequel, we will work with the parameter  $a_t := \bar{a}/V_t$ , for some constant  $\bar{a} > 0$ , and define

$$U_t(c) = \begin{cases} u_t(c), & c \in [0, \infty), \\ -\infty, & \text{otherwise.} \end{cases}$$

In terms of the Arrow-Pratt measure for absolute risk aversion, we thus obtain for  $C_t \in (0, V_t)$ :

$$\mathcal{A}_t\left(C_t\right) = -\frac{U_t''\left(C_t\right)}{U_t'\left(C_t\right)} = \frac{\bar{\gamma} - 1}{V_t - C_t},$$

### 4 Optimal Consumption and Investment

supporting the well-known feature of absolute risk aversion decreasing in wealth. Note that  $\mathcal{A}_t(C_t)$  is increasing in  $\bar{\gamma}$ . In terms of the relative risk aversion, we obtain

$$\mathcal{R}_{t}(C_{t}) = -C_{t} \frac{U_{t}''(C_{t})}{U_{t}'(C_{t})} = \frac{C_{t}}{V_{t} - C_{t}} \cdot (\bar{\gamma} - 1) = \frac{\bar{C}_{t}}{1 - \bar{C}_{t}} \cdot (\bar{\gamma} - 1).$$

Furthermore, on  $[0, V_t]$ , we can transform  $U_t(C_t)$  to be a function of our control on consumption  $\hat{C}_t$  via

$$U_t (C_t) = -a_t^{\bar{\gamma}} \left[ (V_t - C_t)^+ \right]^{\bar{\gamma}}$$
  
$$= -\bar{a}^{\bar{\gamma}} \left( \frac{V_t - C_t}{V_t} \right)^{\bar{\gamma}}$$
  
$$= -\bar{a}^{\bar{\gamma}} \exp\left\{ -\bar{\gamma} \log\left( \frac{V_t}{V_t - C_t} \right) \right\}$$
  
$$= -\bar{a}^{\bar{\gamma}} \exp\left\{ -\bar{\gamma} \hat{C}_t \right\}, \qquad (4.8)$$

which means that we are maximizing an exponential utility function w.r.t.  $\hat{C}_t \in [0, \infty)$ . For convenience, we denote  $U_t$  as a function of  $\hat{C}_t$  as in (4.8) in the sequel. At time T, utility only depends on the final wealth  $V_T$ , following a CRRA power utility function  $U_T(v) = \frac{1}{\gamma}v^{\gamma}$  for some  $\gamma < 1$ . This implies that the investor's absolute risk aversion  $\mathcal{A}_T(v) = -\frac{U''_T(v)}{U'_T(v)} = \frac{1-\gamma}{v}$  is decreasing in  $\gamma$ . The Arrow-Pratt measure for relative risk aversion yields  $\mathcal{R}_T(v) \equiv \mathcal{R}_T = 1-\gamma$ .

Following the strict separation between evaluating the utility of consumption at all intermediate time points and measuring the utility from terminal wealth via the power utility, we intend to solve:

$$\max_{\left\{\left(\pi_{s},\hat{C}_{s}\right)\right\}_{t}^{T-1}} \mathbb{E}\left[\sum_{s=t}^{T-1} \bar{\beta}^{s-t} \cdot U_{s}\left(C_{s}\right) + \bar{\beta}^{T-t} \cdot U_{T}\left(V_{T}\right) \mid \mathcal{F}_{t}\right],$$

where  $\bar{\beta} \in [0, 1]$  is the corresponding factor for intertemporal substitution. We work with the log wealth and, as outlined in Chapter 3, apply the value iteration in Theorem 2.15, using the SFC approximation (4.3). This leads to the following value function:

$$\begin{cases} \phi_T (W_T, h_{T+1}) \coloneqq \phi_T (W_T) = U_T (e^{W_T}) = \frac{1}{\gamma} \exp \{\gamma W_T\} \\ \phi_t (W_t, h_{t+1}) = \max_{\pi_t, \hat{C}_t} \left\{ -\bar{a}^{\bar{\gamma}} \exp \{-\bar{\gamma} \hat{C}_t\} + \bar{\beta} \cdot \mathbb{E} \left[ \phi_{t+1} (\mathbb{T} (\Xi_t)) \mid \mathcal{F}_t \right] \right\}, \\ t = 0, \dots, T-1. \end{cases}$$

$$(4.9)$$

Here,  $\Xi_t = (W_t, h_{t+1}, \pi_t, \hat{C}_t, Y_{t+1}).$ 

**Remark 4.2** (On the choice of utility functions). Concerning the optimization setting introduced above, note the following:

(1) For  $\bar{C}_t = C_t/V_t$  close to zero, the first-order Taylor approximation

$$\hat{C}_t = -\log\left(1 - \bar{C}_t\right) \approx \bar{C}_t$$

works well. Plugging this into the utility for consumption leads to

$$U_t(C_t) = -\bar{a}^{\bar{\gamma}} \exp\left\{-\bar{\gamma}\hat{C}_t\right\} \approx -\bar{a}^{\bar{\gamma}} \exp\left\{-\bar{\gamma}\bar{C}_t\right\} = -\bar{a}^{\bar{\gamma}} \exp\left\{-\frac{\bar{\gamma}}{V_t}C_t\right\}.$$

Calculating the Arrow-Pratt measure of absolute risk aversion for the expression on the right-hand side yields  $\mathcal{A}_t(C_t) = \bar{\gamma}/V_t$ , again supporting the idea of risk aversion decreasing with the level of wealth. Having the same relative risk aversion as for the CRRA function measuring utility from terminal wealth requires  $\mathcal{R}_t(C_t) = C_t \cdot \mathcal{A}_t(C_t) \approx \bar{C}_t \bar{\gamma} \stackrel{!}{=} 1 - \gamma$ .<sup>3</sup>

- (2) Dropping the convention that  $C_T = 0$  leads to an optimization problem with one additional variable at time T the proportion of wealth to consume at maturity. The question of how much to consume (value  $C_T$ , measured via utility for consumption) and how much to leave untouched ( $V_T C_T$ , measured via power utility), will essentially be determined by the exact shape of the two utility functions.
- (3) Using power utility also for consumption at time  $t \in \{0, ..., T-1\}$  leads to absolute risk aversion being inversely proportional to  $C_t$  and disconnected from wealth. Furthermore, there is no inherent bound preventing consumption from exceeding the current wealth. The approach above solves this puzzle by connecting consumption directly to wealth.

# 4.3 Solution to the Portfolio Optimization Problem

## 4.3.1 Main Results

This section presents our main theoretical results, deriving an optimal strategy for the dynamic optimization problem in Section 4.2.3. For the sake of readability, the proofs are to be found in Appendix C.1.

In order to prove our main result in a general context, a structural result about the concavity of value functions in stochastic control models is needed, see Proposition 2.21 in the Mathematical Preliminaries. In particular, we refer to our model as an stochastic control model (SCM) with arbitrary disturbances, based on Definition 2.12.

**Remark 4.3.** We can relax the requirement of *i.i.d.* disturbances to accommodate the IG-GARCH model, asking that the distribution of innovations allow for the exponentially affine representation of the conditional bivariate generating function of the log-asset return and its conditional variance instead, since all necessary properties of the model are still given in each individual step.

**Theorem 4.4.** In the environment described above, assume that the log asset price of the risky asset follows an affine GARCH model, and that  $\gamma < 0$  and  $\bar{\gamma} > 1$ . The value function can be written as

$$\phi_{t}(W_{t}, h_{t+1}) = -\bar{a}^{\bar{\gamma}} \exp\left\{-\bar{\gamma}\hat{C}_{t}^{*}\right\} - \mathbb{E}_{t}\left[\sum_{s=t+1}^{T-1} \bar{\beta}^{s-t}\bar{a}^{\bar{\gamma}} \exp\left\{-\bar{\gamma}\hat{C}_{s}^{*}\right\}\right] + \frac{\bar{\beta}^{T-t}}{\gamma} \exp\left\{D_{t,T}(\pi_{t}^{*}) + E_{t,T}^{\hat{C}}\hat{C}_{t}^{*} + E_{t,T}^{W}W_{t} + E_{t,T}^{h}(\pi_{t}^{*})h_{t+1}\right\},$$
(4.10)

<sup>&</sup>lt;sup>3</sup>The occurrence of relative consumption in this equation can be solved via considering the initial time point t = 0 and replacing the level of consumption implied by a homoscedastic Gaussian model in our setting with the solution by Merton (1969). Denoting Merton's solution at time t = 0, discretized for step length  $\Delta t$  equal to one day, by  $C^{(M)}$ , we set  $\bar{C}_0 = \frac{C^{(M)}}{v_0}$ , where  $v_0$  denotes the investor's initial wealth. For more details, see Section 4.4.

with the optimal solution  $\left\{\left(\pi_t^*, \hat{C}_t^*\right)\right\}_t$  and coefficients  $D, E^W, E^{\hat{C}}, E^h$  as defined in Equation (C.2) in the proof. The optimal relative consumption  $\hat{C}_t^*$  is affine in the log wealth and the conditional variance, and admits the representation

$$\hat{C}_{t}^{*} = P_{t,T}\left(\pi_{t}^{*}\right) + Q_{t,T} \cdot W_{t} + R_{t,T}\left(\pi_{t}^{*}\right) \cdot h_{t+1},$$
(4.11)

for coefficients P, Q, R as defined in (C.6). By denoting the coefficients of the bivariate MGF of the log-asset return and its conditional variance A and B, the optimal risky allocation  $\pi_t^*$  satisfies

$$\frac{\partial}{\partial u} A \left( E_{t,T}^W \pi_t^*, \ E_{t+1,T}^h \left( \pi_{t+1}^* \right) + E_{t+1,T}^{\hat{C}} R_{t+1,T} \left( \pi_{t+1}^* \right) \right) = r, \tag{4.12a}$$

$$\frac{\partial}{\partial u} B\left(E_{t,T}^{W} \pi_{t}^{*}, E_{t+1,T}^{h}\left(\pi_{t+1}^{*}\right) + E_{t+1,T}^{\hat{C}} R_{t+1,T}\left(\pi_{t+1}^{*}\right)\right) = \pi_{t}^{*} - \frac{1}{2}.$$
(4.12b)

Remark 4.5 (On Theorem 4.4).

- (1) The conditional expectation in the value function (4.10) can be solved recursively using (4.11), the SFC approximation (4.3) and the conditional bivariate MGF of the log-asset return and its conditional variance in the underlying affine GARCH model.
- (2) Note that the (first-order) optimality equations for  $\pi_t^*$  are the same as in the CRRA case without consumption, although the arguments of the coefficients A and B are different, of course.

Using the approximation of the SFC (4.3) and the conditional bivariate generating function of the log-asset return and its conditional variance in the underlying affine GARCH model analogously to the proof of Theorem 4.4, we can establish that the optimal log-wealth process actually is an affine GARCH process itself.

**Corollary 4.6** (Affine GARCH process for optimal log wealth). The optimal log-wealth process  $\{W_t\}_t$  follows an affine GARCH process. Specifically,

$$\Psi_{\left(W_{t},\left(\pi_{t}^{*}\right)^{2}h_{t+1}\right)}\left(u,v\mid\mathcal{F}_{t-1}\right)$$

$$=\exp\left\{A\left(u\pi_{t-1}^{*},v\left(\pi_{t}^{*}\right)^{2}\right)-uP_{t-1}\left(\pi_{t-1}^{*}\right)+\left(1-\pi_{t-1}^{*}\right)ur+u\left(1-Q_{t-1}\right)W_{t-1}\right.$$

$$\left.+\frac{1}{\left(\pi_{t-1}^{*}\right)^{2}}\left[B\left(u\pi_{t-1}^{*},v\left(\pi_{t}^{*}\right)^{2}\right)-uR_{t-1}\left(\pi_{t-1}^{*}\right)+\frac{u}{2}\left(\pi_{t-1}^{*}-\left(\pi_{t-1}^{*}\right)^{2}\right)\right]\cdot\left(\pi_{t-1}^{*}\right)^{2}h_{t}\right\}.$$

$$(4.13)$$

Furthermore, the optimal investment strategy  $\pi_t^*$  does not depend on the conditional variance or the wealth.

**Remark 4.7.** Since the optimal control on consumption,  $\hat{C}_t^*$ , is affine in  $W_t$  and  $h_{t+1}$ , the MGF of  $\hat{C}_t^*$  is also exponentially affine in the log wealth and the conditional variance of the log return.

Using Corollary 4.6, and the MGF (4.13) in particular, we can now establish a shorter form of the value function (4.10).

**Corollary 4.8.** For  $t \in \{0, ..., T-1\}$ , the value function (4.10) can be written as

$$\phi_t \left( W_t, h_{t+1} \right) = -\exp\left\{ \log \left( T - t + 1 \right) + E_{t,T}^W \left( 1 - Q_{t,T} \right) W_t + \left( E_{t,T}^h \left( \pi_t^* \right) - E_{t,T}^W R_{t,T} \left( \pi_t^* \right) \right) h_{t+1} \right\} - \sum_{\tau=t}^T \exp\left\{ \tilde{D}_{t,\tau,T} \left( \{ \pi_k \}_t^\tau \right) \right\},$$
(4.14)

for deterministic coefficients  $D_{t,\tau,T}$  with  $\tau \in \{t, \ldots, T\}$ .

We omit the dependence of  $D_{t,\tau,T}$  on the optimal strategy for readability.

**Remark 4.9** (Wealth-equivalent loss). Given the results of Corollary 4.8, we can introduce a concept for measuring the wealth-equivalent loss (WEL) arising from following a suboptimal strategy instead of the optimum according to Theorem 4.4. We assume the same transition law (4.6) for both the optimal and the suboptimal solution, dependent on  $\left\{ \left(\pi_t^s, \hat{C}_t^s\right) \right\}_t$  and  $\left\{ \left(\pi_t^s, \hat{C}_t^s\right) \right\}_t$ , respectively. In particular, the value function for the suboptimal variant is given by

$$\phi_t^s\left(W_t, h_{t+1}\right) = -\bar{a}^{\bar{\gamma}} \exp\left\{-\bar{\gamma}\hat{C}_t^s\right\} + \bar{\beta} \cdot \mathbb{E}_t\left[\phi_{t+1}^s\left(\mathbb{T}\left(W_t, h_{t+1}, \pi_t^s, \hat{C}_t^s, Y_{t+1}\right)\right)\right]$$

We now define the wealth-equivalent loss from following the suboptimal strategy instead of the optimal as the solution  $L_t^s$  to the equation

$$\phi_t \left( \log \left( V_t \left( 1 - L_t^s \right) \right), h_{t+1} \right) = \phi_t^s \left( \log \left( V_t \right), h_{t+1} \right).$$
(4.15)

If the suboptimal allocation is deterministic and the suboptimal control on consumption is affine in the log wealth and the conditional variance of the log-asset return, by Corollary 4.8, (4.15) is equivalent to

$$\exp\left\{E_{t,T}^{W}\left(1-Q_{t,T}\right)\log\left(V_{t}\right)+\left(E_{t,T}^{h}\left(\pi_{t}^{s}\right)-E_{t,T}^{W}R_{t,T}\left(\pi_{t}^{s}\right)\right)h_{t+1}\right\}\\-\exp\left\{E_{t,T}^{W}\left(1-Q_{t,T}\right)\log\left(V_{t}\left(1-L_{t}^{s}\right)\right)+\left(E_{t,T}^{h}\left(\pi_{t}^{*}\right)-E_{t,T}^{W}R_{t,T}\left(\pi_{t}^{*}\right)\right)h_{t+1}\right\}\\=\sum_{\tau=t}^{T}\left(\exp\left\{\tilde{D}_{t,\tau,T}^{*}-\log\left(T-t+1\right)\right\}-\exp\left\{\tilde{D}_{t,\tau,T}^{s}-\log\left(T-t+1\right)\right\}\right).$$
(4.16)

Solving for  $L_t^s$  leads to a closed-form expression for the wealth-equivalent loss. The above requirement in particular accommodates all affine GARCH models as potential candidates for suboptimal choices.

Lastly, we note that the previously tackled investment problem for an investor deriving utility from terminal wealth only is nested in the more general environment described here.

**Corollary 4.10** (Special case without consumption). The setting in this document nests the setup without consumption, i.e., with an investor deriving utility from terminal wealth only, as presented in Chapter 3 (Escobar-Anel et al., 2022a, 2021), as a limiting case for the parameter choice  $\bar{a} = 1$  and letting  $\bar{\gamma} \to \infty$ .

### 4.3.2 Examples of Applications to Affine GARCH Models

### **HN-GARCH** Model

Assume that the log asset price and its conditional variance follow an affine GARCH model as of Heston and Nandi (2000), and as already specified in (3.17), such that the set of parameters is given by  $\theta = (\lambda, \omega, \beta, \alpha, \rho), \{\epsilon_t\}_t$  is a sequence of standard normal innovations, and (4.4) is of the form

$$Y_t = X_t - X_{t-1} = r + \lambda h_t + \sqrt{h_t} \epsilon_t, \qquad (4.17a)$$

$$h_{t+1} = \omega + \beta h_t + \alpha \left(\epsilon_t - \rho \sqrt{h_t}\right)^2.$$
(4.17b)

As presented in Appendix B.1, this implies in particular that for  $t = 0, \ldots, T - 1$ ,

$$\Psi_{(Y_{t+1},h_{t+2})}(u,v \mid \mathcal{F}_t) \coloneqq \mathbb{E}\left[\exp\left\{u \cdot Y_{t+1} + v \cdot h_{t+2}\right\} \mid \mathcal{F}_t\right]$$
  
=  $\exp\left\{A(u,v) + B(u,v) \cdot h_{t+1}\right\}$  (4.18)

with the two coefficients A and B given by

$$A(u,v) = ur + v\omega - \frac{1}{2}\log(1 - 2v\alpha),$$
(4.19a)

$$B(u,v) = u\lambda + v\left(\alpha\rho^2 + \beta\right) + \frac{(u - 2v\alpha\rho)^2}{2(1 - 2v\alpha)}.$$
(4.19b)

Therefore, the HN-GARCH model is an affine GARCH model and yields a special case of our main result. Using that the equations in (4.12) have the same structure as in the case without consumption, we obtain the following corollary:

**Corollary 4.11** (HN-GARCH). In the situation of Theorem 4.4, if the underlying log asset price process follows an HN-GARCH model and is given by (4.17), the value function and the optimal control on consumption are given by (4.10) and (4.11), respectively, using the explicit expressions (4.19) for the coefficients A and B. Concerning the optimal allocation, (4.19) reduces (4.12) to

$$\pi_t^* = \frac{(\lambda + 1/2) \left(1 - 2v\alpha\right) - 2v\alpha\rho}{1 - E_{t,T}^W - 2v\alpha},\tag{4.20}$$

using the abbreviation

$$v \coloneqq E_{t+1,T}^{h}\left(\pi_{t+1}^{*}\right) + E_{t+1,T}^{\hat{C}}R_{t+1,T}\left(\pi_{t+1}^{*}\right).$$
(4.21)

*Proof.* See Appendix C.1.

### Remark 4.12 (On Corollary 4.11).

(1) A common way to look at an optimal allocation like (4.20) is to split the solution into two terms. A myopic term, independent of the horizon and related to the celebrated solution by Merton (1973), and a time-dependent term coming from hedging the extra sources of randomness, see Kraft (2005) for a popular continuous-time example with stochastic volatility without consumption. The same exercise for the HN-GARCH without consumption, which is a special case of our equation (4.20), can be found in Escobar-Anel et al. (2022a).

(2) In the presence of consumption, our Equation (4.20) is different from Merton's solution even in the absence of GARCH (i.e., in the homoscedastic case, where  $\alpha = \beta = 0$ ). More importantly, in such a scenario, our optimal allocation is time-dependent and non-myopic (i.e.,  $\pi_t^* = \frac{\lambda + 1/2}{1 - \gamma \bar{\gamma} (\bar{\gamma} - (T - t - 1)\gamma)^{-1}}$  as opposed to  $\pi_t^M = \frac{\lambda + 1/2}{1 - \gamma}$  for Merton), therefore the split into a myopic and a non-myopic term is no longer feasible.

### **IG-GARCH** Model

Assume that the log asset price and its conditional variance follow the IG-GARCH model (cf. Christoffersen et al., 2006), where the set of parameters is given by  $\theta = (\nu, \eta, w, a, b, c)$ , having  $\eta < 0$ . The log return and its conditional variance, as presented in (3.18), are given by

$$Y_t = X_t - X_{t-1} = r + \nu h_t + \eta y_t, \tag{4.22a}$$

$$h_{t+1} = w + bh_t + cy_t + a\frac{h_t^2}{y_t},$$
(4.22b)

where  $\{y_t\}_t$  is a sequence of random variables with inverse Gaussian distribution and single parameter  $\delta(t) = \delta_t = h_t/\eta^2$ . As stated by Christoffersen et al. (2006), the MGF is again given by (4.18) with the coefficients A and B defined as (see Appendix B.2)

$$A(u,v) = ur + vw - \frac{1}{2}\log(1 - 2va\eta^4), \qquad (4.23a)$$

$$B(u,v) = vb + u\nu + \frac{1}{\eta^2} \left( 1 - \sqrt{(1 - 2va\eta^4)(1 - 2u\eta - 2vc)} \right).$$
(4.23b)

This matches the characterization of general affine GARCH models as of Section 4.2.2 and, together with the fact that the structure of the optimality equations (4.12) for the optimal risky allocation is the same as in the case without consumption, yields the following corollary:

**Corollary 4.13** (IG-GARCH). In the situation of Theorem 4.4, if the underlying log asset price process follows an HN-GARCH model and is given by (4.22), the value function and the optimal control on consumption are given by (4.10) and (4.11), respectively, using the explicit expressions (4.23) for the coefficients A and B. Concerning the optimal allocation, plugging in (4.23) reduces (4.12) to

$$\nu + \frac{\sqrt{1 - 2va\eta^4}}{\eta\sqrt{1 - 2\eta E_{t,T}^W \pi_t^* - 2vc}} = \pi_t^* - \frac{1}{2},\tag{4.24}$$

where v is again defined as in (4.21).

*Proof.* See Appendix C.1.

Remark 4.14 (On Corollary 4.13).

(1) Note that Equation (4.24) is not linear in  $\pi_t^*$ . Instead, separating radicals and squaring the equation yields a cubic equation. The optimal portfolio strategy  $\pi_t^*$  thus is a root of a polynomial of degree 3, the discriminant of which indicates the number of real solutions, see Section 3.4. In the case of a CRRA utility without consumption, we could show numerically that there is exactly one real solution (cf. Escobar-Anel et al., 2021).

### 4 Optimal Consumption and Investment

(2) Given the connection between IG- and HN-GARCH models, we can show numerically, that the real solution to equation (4.24) converges to the solution of the HN-GARCH model in equation (4.20). To this end, we have to let η → 0 according to the parametrization in Equation (6) in Christoffersen et al. (2006), see Figure 4.4 and the corresponding comments.

# 4.4 Numerical Analysis

The purpose of this section is to investigate the behavior of the solution to our optimization problem numerically, based on the special cases of the IG-GARCH and the HN-GARCH model from Section 4.3.2, and on the homoscedastic case. Two different types of analysis are provided: The first focuses on the sensitivity of the optimal allocation and consumption strategy w.r.t. the model and investment parameters, and studies in particular the impact of conditional skewness and kurtosis on consumption and risky allocation. The second uses real-world data from the S&P 500 stock index to compare the strategy performance over two separate periods of five years each.

## 4.4.1 Optimal Solution Analysis

This subsection shows how the optimal solution to our optimization problem as presented in Section 4.3, implemented for the special cases of an HN-GARCH model, an IG-GARCH model, and a homoscedastic model (HS, constant variance), depends on important model and investment parameters. In particular, provided that the inverse Gaussian distribution embeds the HN and HS models while allowing for negatively skewed and leptokurtic innovations for the log-asset return, we also explore the impact of skewness and kurtosis on our solution. We will focus on the sensitivity of the optimal risky allocation and of the optimal consumption process, measured in relative consumption  $\bar{C}_t^* = C_t^*/V_t = 1 - \exp\left\{-\hat{C}_t^*\right\}$ . The analysis is conducted using the maximum likelihood estimates (MLEs) from Babaoğlu et al. (2018) as parameter values (see Table 4.1), an additional analysis with the earlier estimates from Christoffersen et al. (2006) can be found in the appendix. If not stated otherwise, the time horizon is five years with 252 trading days each, the interest rate is r = 0.01/252, the parameter for intertemporal substitution is  $\bar{\beta} = 1$  and the initial wealth is  $v_0 = 1$ . In view of Corollary 4.10, we work with  $\bar{a} = 1$ . Concerning the investor's risk aversion, the default value for utility from terminal wealth is  $\gamma = -4$ , corresponding to  $\mathcal{R}_T = 5$  as the level of relative risk aversion. For the parameter  $\bar{\gamma}$ , we refer to Remark 4.2. The proxy for relative consumption needed to obtain the same level of risk aversion as at time T is derived via matching the consumption of a homoscedastic Gaussian model in our setting with the optimal solution proposed by Merton (1969) at the initial time point t = 0. Plugging in the corresponding value for relative consumption amounts to  $\bar{\gamma} = 7.559 \times 10^3$ .

Figure 4.2 shows how the optimal initial solution depends on the investor's time horizon and how solutions to different special cases of our general model compare. The plots show the optimal relative consumption and the optimal risky allocation, respectively, for the IG-GARCH model, for the HN-GARCH model, and for a homoscedastic Gaussian model. The MLEs for the parameter values are taken from Babaoğlu et al. (2018), and all models are calibrated to the same data set. For the optimal risky allocation, we also plot the solution without consumption, i.e., for an investor deriving utility from terminal wealth only. Given that this case is nested in our model as the limit for  $\bar{\gamma} \to \infty$  (see Corollary 4.10), we calculate these solutions via setting this parameter to a very large value and leaving the rest unchanged. Here, we use  $\bar{\gamma} = 1 \times 10^8$ .

Table 4.1:	Values for the parameters used for the IG-GARCH, the HN-GARCH model and the ho-
	moscedastic Gaussian - consisting of the MLEs as of Babaoğlu et al. (2018) for all param-
	eters. For the IG-GARCH model, the value of $\nu$ implies that the market price of risk is
	$\nu + \eta^{-1} \approx 1.16.$

Panel A: IG-GARCH		Panel B: HN-GARCH		Panel C	Panel C: Hom. Gauss.	
Param.	Values	Param.	Values	Param.	Values	
ν	$1.747 \times 10^3$	$\lambda$	1.100	$\lambda$	0.78	
η	$-5.729 \times 10^{-4}$	ω	$-1.396 \times 10^{-6}$	ω	$1.373 \times 10^{-4}$	
w	$-1.469 \times 10^{-6}$	β	$9.000 \times 10^{-1}$			
a	$3.190  imes 10^7$	α	$3.761\times 10^{-6}$			
b	$-2.182 \times 10^1$	ρ	$1.457 \times 10^2$			
с	$4.047 \times 10^{-6}$					

Figure 4.2a shows that the specific model for the log-asset return within the class of affine GARCH models is almost irrelevant for the optimal relative consumption – the difference is of order  $1 \times 10^{-6}$ . In this context, we note that, from the proof of Theorem 4.4, the coefficient  $Q_{t,T}$  for the affine relation of  $\hat{C}_t^*$  and  $W_t$  is independent of the specific model parameters. This picture is changed when we look at the optimal  $\pi_0^*$ : In Figure 4.2b, we observe a clear difference for the optimal risky allocation, with the highest fraction invested in the risky asset when the IG-GARCH model is used, directly followed by the Gaussian HN-GARCH model. With a larger difference to the heteroscedastic models, a homoscedastic investor invests the least fraction of wealth in the risky asset. Note that a similar pattern has been observed without consumption by Escobar-Anel et al. (2021). In general, the green lines reproducing the solutions without consumption show that the relative risky allocation is increased significantly over time as consumption is added to the model.

Since the IG-GARCH contains the HN-GARCH as a special case, we continue the analysis with the IG-GARCH model. Figure 4.3a shows the impact of the risk aversion parameters – with  $\gamma$ corresponding to the utility derived from terminal wealth (left plot) and  $\bar{\gamma}$  corresponding to the utility derived from consumption (right plot) – on the initial value of relative consumption. We use  $\gamma \in [-6, -0.1]$  and  $\bar{\gamma} \in [5 \times 10^2, 1 \times 10^4]$  to cover reasonable risk aversion levels for terminal wealth and consumption, respectively. Our default parameter choice  $\gamma = -4$  corresponds to a risk aversion level of  $\mathcal{R}_T = 5$ , and the default value  $\bar{\gamma} = 7.559 \times 10^3$  is chosen according to Remark 4.2 based on this value. In general, risk aversion for intermediate consumption influences both relative consumption and the relative risky investment (right plots in Figure 4.3a and 4.3b), but the risk aversion for utility from terminal wealth only has a minor effect on consumption (left plot in Figure 4.3a); the decrease in  $\bar{C}_0^*$  is much larger as  $\mathcal{R}_0$  ( $\bar{C}_0^*$ ) increases. The left plot in Figure 4.3b shows that as the investor's risk aversion concerning utility derived from terminal wealth decreases ( $\mathcal{R}_T \to 0$ ), the optimal risky allocation increases. The impact is similar to the one caused by a decreasing risk aversion for consumption in the right plot in Figure 4.3b. Again, the optimal risky allocation increases as the investor's risk aversion decreases.

Given the presence of non-Gaussianity and heteroscedasticity in the model, we study the impact of these two features on the solution to our optimization problem next. Taking advantage of the



(a) Optimal initial relative consumption, dependent on the time horizon.



(b) Optimal initial relative risky allocation, dependent on the time horizon.

Figure 4.2: Values of the optimal initial strategy at time t = 0, dependent on the time horizon. The plots include an IG-GARCH, an HN-GARCH and a homoscedastic Gaussian, all fitted to the same data set. The solutions without consumption are obtained via setting  $\bar{\gamma} = 1 \times 10^8$  to approximate the case without consumption. One year is assumed to have 252 trading days. We used the parameter MLEs of Babaoğlu et al. (2018).

more general nature of the IG-GARCH among the three models, we transform this analysis into measuring the impact of skewness and kurtosis on the optimal solution while assuming the first two unconditional moments remain the same. As presented by Christoffersen et al. (2006), the IG-GARCH model allows for negatively skewed and leptokurtic log-asset returns. The one-step skewness S and excess kurtosis K are calculated dependent on the value of the parameter  $\eta$  via the formulas

$$\mathcal{S}_{t}[Y_{t+1}] = 3\eta \left(h_{t+1}\right)^{-1/2}, \qquad \mathcal{K}_{t}[Y_{t+1}] = 15\eta^{2} \left(h_{t+1}\right)^{-1}.$$
(4.25)

Since changes in  $\eta$  also influence the expected log-asset return and its variance, we apply the approach of Escobar-Anel et al. (2021) and adjust the remaining parameters in the IG-GARCH model with changes in  $\eta$  such that the first two moments are kept constant. This isolates the impact of negatively skewed and leptokurtic log-asset returns on the optimal strategy. The right plot in Figure 4.4 indicates that relative consumption decreases as negative skewness and excess kurtosis increase ( $\eta$  decreases), but the scale also shows that the impact is very little. The left plot shows a similar picture for the optimal risky allocation; we refer to Escobar-Anel et al. (2021) for more detailed analysis in the context of a pure investment problem and note



(a) Optimal initial relative consumption, dependent on the investor's risk aversion concerning utility from terminal wealth (left plot) and from consumption (right plot).



(b) Optimal initial relative risky allocation, dependent on the investor's risk aversion concerning utility from terminal wealth (left plot) and from consumption (right plot).

Figure 4.3: Values of the optimal initial relative consumption and optimal initial risky allocation, dependent on relative risk aversion. Relative risk aversion for terminal wealth in the left plots is governed by the parameter  $\gamma$ , and relative risk aversion for consumption in the right plots is modified via  $\bar{\gamma}$ , see Remark 4.2. The time horizon is five years.

that the direction of the impact of  $\eta$  on  $\pi_0^*$  is analogous if the model is extended by allowing for consumption. For both figures, we also insert the limiting HN-GARCH solution and the embedded HS solution, all with the same first two unconditional moments.

This figure conveys one of the main conclusions of this chapter, which is that the most consequential feature between heteroscedasticity and non-Gaussianity is the former. In light of this finding, we pick the HN-GARCH and the homoscedastic model for an application on real data in the next section.

Further parameters of the IG-GARCH model have a small impact of the order of  $1 \times 10^{-5}$  (0.001%) on the optimal relative consumption, see Appendix C.2.1. However, it is essential to note that changes in the parameters w and r do change the solution, as opposed to the special case without consumption.



Figure 4.4: Values of the optimal initial risky allocation and relative consumption dependent on the parameter  $\eta$ . The time horizon is five years. The color bar on the right-hand side indicates the corresponding values for the skewness and excess kurtosis of log-asset returns. The vertical dashed line marks the original MLE for  $\eta$  in the considered parameter set. The first two moments are kept constant under changes in  $\eta$ . The blue and red dots mark the solutions for HN-GARCH and homoscedastic models with the same first two unconditional moments.

### 4.4.2 Data-based Performance Analysis

This section presents a data-based analysis of the strategy performance, using parameters estimated from S&P 500 return values in two periods from January 3, 2013, through December 30, 2017, and from January 3, 2018, through December 30, 2022. In our study, we focus on the HN-GARCH model in comparison to the embedded homoscedastic variant. As explained before in the context of Figure 4.4, skewness and kurtosis have very little effect on risky allocation and consumption. At the same time, the impact of heteroscedasticity is more significant, particularly on the optimal allocation. Therefore, we select the HN-GARCH and the homoscedastic as the models for the comparison.

For each of the two five-year periods, we estimate the HN-GARCH and the homoscedastic parameters using the returns-only estimation of Escobar-Anel et al. (2022b) for the path of S&P 500 log returns  $\{x_t\}_{t=1,...,T}$ . Given the vector of parameters  $\theta = (\lambda, \omega, \alpha, \beta, \rho)$ , with  $\alpha = \beta = 0$  in the homoscedastic case, and the log return  $x_{t-1}$ , the law for  $X_t$  is

$$X_t \mid (x_{t-1}, \theta) \sim \mathcal{N} \left( x_{t-1} + r + \lambda h_t, h_t \right). \tag{4.26}$$

The joint probability density function (PDF)  $f_{X_1,...,X_T}$  of the log returns can be expressed via conditioning on the previous values:

$$f_{X_1,\dots,X_T}(x_1,\dots,x_T) = f_{X_1}(x_1) \cdot f_{X_2|X_1}(x_2) \cdot \dots \cdot f_{X_T|X_{T-1}}(x_T).$$
(4.27)

Combining (4.26) and (4.27) with the PDF of a normally distributed random variable and subsequently taking the logarithm leads to the log-likelihood function

$$l_T(x_1, \dots, x_T \mid \theta) = -\frac{1}{2} \sum_{t=1}^T \left( \log(2\pi) + \log(h_t) + \epsilon_t^2 \right),$$
(4.28a)

where

$$\epsilon_t = \frac{x_t - x_{t-1} - r - \lambda h_t}{\sqrt{h_t}}.$$
(4.28b)

The estimates  $\hat{\theta}$  for  $\theta$  are given by  $\hat{\theta} = \arg \max_{\theta} l_T(x_1, \ldots, x_T \mid \theta)$ . The standard errors of the estimates are calculated via the Fisher information matrix (see Equation 2.9). The combined results with standard errors are provided in Table 4.2.

Param.	HN-GARCH	Homoscedastic		
	2013-17	2018-22	2013-17	2018-22
λ	8.97	1.79	8.59	1.47
	(3.90)	(2.13)	(3.81)	(2.08)
ω	$2.07\times 10^{-6}$	$4.61\times 10^{-8}$	$5.58 \times 10^{-5}$	$1.91  imes 10^{-4}$
	$(3.75 \times 10^{-7})$	$(5.52 \times 10^{-7})$	$(1.45 \times 10^{-6})$	$(2.86 \times 10^{-6})$
α	$3.05 \times 10^{-6}$	$1.08 \times 10^{-5}$		
	$(4.91 \times 10^{-7})$	$(9.65 \times 10^{-7})$		
β	$5.86 \times 10^{-1}$	$7.16 \times 10^{-1}$		
	$(4.26 \times 10^{-2})$	$(2.92\times10^{-2})$		
ρ	$3.22 \times 10^2$	$1.41 \times 10^2$		
	$(4.26 \times 10^1)$	$(1.30 \times 10^1)$		

Table 4.2: MLEs for the HN-GARCH model and the homoscedastic model, based on the S&P 500 levels from January 3, 2013, through December 30, 2017, and from January 3, 2018, through December 30, 2022, with standard errors in parentheses.

Based on these parameters, we calculate the optimal HN-GARCH and homoscedastic strategies for the default investment parameters described in Section 4.4.1 and evaluate the performance in-sample, with the S&P 500 in the role of the single risky asset in the market. The plots for







Figure 4.5: Relative consumption  $\bar{C}_t^*$  for both periods of five years, plotting the HN-GARCH solution and the homoscedastic variant and using the parameters from Table 4.2.

relative consumption in Figure 4.5 show that, in general, the HN-GARCH and the homoscedastic solution follow the same major trends. However, in both Figure 4.5a and 4.5b, the blue line for the heteroscedastic model clearly exceeds the homoscedastic line for most of the time. The

reverse, i.e., a period where the homoscedastic investor consumes significantly more than the one following the HN-GARCH, cannot be found.



Figure 4.6: Cumulative consumption  $\sum_{s=0}^{t} C_s^*$  for both periods of five years, plotting the HN-GARCH solution and the homoscedastic variant and using the parameters from Table 4.2.

This observation is also reflected in Figure 4.6, which plots the sum of all consumption values at intermediate time points up to time t for both periods. The HN-GARCH investor's total consumption exceeds the homoscedastic investor's in both cases, although the difference is larger in Figure 4.6a for the period 2013-2017, amounting to almost 10% more total consumption by an HN-GARCH investor.



Figure 4.7: The approximate wealth process  $\exp \{W_t\}$  for both periods of five years, plotting the HN-GARCH solution and the homoscedastic variant and using the parameters from Table 4.2.

The wealth processes of the HN-GARCH and the homoscedastic model again develop simultaneously in both periods of five years, see Figure 4.7. A significant decrease in the level of wealth can be observed in 2015 and in the whole period from 2018 to the end of 2022. We note, however, that in both models, the initial level is clearly exceeded between 2013 and 2015. While there is a relatively rapid decrease in wealth in 2015, the total amount of money invested in the risky asset and the bank account falls gradually (except for the beginning of the pandemic in early 2020) during 2018-2022. For the period in Figure 4.7a, the non-consumed wealth of an HN-GARCH investor exceeds the wealth of a homoscedastic investor most of the time, and the former is 8% larger at December 30, 2018.

# 4.5 Conclusion

In this chapter, we derive the first closed-form solution to a consumption and investment problem for a model where the underlying log-asset price follows a general affine GARCH process. In particular, this general solution includes the HN-GARCH, the IG-GARCH, and a homoscedastic model as special cases. We show that the optimal log-wealth process and the optimal control on consumption also follow affine GARCH processes. Our setup includes the existing solution to the problem without consumption, i.e., for an investor maximizing only utility from terminal wealth, as a limiting case. Our numerical findings highlight a greater impact of heteroscedasticity on the solution than non-Gaussianity, where the latter is measured via skewness and excess kurtosis. Based on S&P 500 index data, in two periods of five years each, we analyze the strategy performance of a Gaussian GARCH model against a homoscedastic variant. Our real-world study shows that the heteroscedastic strategy can lead to up to 10% more total consumption while also ending up with 8% more wealth at the end of the investment horizon. These numbers suggest that the model choice is highly important for the strategy performance if consumption is included.

# 5 HARA Utilities and Mean-Variance Optimization

# 5.1 Introduction

This chapter presents an extension of the expected utility theory (EUT) problem for the more general class of hyperbolic absolute risk aversion (HARA) utility functions and for risk aversion levels previously excluded from consideration. Referring to Figure 5.1, where the contributions of Chapters 3 and 4 are already included, we now focus on generalizing the setup for utility derived from terminal wealth by adding the purple box. The essence of this paper was published in *Finance Research Letters* (Escobar-Anel et al., 2024a).



Figure 5.1: Overview of the contributions of this dissertation with regard to portfolio optimization under affine GARCH models, increasing the model complexity as well as the complexity of terminal and intermediate utility, see also Figures 1.1, 3.1, 4.1 and 6.1. The orange box coresponds to Chapter 3, the green box to Chapter 4, and the purple box to Chapter 5.

In the fifties, Harry M. Markowitz laid the foundation of modern portfolio theory, introducing the classical mean-variance (MV) problem in groundbreaking publications (Markowitz, 1952, 1959), earning him the Nobel Prize in 1990. The 1997 Nobel laureate, Robert C. Merton, further advanced portfolio analyses by finding, in the early seventies, closed-form solutions in EUT for a large family of utilities, some of which interpretable as mean-variance objective functions, (Merton, 1969, 1971). These timeless works remain cornerstones and sources of

## 5 HARA Utilities and Mean-Variance Optimization

new applications in finance. In the eighties, Robert F. Engle developed the ARCH model (Engle, 1982), fundamentally changing the way market time series are analyzed and clearing the path to his Nobel Prize in 2003. Bollerslev (1986) extended Engle's work with the GARCH model before Heston and Nandi (2000) introduced the first affine GARCH, drastically improving usability with regard to pricing problems through the existence of closed-form solutions. As mentioned in the preceding chapters of this dissertation, other affine GARCH models were introduced by, e.g., Christoffersen et al. (2006, Inverse-Gaussian GARCH) and Ornthanalai (2014, Lévy GARCH). Finally, Badescu et al. (2019) defined a general framework for the class of affine GARCH models. More recent activities in the field of affine GARCH models include a multivariate setting introduced by Escobar-Anel et al. (2020), the derivation of semi-closed-form solutions for VIX and target volatility options (Cao et al., 2020), and closed-form expressions for risk-maximizing hedging strategies (Augustyniak and Badescu, 2021). A generalization of the framework by Badescu et al. (2019) was recently presented by Augustyniak et al. (2023).

Financial institutions still heavily rely on the closed-form solutions of Markowitz and Merton for portfolio decisions. In more sophisticated and realistic stochastic settings, such as those involving stochastic volatility or jumps, achieving reliable estimates of the underlying Itô processes is undeniably a challenging task. Furthermore, the lack of explicit formulas often leads to the use of models that fail to account for important stylized facts of asset returns, such as conditional heteroscedasticity. Our solution bridges this gap for practitioners seeking to move beyond simplistic models while benefiting from analytical solutions.

This work connects the three branches mentioned above, extending the recent developments in EUT for GARCH models of Escobar-Anel et al. (2021), presented in Chapter 3 in this dissertation, in two directions. The first concerns the class of utility functions: We go beyond the previously tackled subset of constant relative risk aversion (CRRA) and fully solve the whole constant relative risk aversion (CRRA) as well as the HARA class of utilities in Merton (1971). The second direction consists of solving the mean-variance problem after establishing a link via the work of Zhou and Li (2000) and Zhu and Escobar-Anel (2022).

Numerous results have been published in the fields of EUT and MV for rich continuous-time models – see, for instance, Liu (2007) for closed-form solutions to EUT problems, and Basak and Chabakauri (2010) for time consistent and inconsistent solutions within MV. To the best of our knowledge, no work concerning HARA utilities and mean-variance theory exists in the important world of affine GARCH models.

The contributions of this chapter are the following:

- We develop an innovated approach to approximate the self-financing condition (SFC) to accommodate HARA utility functions.
- We derive a closed-form optimal portfolio strategy for an investor maximizing a HARA utility function with general exponent  $\gamma \in \mathbb{R} \setminus \{0, 1\}$ , where the underlying stock exhibits general affine GARCH dynamics.
- We use the HARA approach to reveal solutions to two interesting problems for practitioners, the constant proportion portfolio insurance (CPPI) and the MV problem.
- We provide numerical insight to MV optimization with affine GARCH models via an analysis based on 20 years of S&P 500 data, showing that a heteroscedastic GARCH investor could outperform a homoscedastic strategy by three percentage points in annualized standard deviation over a five-year horizon.

The remainder of this chapter is organized as follows: In Section 5.2, we state the general optimization problem and present two approaches to deriving a suitable approximation for the self-financing condition (SFC). Section 5.3 contains the main theorem concerning the optimal

solution. Section 5.4 connects the result to mean-variance and higher-order portfolio theory, and Section 5.5 presents the numerical analysis. We conclude in Section 5.6.

## 5.2 Formulation of the EUT Problem

Assume the investor maximizes a HARA utility function of terminal wealth given by

$$U(v) = \frac{1 - b_H}{b_H} \left( \frac{a_H}{1 - b_H} \cdot v + d_H \right)^{b_H} - e_H,$$
(5.1)

with  $b_H < 1$ ,  $b_H \neq 0$ ,  $a_H > 0$  and  $\frac{a_H}{1-b_H} \cdot v + d_H > 0$ . Dropping the intercept  $e_H$  and setting  $\gamma \coloneqq b_H$ ,  $K \coloneqq (1-\gamma)^{1-\gamma} \cdot a_H^{\gamma} \cdot (\operatorname{sgn}(v-L))^{\gamma}$ , and  $L \coloneqq -\frac{d_H(1-\gamma)}{a_H}$ , we can transform this function to a shorter representation:

$$U(v) = \frac{1}{\gamma} \cdot K \cdot |v - L|^{\gamma}, \qquad (5.2)$$

Note that (5.2) emerges as a direct transformation from (5.1) via the definitions of K and L. In case  $\gamma < 1$ , the original constraint  $\frac{a_H}{1-b_H}v + d_H > 0$  for (5.1) translates into v > L, and L can be interpreted as a lower bound on wealth that is exceeded with certainty. If  $\gamma > 1$ , the same constraint imposes an upper bound on wealth – in particular, the domain of v is bounded by v - L < 0. All in all, we thus require

$$\operatorname{sgn}\left(v-L\right) = \operatorname{sgn}\left(1-\gamma\right) \tag{5.3}$$

for (5.2), and consequently note that the expression for K simplifies to  $K = \text{sgn}(1-\gamma) \cdot a_H^{\gamma} \cdot |1-\gamma|^{1-\gamma}$ . In the sequel of this chapter, we will mainly use the representation in (5.2) and refer to the parameters  $\gamma$ , K, and L, with v bounded via (5.3).

Let  $\pi_t$  denote the share of wealth invested in the risky asset in the time interval (t, t + 1], and let  $X_t = \log P_t^{(1)}$  be the log asset price. Assume that the log asset price and its conditional variance  $h_t$  follow an affine GARCH model that can be written in the general form as presented earlier in (3.1) and (4.4) (Badescu et al., 2019):

$$X_t - X_{t-1} = f_1(h_t, \theta) + \sqrt{h_t} \epsilon_t$$
(5.4a)

$$h_{t+1} = f_2(h_t, \theta) + f_3(h_t, \epsilon_t, \theta),$$
 (5.4b)

for a vector of parameters  $\theta$  and a sequence of innovations  $\{\epsilon_t\}_t$ . In particular,  $f_1$  and  $f_2$  are affine in  $h_t$  and  $f_3$  is such that the conditional joint bivariate moment generating function (MGF) of the log-asset return and its conditional variance has the representation

$$\mathbb{E}\left[\exp\left\{u\cdot(X_{t}-X_{t-1})+v\cdot h_{t+1}\right\} \mid \mathcal{F}_{t-1}\right] = \exp\left\{A(u,v)+B(u,v)\cdot h_{t}\right\},\tag{5.5}$$

for some coefficients A and B dependent on the model parameters.

Maximizing utility from terminal wealth and taking into account the representation of our utility function in (5.2), we define  $\hat{V}_t := |V_t - L_t|$ , where  $V_t$  is the investor's wealth at time t and  $L_t := L \cdot e^{-r \cdot (T-t)}$  with the continuously compounded interest rate r. The following two subsections present two different ways of obtaining the same approximation of log  $\hat{V}_t$ , which will play a crucial role in the derivation of the optimal solution to the EUT problem in Section 5.3.

## 5.2.1 Taylor Approach

The budget constraint for the portfolio reads:

$$\frac{V_t - V_{t-1}}{V_{t-1}} = \pi_{t-1} \cdot \frac{P_t^{(1)} - P_{t-1}^{(1)}}{P_{t-1}^{(1)}} + (1 - \pi_{t-1}) \cdot \frac{P_t^{(0)} - P_{t-1}^{(0)}}{P_{t-1}^{(0)}},$$
(5.6)

with  $\{V_t\}_t$ ,  $\{P_t^{(1)}\}_t$  and  $\{P_t^{(0)}\}_t$  being the wealth process, the price process of the risky asset and the price process of the riskless bank account, respectively.

Along the lines of Campbell and Viceira (1999) and Escobar-Anel et al. (2022a, 2021)<sup>1</sup>, we transform returns into log returns by applying a second-order Taylor approximation of the logarithm around 1 to the processes  $\log\left(\frac{V_t-L_t}{V_{t-1}-L_{t-1}}\right)$ ,  $\log\left(\frac{P_t^{(1)}}{P_{t-1}^{(1)}}\right)$ ,  $\log\left(\frac{P_t^{(0)}}{P_{t-1}^{(0)}}\right)$ ,  $\log\left(\frac{L_t}{L_{t-1}}\right)$ , as well as an approximation of the squared return, both consistent with the popular continuous-time literature. The approximation was demonstrated to be accurate in the context of affine GARCH models by Escobar-Anel et al. (2022a) and Escobar-Anel et al. (2021). Note that the cited works seek an approximation for  $\log\left(\frac{V_t}{V_{t-1}}\right)$  and do not include the additional terms  $L_t$  and  $L_{t-1}$ . To obtain an extension based on the true SFC (5.6), we follow the steps below:

- (a) Use a second-order Taylor approximation of the logarithm around 1 for  $\log\left(\frac{V_t L_t}{V_{t-1} L_{t-1}}\right)$ .
- (b) Use the same second-order Taylor approximation of the logarithm of the return of the risky asset around 1 to obtain:

$$X_t - X_{t-1} = \log\left(\frac{P_t^{(1)}}{P_{t-1}^{(1)}}\right) \approx \frac{P_t^{(1)} - P_{t-1}^{(1)}}{P_{t-1}^{(1)}} - \frac{1}{2}\left(\frac{P_t^{(1)} - P_{t-1}^{(1)}}{P_{t-1}^{(1)}}\right)^2 \tag{5.7}$$

$$\frac{P_t^{(1)} - P_{t-1}^{(1)}}{P_{t-1}^{(1)}} \approx X_t - X_{t-1} + \frac{1}{2} \left(\frac{P_t^{(1)} - P_{t-1}^{(1)}}{P_{t-1}^{(1)}}\right)^2.$$
(5.8)

Moreover, with a small size of the time step, we can approximate the squared return by its conditional variance in (5.8), analogously to a continuous-time setting. Subsequently, we observe from (5.8) that  $Var\left[\frac{P_t^{(1)}-P_{t-1}^{(1)}}{P_{t-1}^{(1)}} \mid \mathcal{F}_{t-1}\right] \approx Var\left[X_t - X_{t-1} \mid \mathcal{F}_{t-1}\right] = h_t$ . Together, this leads to:

$$\frac{P_t^{(1)} - P_{t-1}^{(1)}}{P_{t-1}^{(1)}} \approx X_t - X_{t-1} + \frac{1}{2}h_t.$$
(5.9)

(c) Use the linear approximation  $\log x\approx x-1$  with x close to 1 for the riskless asset and the discounted floor to find

$$\frac{P_t^{(0)} - P_{t-1}^{(0)}}{P_{t-1}^{(0)}} \approx \log\left(\frac{P_t^{(0)}}{P_{t-1}^{(0)}}\right) = r, \qquad L_t - L_{t-1} = (e^r - 1) L_{t-1} \approx rL_{t-1}.$$
(5.10)

 $\Leftrightarrow$ 

<sup>&</sup>lt;sup>1</sup>See also Sections 3.2 and 4.2.1.

(d) Use the same approach as in (b), now for the return on wealth, noting that, according to (5.9),  $Var\left[\frac{V_t-V_{t-1}}{V_{t-1}} \mid \mathcal{F}_{t-1}\right] \approx \pi_{t-1}^2 \cdot h_t$ . This leads to the approximation

$$\left(\frac{V_t - V_{t-1}}{V_{t-1}} - \frac{L_t - L_{t-1}}{V_{t-1}}\right)^2 \approx \pi_{t-1}^2 \cdot h_t.$$
(5.11)

All in all, with  $\hat{\pi}_{t-1} = \pi_{t-1} \cdot \frac{V_{t-1}}{V_{t-1}-L_{t-1}}$ , we obtain

$$\log\left(\frac{V_{t}-L_{t}}{V_{t-1}-L_{t-1}}\right) \stackrel{(a)}{\approx} \left(\frac{V_{t}-L_{t}}{V_{t-1}-L_{t-1}}-1\right) - \frac{1}{2}\left(\frac{V_{t}-L_{t}}{V_{t-1}-L_{t-1}}-1\right)^{2} \\ = \left(\frac{V_{t-1}}{V_{t-1}-L_{t-1}}\right) \left(\frac{V_{t}-L_{t}-V_{t-1}+L_{t-1}}{V_{t-1}}\right) \\ - \frac{1}{2}\left(\frac{V_{t-1}}{V_{t-1}-L_{t-1}}\right)^{2}\left(\frac{V_{t}-V_{t-1}}{V_{t-1}}-\frac{L_{t}-L_{t-1}}{V_{t-1}}\right)^{2} \\ \stackrel{(c).(d)}{\approx}\left(\frac{V_{t-1}}{V_{t-1}-L_{t-1}}\right) \left(\frac{V_{t}-V_{t-1}}{V_{t-1}}-\frac{L_{t-1}\left(e^{r}-1\right)}{V_{t-1}}\right) \\ - \frac{1}{2}\left(\frac{V_{t-1}}{V_{t-1}-L_{t-1}}\right)^{2}\pi_{t-1}^{2}h_{t} \\ \stackrel{(b).(c)}{\approx}\left(\frac{V_{t-1}}{V_{t-1}-L_{t-1}}\right) \left(\pi_{t-1}\left(X_{t}-X_{t-1}+\frac{1}{2}h_{t}\right)+(1-\pi_{t-1})r-\frac{L_{t-1}}{V_{t-1}}r\right) \\ - \frac{1}{2}\left(\frac{V_{t-1}}{V_{t-1}-L_{t-1}}\right)^{2}\pi_{t-1}^{2}h_{t} \\ = \hat{\pi}_{t-1}\left(X_{t}-X_{t-1}+\frac{1}{2}h_{t}\right)+(1-\hat{\pi}_{t-1})r-\frac{1}{2}\hat{\pi}_{t-1}^{2}h_{t}.$$
(5.12)

This approach is independent of  $V_t - L_t$  being positive or negative. If  $V_t - L_t$  and  $V_{t-1} - L_{t-1}$  are both negative, we arrive at the same equation (5.12), but with  $\pi_{t-1}$  and  $\hat{\pi}_{t-1}$  having different signs. Reordering terms in (5.12), we obtain the following expression for  $\hat{W}_t$  as an approximation of log  $\hat{V}_t$ :

$$\hat{W}_{t} = \hat{W}_{t-1} + \hat{\pi}_{t-1} \cdot \left(X_{t} - X_{t-1}\right) + \left(1 - \hat{\pi}_{t-1}\right)r + \frac{1}{2}\left(\hat{\pi}_{t-1} - \hat{\pi}_{t-1}^{2}\right)h_{t}.$$
(5.13)

We note that this expression is identical to the approximation of the log wealth in (3.7) for the CRRA case (cf. Escobar-Anel et al., 2021) if  $\gamma < 0$  and L = 0, which holds exactly in continuous time when Itô's Lemma can be applied to the budget constraint (5.6).

## 5.2.2 CPPI Approach

Alternatively to the approach presented in Section 5.2.1, we may treat the modified CRRA utility in (5.2) as a CPPI setting.<sup>2</sup> Assume that  $L \leq v_0 \cdot e^{rT}$ , with  $v_0$  the initial wealth. Note that this yields an implicit condition on the parameter  $d_H$  of the original HARA utility function, since for  $\gamma < 1$ 

$$L \le v_0 \cdot e^{rT} \quad \Leftrightarrow \quad d_H \ge -\frac{a_H \cdot v_0 \cdot e^{rT}}{1 - \gamma}.$$
(5.14)

<sup>&</sup>lt;sup>2</sup>We restrict this approach to  $\gamma < 1$ , where (5.3) yields the relation  $V_T \ge L$ .

Assuming  $d_H$  as fixed, this can also be seen as a restriction on the investor's initial wealth. Now define the current (time-dependent) level of the *floor*  $L_t$  as above via

$$L_t \coloneqq L \cdot e^{-r \cdot (T-t)}$$

 $L_t$  can be interpreted as the amount of money that needs to be invested in the riskless asset at time t to reach the terminal threshold L for sure. Furthermore, define  $\hat{V}_t \coloneqq V_t - L_t$  as the cushion, i.e., the money beyond the threshold  $L_t$  that is used for return maximization. For this part of the wealth,  $\hat{V}_t$ , we can choose our allocation between the riskless and the risky asset freely. This leads to the existing approach using CRRA utility as of Chapter 3, replacing  $V_t$ by  $\hat{V}_t$  and writing  $\hat{\pi}_t$  for the share of  $\hat{V}_t$  that is invested in the risky asset in the time interval (t, t + 1]. In particular, we approximate  $\hat{V}_t$  via  $e^{\hat{W}_t}$ , where

$$\hat{W}_{t} = \hat{W}_{t-1} + \hat{\pi}_{t-1} \cdot (X_{t} - X_{t-1}) + (1 - \hat{\pi}_{t-1}) \cdot r + \frac{1}{2} \left( \hat{\pi}_{t-1} - \hat{\pi}_{t-1}^{2} \right) \cdot h_{t},$$
(5.15)

according to (3.7). Note that (5.15) is identical to (5.13). Furthermore, the above reasoning implies that the wealth dropping below the floor in this environment is equivalent to the wealth dropping below zero in the CRRA environment, which is excluded by construction.<sup>3</sup>

# 5.3 Optimal Investment Maximizing HARA Utility

Now, with T > 0 some finite time horizon, and the log asset price following an affine GARCH model, consider the expected utility problem from terminal wealth for an investor maximizing a HARA utility function U in the form of (5.2). Our goal is to solve the following optimization problem:

$$\max_{\{\pi_s\}_t^{T-1}} \frac{K}{\gamma} \mathbb{E}_t \left[ \exp\left\{ \gamma \hat{W}_T \right\} \right] \rightleftharpoons \phi_t \left( \hat{W}_t, h_{t+1} \right), \tag{5.16}$$

with  $\mathbb{E}_t$  again denoting the conditional expectation under  $\mathcal{F}_t$  at time t. On the optimal solution to this problem, we establish the following result:

**Theorem 5.1.** Assume that the log asset price follows an affine GARCH model, and assume that the investor maximizes a HARA utility function of terminal wealth in the form of (5.2) with a general exponent  $\gamma \in \mathbb{R} \setminus \{0, 1\}$ . Furthermore, assume that the coefficient A of the conditional joint bivariate MGF (5.5) satisfies

$$\frac{\partial}{\partial u}A(u,v) = r \qquad \forall u, v \in \mathbb{R}.$$
(5.17)

Then, the solution to the problem (5.16) is given by

$$\phi_t \left( \hat{W}_t, h_{t+1} \right) = \frac{K}{\gamma} \exp\left\{ \gamma \hat{W}_t + D_{t,T} \left( \hat{\pi}_t^* \right) + E_{t,T} \left( \hat{\pi}_t^* \right) \cdot h_{t+1} \right\},$$
(5.18)

<sup>&</sup>lt;sup>3</sup>Since we are working with an approximation of the log wealth, the wealth process in the model is always going to stay positive. Plugging the model parameters, e.g., of the HN-GARCH model, into the true budget constraint (5.6) and then solving for the innovation of the log-asset return under the condition that  $V_t \leq 0$ yields an interval for the innovation where the actual wealth process would be pulled below zero. Working with parameter estimates for the HN-GARCH model (Christoffersen et al., 2006) and numerically evaluating the probability of the innovation ending up in this extreme interval (e.g., via the software R) yields zero, i.e., the probability is so small that it is not numerically measurable.

with some recursive coefficients D and E dependent on the (partial) optimal solution  $\hat{\pi}_t^*$ , which solves the optimality equation

$$\frac{\partial}{\partial u}B\left(\gamma\hat{\pi}_t^*, E_{t,T}\left(\hat{\pi}_t^*\right)\right) = \hat{\pi}_t^* - \frac{1}{2}$$
(5.19)

and the second-order condition

$$\frac{\partial^2}{\partial u^2} B\left(\gamma \hat{\pi}_t^*, E_{t,T}\left(\hat{\pi}_t^*\right)\right) \begin{cases} > \frac{1}{\gamma}, & \gamma \in \mathbb{R} \setminus [0,1], \\ < \frac{1}{\gamma}, & \gamma \in (0,1). \end{cases}$$
(5.20)

The optimal strategy  $\pi_t^*$  is obtained via

$$\pi_t^* = \hat{\pi}_t^* \cdot \frac{V_t - L_t}{V_t}.$$
(5.21)

### Remark 5.2 (On Theorem 5.1).

- (1) Theorem 5.1 generalizes the known solution presented in Chapter 3 (Escobar-Anel et al., 2021) in several directions. Besides the use of the more general HARA class of utility functions, the range of exponents  $\gamma$  is extended to investors with very low risk aversion levels ( $0 < \gamma < 1$ ) and to exponents  $\gamma > 1$ , which establishes a connection to mean-variance theory and higher-moment portfolio selection, as we will see in the following sections.
- (2) The additional requirement  $\frac{\partial}{\partial u}A(u,v) = r$  is satisfied for both the HN-GARCH (Heston and Nandi, 2000) and the IG-GARCH (Christoffersen et al., 2006) model, which form the two well-known special cases of the general formulation of an affine GARCH model as of (5.4). The condition can be relaxed to  $\frac{\partial^2}{\partial^2 u}A(u,v) \ge 0$ , and the theorem still stands, although with more tedious expressions, especially for the second-order conditions.
- (3) The optimality equation (5.19) is the same for all values of  $\gamma$ . This includes, in particular, the known case of  $\gamma < 0$  with the second-order condition (5.20).

Proof of Theorem 5.1. Focusing on the partial strategy  $\hat{\pi}_t$  only, we assume the investor's initial wealth to be equal to  $\hat{v}_0 = v_0 - L_0$ , we set  $\hat{W}_0 = \log \hat{v}_0$ , and approximate the log wealth via (5.13). This sub-problem is identical to the case of an investor maximizing a CRRA utility and can be solved recursively for the deterministic optimal strategy  $\{\hat{\pi}_t^*\}_{t=0}^{T-1}$  via guessing the correct form for the value function, using the MGF (5.5), and then applying Bellman's value iteration to obtain (5.18), see Theorem 3.1.  $D_{t,T}$  and  $E_{t,T}$  are defined via  $D_{T,T} = E_{T,T} = 0$  and via the coefficients A and B in (5.5):

$$D_{t,T}(\hat{\pi}_t) = D_{t+1,T}(\hat{\pi}_{t+1}^*) + (1 - \hat{\pi}_t) \gamma r + A\left(\gamma \hat{\pi}_t, E_{t+1,T}(\hat{\pi}_{t+1}^*)\right), \qquad (5.22a)$$

$$E_{t,T}(\hat{\pi}_t) = B\left(\gamma \hat{\pi}_t, E_{t+1,T}(\hat{\pi}_{t+1}^*)\right) + \frac{\gamma}{2} \left(\hat{\pi}_t - \hat{\pi}_t^2\right).$$
(5.22b)

Taking derivatives of the objective function w.r.t.  $\hat{\pi}_t$ , using the definitions of  $D_{t,T}$  and  $E_{t,T}$ , yields the equations (5.17) and (5.19). These first-order conditions are identical to the CRRA case, and (5.17) is satisfied by assumption. Concerning the second-order condition, assume that

### 5 HARA Utilities and Mean-Variance Optimization

 $\hat{\pi}_t^*$  satisfies (5.19) and define

$$\begin{split} f\left(\hat{\pi}_{t}\right) &= \frac{K}{\gamma} \exp\left\{\gamma \hat{W}_{t} + D_{t,T}\left(\hat{\pi}_{t}\right) + E_{t,T}\left(\hat{\pi}_{t}\right) \cdot h_{t+1}\right\},\\ g\left(\hat{\pi}_{t}\right) &= -\gamma r + \gamma \frac{\partial}{\partial u} A\left(\gamma \hat{\pi}_{t}, E_{t+1,T}\left(\hat{\pi}_{t+1}^{*}\right)\right) \\ &+ \left(\frac{\gamma}{2} - \gamma \hat{\pi}_{t} + \gamma \frac{\partial}{\partial u} B\left(\gamma \hat{\pi}_{t}, E_{t+1,T}\left(\hat{\pi}_{t+1}^{*}\right)\right)\right) \cdot h_{t+1} \\ &= \left(\frac{\gamma}{2} - \gamma \hat{\pi}_{t} + \gamma \frac{\partial}{\partial u} B\left(\gamma \hat{\pi}_{t}, E_{t+1,T}\left(\hat{\pi}_{t+1}^{*}\right)\right)\right) \cdot h_{t+1}, \end{split}$$

using again (5.17) in the last equality. Note that  $g(\hat{\pi}_t^*) = 0$  due to (5.19), and the second derivative of the objective function f can be expressed as

$$\frac{\partial^2}{\partial \hat{\pi}_t^2} f\left(\hat{\pi}_t\right) = f\left(\hat{\pi}_t\right) \left[ g^2\left(\hat{\pi}_t\right) + \left(\gamma^2 \frac{\partial^2}{\partial u^2} B\left(\gamma \hat{\pi}_t, E_{t+1,T}\left(\hat{\pi}_{t+1}^*\right)\right) - \gamma\right) h_{t+1} \right].$$
(5.23)

We distinguish between two cases to justify (5.20):

- $\gamma \notin [0,1]$ : If  $\gamma < 0$  or  $\gamma > 1$ , we have  $\frac{K}{\gamma} < 0$  and negativity of (5.23) at the extreme point can be ensured via making the coefficient in front of the conditional variance  $h_{t+1}$  positive.
- $\gamma \in (0,1)$ : In this case,  $\frac{K}{\gamma} > 0$ , implying that the coefficient in front of  $h_{t+1}$  in (5.23) needs to be negative.

The actual strategy  $\pi_t^*$  can be derived via scaling the partial strategy  $\hat{\pi}_t^*$ , including also the additional investment  $L_t$  in the riskless asset. Using the true process  $V_t$ , this leads to a share of  $\hat{\pi}_t^* \cdot \frac{V_t - L_t}{V_t}$  of the total wealth invested in the risky asset, and to a proportion of  $1 - \hat{\pi}_t^* \cdot \frac{V_t - L_t}{V_t} = (1 - \hat{\pi}_t^*) \cdot \frac{V_t - L_t}{V_t} + \frac{L_t}{V_t}$  for the riskless asset.  $\Box$ 

# 5.4 Connection to Mean-Variance Theory

In a dynamic mean-variance portfolio problem, the objective is to maximize expected terminal wealth, given an upper bound for the variance of terminal wealth. The Pareto-optimal solutions to this trade-off can be found by solving the Markowitz- $\lambda$  problem

$$\max_{\{\pi_s\}_t^{T-1}} \left\{ \mathbb{E}\left[ V_T \mid \mathcal{F}_t \right] - \lambda \cdot Var\left[ V_T \mid \mathcal{F}_t \right] \right\}$$
(5.24)

for different values of  $\lambda > 0$ . The optimization problem in (5.24) is not solvable via dynamic programming due to time inconsistency, but the following result, in its original version presented by Zhou and Li (2000, Thm. 3.1)<sup>4</sup> provides a workaround.

**Proposition 5.3** (cf. Zhou and Li, 2000, Theorem 3.1). Any pre-commitment solution to (5.24) will also solve

$$\min_{\{\pi_s\}_t^{T-1}} \left\{ \mathbb{E}\left[ (V_T - \mu)^2 \mid \mathcal{F}_t \right] \right\},$$
(5.25)

where  $\mu \coloneqq \frac{1}{2\lambda} + \mathbb{E}\left[V_T \mid \mathcal{F}_t\right]$ .

<sup>&</sup>lt;sup>4</sup>See also Zhu and Escobar-Anel (2022, Thm. 2).

Proof of Proposition 5.3. Assume that  $\pi = {\pi_t}_{t=0}^{T-1}$  is optimal for (5.24), and denote the corresponding wealth process by V. Furthermore, assume that  $\pi$  is not optimal for (5.25), i.e., assume the existence of a pair  $\pi^*$  and  $V^*$  such that

$$\mathbb{E}_{t}\left[(V_{T}-\mu)^{2}\right] > \mathbb{E}_{t}\left[(V_{T}^{*}-\mu)^{2}\right]$$
  

$$\Leftrightarrow \mathbb{E}_{t}\left[V_{T}^{2}\right] - 2\mu\mathbb{E}_{t}\left[V_{T}\right] + \mu^{2} > \mathbb{E}_{t}\left[(V_{T}^{*})^{2}\right] - 2\mu\mathbb{E}_{t}\left[V_{T}^{*}\right] + \mu^{2}$$
  

$$\Leftrightarrow \mathbb{E}_{t}\left[V_{T}^{2}\right] - 2\mu\mathbb{E}_{t}\left[V_{T}\right] > \mathbb{E}_{t}\left[(V_{T}^{*})^{2}\right] - 2\mu\mathbb{E}_{t}\left[V_{T}^{*}\right].$$
(5.26)

As in Zhou and Li (2000), we define  $f(x, y) := \lambda x - \lambda y^2 - y$ , which is a concave function in (x, y) having first partial derivatives  $\frac{\partial}{\partial x} f(x, y) = \lambda$  and  $\frac{\partial}{\partial y} f(x, y) = -(1 + 2\lambda y)$ . It is easy to see that  $f\left(\mathbb{E}_t\left[V_T^2\right], \mathbb{E}_t\left[V_T\right]\right)$  yields the negative of the optimal value of the objective function in (5.24). But by the concavity of f, using its partial derivatives mentioned above and  $2\lambda\mu = 1 + \mathbb{E}_t\left[V_T\right]$  according to (5.25), we obtain

$$\begin{split} f\left(\mathbb{E}_{t}\left[\left(V_{T}^{*}\right)^{2}\right],\mathbb{E}_{t}\left[V_{T}^{*}\right]\right) &\leq f\left(\mathbb{E}_{t}\left[V_{T}^{2}\right],\mathbb{E}_{t}\left[V_{T}\right]\right) \\ &\quad +\lambda\left(\mathbb{E}_{t}\left[\left(V_{T}^{*}\right)^{2}\right]-\mathbb{E}_{t}\left[V_{T}^{2}\right]\right)-\left(1+2\lambda\mathbb{E}_{t}\left[V_{T}\right]\right)\left(\mathbb{E}_{t}\left[V_{T}^{*}\right]-\mathbb{E}_{t}\left[V_{T}\right]\right)\right) \\ &\quad =f\left(\mathbb{E}_{t}\left[V_{T}^{2}\right],\mathbb{E}_{t}\left[V_{T}\right]\right) \\ &\quad +\lambda\left(\mathbb{E}_{t}\left[\left(V_{T}^{*}\right)^{2}\right]-\mathbb{E}_{t}\left[V_{T}^{2}\right]-2\mu\mathbb{E}_{t}\left[V_{T}^{*}\right]+2\mu\mathbb{E}_{t}\left[V_{T}\right]\right) \\ &\quad$$

where we used (5.26) and  $\lambda > 0$  for the last inequality. This is equivalent to

$$\mathbb{E}\left[V_T \mid \mathcal{F}_t\right] - \lambda \cdot Var\left[V_T \mid \mathcal{F}_t\right] < \mathbb{E}\left[V_T^* \mid \mathcal{F}_t\right] - \lambda \cdot Var\left[V_T^* \mid \mathcal{F}_t\right],$$

which is a contradiction to the assumption that  $\pi$  with wealth process V is optimal for (5.24). The pair of  $\pi$  and V thus is also optimal for (5.25).

With Proposition 5.3, assuming existence of the optimal solution to (5.24) and uniqueness of the optimal solution to (5.25), the mean-variance problem can be solved via the auxiliary problem (5.25). On the other hand, (5.25) is a special case of (5.2) if we set  $\gamma = 2$ , K = -1, and  $L = \mu$ . Fixing  $\bar{\lambda} > 0$  as the parameter for (5.24) and choosing  $L = \bar{\mu} > \frac{1}{2\bar{\lambda}}$  fixes the level of expected terminal wealth to  $\bar{\mu} - \frac{1}{2\bar{\lambda}}$ . Theorem 5.1 then yields the optimal solution to

$$\max_{\{\pi_s\}_t^{T-1}} \left\{ -\frac{1}{2} \mathbb{E} \left[ (V_T - \bar{\mu})^2 \mid \mathcal{F}_t \right] \right\} = \max_{\{\pi_s\}_t^{T-1}} \left\{ -\frac{1}{2} \mathbb{E} \left[ V_T^2 - 2V_T \bar{\mu} + \bar{\mu}^2 \mid \mathcal{F}_t \right] \right\} 
= \max_{\{\pi_s\}_t^{T-1}} \left\{ -\frac{1}{2} \mathbb{E} \left[ V_T^2 \mid \mathcal{F}_t \right] + \bar{\mu} \left( \bar{\mu} - \frac{1}{2\bar{\lambda}} \right) - \frac{1}{2} \bar{\mu}^2 \right\} 
= \max_{\{\pi_s\}_t^{T-1}} \left\{ -\frac{1}{2} \mathbb{E} \left[ V_T^2 \mid \mathcal{F}_t \right] + \frac{1}{2} \left( \bar{\mu}^2 - \frac{\bar{\mu}}{\bar{\lambda}} \right) \right\}.$$
(5.27)

The variance of the optimal portfolio can be recovered using this optimal value of the objective function (5.27) and the fact that  $Var[V_T | \mathcal{F}_t] = \mathbb{E}[V_T^2 | \mathcal{F}_t] - (\mathbb{E}[V_T | \mathcal{F}_t])^2$ , yielding the corresponding value on the efficient frontier.

**Remark 5.4** (Higher-order moments in portfolio selection). Numerous authors have investigated the impact of higher moments on the portfolio return (see, e.g., Guidolin and Timmermann, 2008; Harvey et al., 2010; Lai, 1991). In the present version, the general formulation of Theorem 5.1 allows for a connection to a problem including skewness or skewness and kurtosis, choosing exponents of 3 or 4 in the utility function. A numerical analysis studying the impact of higher-order moments on the portfolio choice could be subject to further research.

# 5.5 Numerical Analysis

For our numerical analysis, we focus on the well-studied HN-GARCH model (see Section 4.3.2), forming a special case of our general setup (5.4) with  $\theta = (\lambda, \omega, \alpha, \beta, \rho)$ ,  $\{\epsilon_t\}_t$  a sequence of i.i.d. standard normal random variables and

$$f_1(h_t,\theta) = r + \lambda h_t, \qquad f_2(h_t,\theta) = \omega + \beta h_t, \qquad f_3(h_t,\epsilon_t,\theta) = \alpha \left(\epsilon_t - \rho \sqrt{h_t}\right)^2, \qquad (5.28)$$

leading the following representations of the coefficients A and B in (5.5), see Appendix B.1:

$$A(u, v) = ur + v\omega - \frac{1}{2}\log(1 - 2v\alpha),$$
 (5.29a)

$$B(u,v) = u\lambda + v\left(\alpha\rho^2 + \beta\right) + \frac{(u - 2v\alpha\rho)^2}{2(1 - 2v\alpha)}.$$
(5.29b)

Our parameters are estimated based on the S&P 500 levels from January 2, 2003, through December 30, 2022. While working with larger datasets could affect parameter stability, our time series includes two major crises and can thus be considered a representative sample. The 20-year time span falls within the range of various other studies that estimated affine GARCH models, such as Ornthanalai (2014) and Babaoğlu et al. (2018), with time frames of 15 and 23 years, respectively. We once more follow the lines of the returns-only estimation described in Section 4.4.2 (cf. Escobar-Anel et al., 2022b), exploiting the normal distribution of the HN-GARCH innovations to obtain the log-likelihood function

$$l_T(x_1, \dots, x_T \mid \theta) = -\frac{1}{2} \sum_{t=1}^T \left( \log(2\pi) + \log(h_t) + \left(\frac{x_t - x_{t-1} - r - \lambda h_t}{\sqrt{h_t}}\right)^2 \right), \quad (5.30)$$

for the path of log returns  $\{x_t\}_{t=1,...,T}$ . The estimates  $\hat{\theta}$  for the vector of parameters  $\theta$  are given by  $\hat{\theta} = \arg \max_{\theta} l(x_1, \ldots, x_T \mid \theta)$ . The standard errors of the estimates are again calculated via the Fisher information matrix. The same dataset of S&P 500 returns is used to obtain the parameter values for the corresponding homoscedastic Gaussian model (HS), i.e., an HN-GARCH model with constant conditional variance ( $\alpha = \beta = 0$ ). The combined results with standard errors are provided in Table 5.1.<sup>5</sup>

We start our numerical study with the time horizon T set to five years with 252 trading days, aiming to study medium-term horizons. The interest rate is set to r = 2%, slightly higher than the average Federal Funds Rate of 1.31% over the time of the underlying dataset. We choose  $a_H = 1$  with the initial wealth  $V_0 = 1$ .

Focussing on the exponent  $\gamma = 2$  and the connection to mean-variance theory established in Section 5.4, we target a special case accessible only through the generalizations in Theorem 5.1. Figure 5.2 displays the efficient frontiers<sup>6</sup> both for the HN-GARCH and the homoscedastic model with the parameters from Table 5.1. The HN-GARCH dominates the homoscedastic

<sup>&</sup>lt;sup>5</sup>Since  $\omega < 0$  in the context of the HN-GARCH model is known to render the model ill-defined and given the large standard error, we set  $\omega = 0$  when working with the HN-GARCH parameter set from Table 5.1.

<sup>&</sup>lt;sup>6</sup>In the presence of a riskless asset in the market, the efficient frontier (or *capital market line*, *CML*), is a straight line with the leftmost point marking the return of this riskless asset.

Table 5.1:	: Maximum likelihood estimates (MLEs) for the HN-GARCH model and the homoscedastic
	model, based on the S&P 500 levels from January 2, 2003, through December 30, 2022, with
	standard errors in parentheses.

Param.	HN-GARCH		Homoscedastic		
λ	2.46	(1.19)	1.94	(1.17)	
ω	$-1.31\times10^{-7}$	$(1.53 \times 10^{-7})$	$1.48 \times 10^{-4}$	$(1.09 \times 10^{-6})$	
α	$5.85 \times 10^{-6}$	$(2.85 \times 10^{-7})$			
β	$7.69 \times 10^{-1}$	$(1.16 \times 10^{-2})$			
ρ	$1.76 \times 10^2$	(8.40)			



Figure 5.2: Efficient frontiers for the HN-GARCH and the homoscedastic model with the parameter values as of Table 5.1. The time horizon is five years with 252 trading days; the results are annualized.

solution in this context, yielding lower values in terms of the standard deviation for the same expected return. The annualized difference between the two models increases as the expected return is increased and amounts to about three percentage points in standard deviation for an expected return of 10%. This emphasizes the importance of accounting for the time-changing nature of return volatility in portfolio selection and thus concerns a widely debated feature in the area.

Figure 5.3 shows the optimal initial strategy  $\pi_0^*$  for different lengths of the time horizon T. While the optimality equation (5.19) delivers the partial strategy  $\hat{\pi}_0^*$ , the optimal overall strategy  $\pi_0^*$  is obtained via scaling the partial strategy by the factor  $\frac{V_0-L_0}{V_0}$ . For the plot in Figure 5.3, we used  $\frac{V_0-L}{V_0}$  with  $V_0 = 1$  and L as described in Section 5.4 as a proxy for this factor. Note that (5.3) implies that  $\frac{V_0-L}{V_0} < 0$ . The plot shows that, while the homoscedastic strategy is independent of



Figure 5.3: Optimal initial risky allocations for the HN-GARCH and the homoscedastic model with the parameter values as of Table 5.1 and  $\gamma = 2$ , dependent on the time horizon T.

T, the initial HN-GARCH strategy decreases significantly for short horizons below half a year. Since there are no significant changes for longer time horizons, we limit Figure 5.3 to  $T \leq 1$ .



(a) Efficient frontiers for a one-year horizon.

(b) Efficient frontiers for a ten-year horizon.

Figure 5.4: Efficient frontiers for the HN-GARCH and the homoscedastic model with the parameter values as of Table 5.1. One year is assumed to have 252 trading days, the results are annualized.

Figure 5.4 targets the changes in the difference between the two models as the time horizon varies. Both plots show that the HN-GARCH model outperforms the homoscedastic variant. However, the difference increases significantly as the time horizon is extended to ten years, while it is relatively small for one year with 252 trading days. Overall, the results emphasize the key finding that incorporating the heteroscedasticity of log returns in the underlying model is of significant importance for investors.

5.6 Conclusion

# 5.6 Conclusion

In this chapter, we present a dynamic portfolio optimization problem within the class of affine GARCH models for an investor maximizing a HARA utility function under significantly generalized conditions. Theorem 5.1 not only extends the family of utilities in use but also solves the problem for previously excluded ranges of risk aversion. Choosing the exponent of the utility function to be 2, we establish a connection to mean-variance theory, which we analyze numerically using real-world market data. Based on 20 years of S&P 500 log returns, we estimate the parameters both for an HN-GARCH model and a homoscedastic variant, and show that the heteroscedastic strategy clearly dominates by calculating the efficient frontier. The numerical results suggest that the advantage of a heteroscedastic investor is larger if longer time horizons are considered.

# 6 Expected Utility Theory within Lévy GARCH Models

# 6.1 Introduction

In this chapter, we extend the expected utility theory (EUT) approach presented in Chapter 3 to a different model class, allowing for jumps in the asset price process. While we considered only one – potentially non-Gaussian – innovation per time step in the setting of the previous chapters, the model for the underlying price process used here allows for a Gaussian and an additional jump component at each time step. As visualized in Figure 6.1, we thus add more complexity concerning the model for the underlying asset. This chapter was submitted for publication in a similar form (Escobar-Anel et al., 2023).



Figure 6.1: Overview of the contributions of this dissertation with regard to portfolio optimization under affine GARCH models, increasing the model complexity as well as the complexity of terminal and intermediate utility, see also Figures 1.1, 3.1, 4.1 and 5.1. The orange box coresponds to Chapter 3, the green box to Chapter 4, the purple box to Chapter 5, and the blue box to Chapter 6.

Jumps have been a topic of immense interest in portfolio optimization for decades, particularly in a continuous-time setting thanks to analytical solutions – see, for instance, Merton (1971). Since empirical evidence suggests that financial markets exhibit occasional large price movements, considering jump components allows investors to model the underlying asset price movements more adequately with the corresponding improvement in financial decisions, from derivative pricing to risk analysis and portfolio management. On the other hand, models are desired to remain mathematically tractable and, in the best case, deliver closed-form expressions for pricing and portfolio optimization problems, a property that is easily imperiled.

The first affine GARCH models by Heston and Nandi (2000) and Christoffersen et al. (2006) overcame the lack of analytical solutions, but did not account for potential jumps in the underyling processes. The most common approach to extending these models and incorporate jumps consists of adding a (dynamic) Compound Poisson component in addition to the normal noise. In this manner, Christoffersen et al. (2012) introduced a model of GARCH type with stochastic volatility and separate dynamics for jumps, related to the non-affine work by Maheu and McCurdy (2004) and Duan et al. (2006a,b). In their version, however, Christoffersen et al. (2012) allow for jumps not only in the dynamics of the log-price process but also in the variance of the normal innovation and in the jump intensity, jeopardizing the exponentially affine form of the moment generating function (MGF). Ornthanalai (2014) formulated the corresponding model with jumps incorporated in the return dynamics only, preserving the affine nature of the model. The model includes two contemporaneously independent innovations: A Gaussian innovation and a Lévy jump component. The general framework for the latter allows for finite and infinite-activity jumps and extends the range of jump innovations beyond the Compound Poisson approach. This framework has been generalized recently and forms a special case of the affine multi-factor model introduced by Augustyniak et al. (2023).

Both the HN-GARCH and the Lévy GARCH framework were originally introduced for option pricing; Escobar-Anel et al. (2022a, 2021) first used the form of the generating function to derive the solution to a discrete-time portfolio optimization problem based on the jump-free model formulation by Badescu et al. (2019), as presented in Chapter 3. This chapter extends this approach to the Lévy GARCH framework and thus provides the opportunity to study the relevance of modeling jumps for portfolio optimization problems in the affine GARCH environment.

The impact of capturing extreme events on investment strategies has been studied extensively, although mostly in continuous time.<sup>1</sup> The analysis by Liu et al. (2003) suggests that price jumps have a larger effect than jumps in volatility. Ascheberg et al. (2016) claim that jumps matter more if the expected jump size or the jump intensity is large. These lines serve as a guideline for creating extreme jump settings to challenge the capability of jump-free models to deliver comparable investment strategies. Our paper differs from the existing literature on one key aspect. We take advantage of the capacity of the HN-GARCH model to deliver accurate estimates of the parameters (cf. Escobar-Anel et al., 2022b) in order to assess if the portfolio of an investor with a simple model could perform as well as that of an investor with the most advanced Lévy GARCH model. In other words, given that the true model of reality is unknown, we act as practical investors, working with a simple model to make decisions. This allows us to check in a control experiment if a simple, albeit wrong, model could match the performance of a correct advanced model. Our conclusions are eye-opening concerning the power of simple models in a discrete-time portfolio optimization setting.

Many authors have investigated the role of non-Gaussianity in asset returns in the context of portfolio selection. One prominent example is the work by Bekaert et al. (1998) suggesting that it is rather immaterial for an investor considering emerging market allocations to accommodate skewness and excess kurtosis. The crucial point that a wide range of expected utility problems can be well approximated by a mean-variance approach was summarized by Markowitz (2012).

<sup>&</sup>lt;sup>1</sup>We are unaware of any work investigating the impact of jumps in discrete-time portfolio optimization.

Our results are not contradictory with the continuous-time literature for three main reasons. Firstly, a continuous-time investor, although hypothetical, could, in principle, realize small benefits between a true and a wrong model more frequently. Such dynamics could be very profitable in the long term, and a detailed analysis in this regard could certainly be subject to further research. Secondly, most studies of complex models rely on sensitivity analyses for the parameters. However, the same model with different parameters does not describe the same reality, so results could be misleading. Lastly, even when two models are estimated on the same data, small sample sizes may incorporate estimation errors into the analysis, possibly creating a distorted conclusion since the estimation error could exacerbate a small difference in performance. In conclusion, our analyses provide a fresh perspective on the importance of jumps for portfolio decisions.

Our main contributions are:

- We fully develop a discrete-time dynamic portfolio optimization problem in an affine GARCH framework with jumps and provide a closed-form solution based on an established and reliable approximation of the self-financing condition.
- We apply the general result to three different types of jump structures: the Merton jump, the Normal Inverse Gaussian, and the Variance Gamma model. Furthermore, we derive the well-known HN-GARCH model as a jump-free special case of our setting.
- We study the impact of jumps on investment strategies via a comparison to the HN-GARCH model and a homoscedastic variant. We find that correctly calibrated jump-free models can imitate all three investigated jump models very well. This result is reflected in a low wealth-equivalent loss (WEL) for investors following suboptimal jump-free strategies, and it is robust with respect to more extreme jump environments.

The remainder of this chapter is organized as follows: In Section 6.2, we introduce the framework for the portfolio optimization problem, which is solved in Section 6.3, providing optimality conditions in general form and explicitly for three different types of jump increments. The numerical analysis follows in Section 6.4, targeting specifically the impact of jumps and a comparison to jump-free HN-GARCH and homoscedastic models. We conclude in Section 6.5.

# 6.2 The Lévy GARCH Model

Adopting the setting proposed by Ornthanalai (2014), we use two contemporaneously independent random variables  $z_t$  and  $y_t$  to model the log return  $X_t - X_{t-1} = \log \left( \frac{P_t^{(1)}}{P_{t-1}^{(1)}} \right)$  of the risky asset. While  $z_t$  follows a normal  $\mathcal{N}(0, h_{z,t})$  distribution,  $y_t$  is a pure Lévy jump innovation with generalized conditional Fourier transform

$$F_{y}(u; t-1, t) = \mathbb{E}_{t-1}[e^{uy_{t}}] = \exp\left\{\psi_{y}(u; t-1, t)\right\},\$$

where  $\psi$  denotes the cumulant exponent (or cumulant generating function) of  $y_t$ . The Lévy jump innovation  $y_t$  is assumed to be *time-homogeneous* in one of the parameters of its distribution, which we will denote by  $h_y$ , meaning that (Ornthanalai, 2014, Def. 1) the cumulant exponent is multiplicatively separable with respect to  $h_y$  and thus admits the representation

$$\psi_y(u; t-1, t) = h_{y,t} \cdot \xi_y(u), \tag{6.1}$$

for some  $\xi_y(u)$  independent of  $h_{y,t}$ . Note that the explicit distribution of  $y_t$  remains unspecified at this point. Potential choices include the Merton jump (MJ), a Normal-inverse Gaussian (NIG) or a Variance Gamma (VG) innovation. Note that the Gaussian innovation  $z_t$  is timehomogeneous in its variance. In particular, we have

$$\psi_z(u; t-1, t) = h_{z,t} \cdot \frac{u^2}{2}$$
, i.e.,  $\xi_z(u) = \frac{u^2}{2}$ .

Let the log-return process follow the dynamics given in the equation

$$R_t \coloneqq X_t - X_{t-1} = r_t + (\lambda_z - \xi_z(1)) \cdot h_{z,t} + (\lambda_y - \xi_y(1)) \cdot h_{y,t} + z_t + y_t, \tag{6.2}$$

where  $\xi_z$  and  $\xi_y$  are involved as convexity adjustments making  $z_t + y_t$  a martingale,  $r_t$  is timedeterministic, and  $\lambda_z h_{z,t} + \lambda_y h_{y,t}$  is referred to as the conditional equity premium (rate of return in excess of the risk-free rate). Note that by the properties of the cumulant generating function, Equation (6.2), in connection with the setting introduced above, implies that

$$Var_t [X_t - X_{t-1}] \equiv \sigma_t^2 = h_{z,t} + h_{y,t} \cdot \xi_y''(0).$$
(6.3)

Concerning the evolution of the parameters  $h_{z,t}$  and  $h_{y,t}$ , we follow Ornthanalai (2014) and Heston and Nandi (2000), respectively, and assume the following dynamics:

$$h_{z,t+1} = \omega_z + b_z \cdot h_{z,t} + \frac{a_z}{h_{z,t}} \left( z_t - c_z h_{z,t} \right)^2,$$
(6.4a)

$$h_{y,t+1} = \omega_y + b_y \cdot h_{y,t} + \frac{a_y}{h_{z,t}} \left( z_t - c_y h_{z,t} \right)^2.$$
 (6.4b)

This is equivalent to the evolution of the conditional variance in the HN-GARCH model by Heston and Nandi (2000), adapted for non-standardized innovations, but nonetheless converging to the Cox-Ingersoll-Ross process if the length of the time step is shrunk to zero (cf. Badescu et al., 2019).

Following the approach by Campbell and Viceira (1999) and Escobar-Anel et al. (2021) presented in Chapter 3, we use a second-order approach in order to approximate the self-financing condition. With (6.3) as the variance of the log return, we obtain

$$W_{t} = W_{t-1} + \pi_{t-1}R_{t} + \frac{1}{2} \left(\pi_{t-1} - \pi_{t-1}^{2}\right) \sigma_{t}^{2} + (1 - \pi_{t-1}) r_{t}$$
  
=  $W_{t-1} + \pi_{t-1}R_{t} + \frac{1}{2} \left(\pi_{t-1} - \pi_{t-1}^{2}\right) \left(h_{z,t} + h_{y,t}\xi_{y}''(0)\right) + (1 - \pi_{t-1}) r_{t}.$  (6.5)

In order to derive a representation of the maximum expected utility later, we investigate the generating function of the log-price process under the physical measure. For enhanced readability, proofs are postponed to Appendix D.1. We establish the following result:

**Proposition 6.1.** Let T be some finite time horizon and let t < T - 1. In the setting of a Lévy GARCH model as introduced above, the one-step multivariate generating function of the log return  $R_{t+1}$  and the homogeneous parameters  $h_{y,t+2}$  and  $h_{z,t+2}$ , under the physical measure, conditioned on the information available at time t, is given by

$$\Psi_{(R_{t+1}, h_{z,t+2}, h_{y,t+2})}(u, v_z, v_y \mid \mathcal{F}_t) \coloneqq \mathbb{E}_t \left[ \exp \left\{ uR_{t+1} + v_z h_{z,t+2} + v_y h_{y,t+2} \right\} \right]$$
  
=  $\exp \left\{ A(u, v_z, v_y; t, t+1) + B(u, v_z, v_y; t, t+1) + h_{z,t+1} + B(u, v_z, v_y; t, t+1) + h_{z,t+1} + C(u, v_y; t, t+1) + h_{y,t+1} \right\},$  (6.6)
where we define

$$A(u, v_z, v_y; t, t+1) = ur_{t+1} + v_z \omega_z + v_y \omega_y - \frac{1}{2} \log \left(1 - 2 \left(v_z a_z + v_y a_y\right)\right),$$
(6.7a)  
$$B(u, v_z, v_y; t, t+1) = u \left(\lambda_z - \xi_z(1)\right) + v_z \left(b_z + a_z c_z^2\right)$$

$$t, t+1) = u \left(\lambda_z - \xi_z(1)\right) + v_z \left(b_z + a_z c_z^2\right) + v_y a_y c_y^2 + \frac{\left(u - 2v_z a_z c_z - 2v_y a_y c_y\right)^2}{2\left(1 - 2\left(v_z a_z + v_y a_y\right)\right)},$$
(6.7b)

$$C(u, v_y; t, t+1) = u(\lambda_y - \xi_y(1)) + v_y b_y + \xi_y(u).$$
(6.7c)

In the sequel, we will omit the arguments t and t+1 from A, B, and C if the use of the one-step generating function is clear.

## 6.3 Optimal Investment Maximizing CRRA Utility

### 6.3.1 Maximum Utility Representation

The main result concerning dynamic portfolio optimization in the setting introduced in Section 6.2 is presented in Theorem 6.2. Again, for the sake of readability, the proofs in this section are postponed to Appendix D.1.

**Theorem 6.2** (Maximum utility representation). Assume that the decision maker maximizes a power utility function with corresponding parameter  $\gamma < 0$ . Furthermore, let the log price of the risky asset be described by a Lévy GARCH model as outlined above, satisfying the additional condition that  $h_{y,t+1} = \Lambda \cdot h_{z,t+1}$  holds for some  $\Lambda \in \mathbb{R}$ , for all  $z_t \in \mathbb{R}$  and  $t \in \{0, \ldots, T-1\}$ . Then, the optimal value function at time t can be written as

$$\phi_t (W_t, h_{z,t+1}, h_{y,t+1}) = \frac{1}{\gamma} \exp \left\{ D_{t,T}(\pi_t^*) + \gamma W_t + E_{t,T}^z(\pi_t^*) \cdot h_{z,t+1} + E_{t,T}^y(\pi_t^*) \cdot h_{y,t+1} \right\},$$

$$= \frac{1}{\gamma} \exp \left\{ D_{t,T}(\pi_t^*) + \gamma W_t + \left[ E_{t,T}^z(\pi_t^*) + \Lambda E_{t,T}^y(\pi_t^*) \right] \cdot h_{z,t+1} \right\},$$
(6.8)

with  $D_{t,T}$ ,  $E_{t,T}^z$  and  $E_{t,T}^y$  given via  $D_{T,T} = E_{T,T}^z = E_{T,T}^y = 0$  and the recursive equations

$$D_{t,T}(\pi_t^*) = D_{t+1,T}(\pi_{t+1}^*) + (1 - \pi_t^*) \ \gamma r_{t+1} + A\left(\gamma \pi_t^*, \ E_{t+1,T}^z(\pi_{t+1}^*), \ E_{t+1,T}^y(\pi_{t+1}^*)\right), \quad (6.9a)$$

$$E_{t,T}^{z}(\pi_{t}^{*}) = \frac{\gamma}{2} \left( \pi_{t}^{*} - (\pi_{t}^{*})^{2} \right) + B \left( \gamma \pi_{t}^{*}, E_{t+1,T}^{z}(\pi_{t+1}^{*}), E_{t+1,T}^{y}(\pi_{t+1}^{*}) \right),$$
(6.9b)

$$E_{t,T}^{y}(\pi_{t}^{*}) = \frac{\gamma}{2} \left( \pi_{t}^{*} - (\pi_{t}^{*})^{2} \right) \cdot \xi_{y}^{\prime\prime}(0) + C \left( \gamma \pi_{t}^{*}, E_{t+1,T}^{y}(\pi_{t+1}^{*}) \right).$$
(6.9c)

The optimal solution  $\pi_t^*$  satisfies

$$\begin{bmatrix} \lambda_z - \pi_t^* + \frac{\gamma \pi_t^* - 2E_{t+1,T}^z \left(\pi_{t+1}^*\right) a_z c_z - 2E_{t+1,T}^y \left(\pi_{t+1}^*\right) a_y c_y}{1 - 2E_{t+1,T}^z \left(\pi_{t+1}^*\right) a_z - 2E_{t+1,T}^y \left(\pi_{t+1}^*\right) a_y} \end{bmatrix} + \Lambda \cdot \begin{bmatrix} \xi_y''(0) \left(\frac{1}{2} - \pi_t^*\right) + (\lambda_y - \xi_y(1)) + \xi_y' \left(\gamma \pi_t^*\right) \end{bmatrix} = 0,$$
(6.10)

together with the second-order condition

$$\Lambda \cdot \xi_y''(\gamma \pi_t^*) > \frac{1 + \Lambda \xi_y''(0)}{\gamma} - \frac{1}{1 - 2E_{t+1,T}^z(\pi_{t+1}^*) a_z - 2E_{t+1,T}^y(\pi_{t+1}^*) a_y}.$$
 (6.11)

#### 6 Expected Utility Theory within Lévy GARCH Models

In the case of  $h_{y,t+1} \neq \Lambda \cdot h_{z,t+1}$ , the optimal strategy needs to satisfy two equations to meet the corresponding optimality condition (explicitly given as (D.5) in the proof) for any values of  $h_{z,t+1}$  and  $h_{y,t+1}$ . In particular, both coefficients in square brackets in (6.10) need to be set to zero. Since the coefficient corresponding to  $h_{z,t+1}$  (first square bracket) is linear in  $\pi_t$ , we obtain an explicit expression for the optimal strategy dependent on the other parameters. Note that this solution candidate – name it  $\pi_t^{(z)}$  – is time-dependent because of  $E_{t+1,T}^{z,*}$  and  $E_{t+1,T}^{y,*}$ .<sup>2</sup> Concerning the coefficient corresponding to  $h_{y,t+1}$  (second square bracket), this yields a condition for  $\xi_y$ , stating that

$$\xi_y''(0)\left(\frac{1}{2} - \pi_t^{(z)}\right) + (\lambda_y - \xi_y(1)) + \xi_y'\left(\gamma \pi_t^{(z)}\right) = 0, \tag{6.12}$$

for all  $t \in \{0, ..., T-1\}$ . These T different conditions, however, need to be satisfied for all investors, i.e., for all possible choices for  $\gamma$  and the corresponding values of  $\pi_t^{(z)}$ . Thus, we need to look at (6.12) as a differential equation for  $\xi_y$ , which after the substitution  $u = \gamma \pi_t^{(z)}$  has the form

$$\xi_y'(u) = \frac{k_2}{\gamma}u + k_1 - \frac{k_2}{2} - \lambda_y, \tag{6.13}$$

subject to the two conditions  $\xi_y(1) = k_1$  and  $\xi''_y(0) = k_2$ . This still – counter-intuitively – imposes a condition on the statistical properties of log-asset returns related to the investor's risk aversion parameter  $\gamma$ . Furthermore, integrating yields

$$\xi_y(u) = \frac{k_2}{2\gamma}u^2 + \left(k_1 - \frac{k_2}{2} - \lambda_y\right)u + C$$
(6.14)

for some constant  $C \in \mathbb{R}$ , with

$$\xi_y(1) = \frac{k_2}{2\gamma} + k_1 - \frac{k_2}{2} - \lambda_y + C = k_1, \qquad (6.15a)$$

$$\xi_y''(0) = \frac{k_2}{\gamma} = k_2. \tag{6.15b}$$

It is easy to see that (6.15b) requires  $\gamma = 1$  for the risk aversion parameter, which implies  $C = \lambda_y$  from (6.15a). Hence, this restricts the analysis to risk-neutral investors and contradicts our assumption that  $\gamma < 0$ .

An analog to our condition  $h_{y,t} = \Lambda h_{z,t}$  can be found, e.g., in the setting by Liu and Pan (2003), where the stochastic jump intensity is connected to the instantaneous variance process. In practice, we ensure the linear dependence of time-homogeneous factors by connecting the parameters in (6.4b) to the ones in (6.4a), i.e., by assuming the initial condition  $h_{y,1} = \Lambda h_{z,1}$  together with

$$c_y = c_z, \quad b_y = b_z, \quad \omega_y = \Lambda \omega_z, \quad \text{and} \quad a_y = \Lambda a_z,$$
(6.16)

implying a slightly simplified expression for (6.10). We emphasize that a simple reduction of our model delivers a well-known special case without jumps:

 $<sup>{}^{2}</sup>E_{t+1,T}^{x,*}$  is short for  $E_{t+1,T}^{x}(\pi_{t+1}^{*})$ , with  $x \in \{y, z\}$ 

**Corollary 6.3** (HN-GARCH). Disregarding the jump innovation, i.e., setting  $\xi_y(u) = 0$ ,  $\Lambda = 0$ and  $\lambda_y = 0$ , yields the HN-GARCH model.<sup>3</sup> Note that due to the convexity adjustment present in (6.2), we need to choose  $\lambda_z = \lambda_{\text{HN}} + \frac{1}{2}$ , where  $\lambda_{\text{HN}}$  is the maximum likelihood estimate (MLE) for the risk premium in the HN-GARCH model. In particular, (6.10) leads to

$$\pi_t^* = \frac{\lambda_z \left( 1 - 2E_{t+1,T}^{z,*} a_z \right) - 2E_{t+1,T}^{z,*} a_z c_z}{1 - 2E_{t+1,T}^{z,*} a_z - \gamma}, \tag{6.17}$$

and  $1 - 2E_{t+1,T}^{z,*}a_z > 0$  ensures (6.11).

Note that the HN-GARCH model satisfies the second-order condition

$$\Lambda \xi_{y}^{\prime\prime}(\gamma \pi_{t}) > \frac{1 + \Lambda \xi_{y}^{\prime\prime}(0)}{\gamma} - \frac{1}{1 - 2E_{t+1,T}^{z}(\pi_{t+1}^{*}) a_{z} - 2E_{t+1,T}^{y}(\pi_{t+1}^{*}) a_{y}}$$
(6.18)

on the entire real line for  $\pi_t$ , i.e., we know that the objective function is concave in this case, and the solution to the first-order condition is a global maximum.

### 6.3.2 Examples of Jump Innovations

This section considers several choices for the Lévy jump innovation and its distribution. First, the above optimality problem is investigated under the well-known MJ model. Subsequently, we also tackle NIG and VG innovations.

#### Merton Jump Model

In the Merton jump (MJ) model (cf. Ornthanalai, 2014), each jump size is drawn independently from  $\mathcal{N}(\theta, \delta^2)$ . The number of jumps  $\eta_{t+1}$  occurring in the interval (t, t+1] is a Poisson process with predictable intensity  $h_{y,t+1} = \Lambda \cdot h_{z,t+1}$ , i.e., this intensity serves as the time-homogeneous parameter in our Lévy GARCH model. Thus, let  $y_{t+1} = \sum_{k=1}^{\eta_{t+1}} Y_k$ , with  $\{Y_k\}_k$  i.i.d. according to  $\mathcal{N}(\theta, \delta^2)$ . Note that

$$\mathbb{E}_t\left[\exp\left\{u\cdot y_{t+1}\right\} \middle| \eta_{t+1} = n\right] = \exp\left\{\left(u\theta + \frac{1}{2}\delta^2 u^2\right)\cdot n\right\},\$$

which allows us to deduce from the tower property that

$$\mathbb{E}_t \left[ \exp\left\{ u \cdot y_{t+1} \right\} \right] = \exp\left\{ \left( \exp\left\{ u\theta + \frac{1}{2}\delta^2 u^2 \right\} - 1 \right) \cdot h_{y,t+1} \right\}.$$

Having performed the above analysis conditioned on the information at time t, we obtain the cumulant exponent from the last equation and state that in the Merton jump model,

$$\xi_y(u) = \exp\left\{u\theta + \frac{1}{2}\delta^2 u^2\right\} - 1.$$
 (6.19)

<sup>&</sup>lt;sup>3</sup>The HN-GARCH model was originally introduced by Heston and Nandi (2000). The notation  $\lambda_{\rm HN}$  thus refers to this original formulation without convexity adjustment, see, e.g., Chapter 3. Contrary to the previous chapters, this chapter will use the HN-GARCH model in the Lévy GARCH parametrization for clarity.

Transferring this representation back to our optimality equation (6.10), we now need to solve

$$\begin{bmatrix} \lambda_z - \pi_t + \frac{\gamma \pi_t - 2\left(E_{t+1,T}^{z,*} - \Lambda E_{t+1,T}^{y,*}\right) a_z c_z}{1 - 2a_z \left(E_{t+1,T}^{z,*} + \Lambda E_{t+1,T}^{y,*}\right)} \end{bmatrix} + \Lambda \cdot \left[ \left(\theta^2 + \delta^2\right) \left(\frac{1}{2} - \pi_t\right) + \left(\lambda_y - \exp\left\{\left(\theta + \frac{1}{2}\delta^2\right)\right\} + 1\right) + \left(\theta + \delta^2 \gamma \pi_t\right) \cdot \exp\left\{\theta \gamma \pi_t + \frac{1}{2}\delta^2 \gamma^2 \pi_t^2\right\} \right] = 0.$$

$$(6.20)$$

The optimality equation above yields the following statement on existence and uniqueness of an optimal solution in the prevailing model:

**Corollary 6.4.** Suppose (6.18) holds with strict inequality for all  $t \in \{0, ..., T-1\}$  in the Merton jump model with normally distributed jump size. Then, a unique solution exists for the optimal relative portfolio strategy  $\pi_t^*$ .

However, the strategy  $\pi_t$  is involved both in linear and in exponential terms, making (6.20) difficult to solve for  $\pi_t$  explicitly. One straightforward way to an approximate solution uses that the values for the parameters  $\theta$  and  $\delta$  are very small (referring to the MLEs reported by Ornthanalai, 2014), which implies that the approximation  $\exp\{x\} \approx 1 + x$  is sufficiently good in this case.

#### Normal Inverse Gaussian Model

A Normal-inverse Gaussian (NIG) process (Barndorff-Nielsen, 1997; Ornthanalai, 2014) can be constructed via a Brownian motion, which is time-changed using an inverse Gaussian process. In particular, we introduce  $\alpha > 0$  and  $\beta$  with  $|\beta| < \alpha$ . In order to obtain an innovation that is time-homogeneous in one of its parameters later, we refer to Barndorff-Nielsen (1997) and set the starting point for the bivariate Brownian motion, which is used to construct the NIG process, to zero (corresponding to  $\mu = 0$  in the original parametrization). This then leads to:

$$\psi_y\left(u;t,t+1\right) = -\left(\sqrt{\alpha^2 - (\beta + u)^2} - \sqrt{\alpha^2 - \beta^2}\right) \cdot h_{y,t+1},$$

with the part excluding  $h_{y,t+1}$  (note the minus in front!) as  $\xi_y(u)$ . In this setting too, we rely on the condition  $h_{y,t+1} = \Lambda \cdot h_{z,t+1}$ , which then yields the following optimality equation:

$$\begin{bmatrix} \lambda_{z} - \pi_{t} + \frac{\gamma \pi_{t} - 2 \left( E_{t+1,T}^{z,*} - \Lambda E_{t+1,T}^{y,*} \right) a_{z} c_{z}}{1 - 2 a_{z} \left( E_{t+1,T}^{z,*} + \Lambda E_{t+1,T}^{y,*} \right)} \end{bmatrix} + \Lambda \cdot \begin{bmatrix} \frac{\alpha^{2} - 2\beta^{2}}{(\alpha^{2} - \beta^{2})^{\frac{3}{2}}} \left( \frac{1}{2} - \pi_{t} \right) + \lambda_{y} + \sqrt{\alpha^{2} - (\beta + 1)^{2}} - \sqrt{\alpha^{2} - \beta^{2}} \\ + \frac{\beta + \gamma \pi_{t}}{\sqrt{\alpha^{2} - \beta^{2} - 2\beta\gamma\pi_{t} - \gamma^{2}\pi_{t}^{2}}} \end{bmatrix} = 0.$$
(6.21)

#### Variance Gamma Model

Another potential choice concerning the model for the jump innovation is the increment of a Variance Gamma (VG) process (Madan and Seneta, 1990). Following Ornthanalai (2014), we

consider a normal density  $\mathcal{N}(\theta x, x)$ , conditional on  $x \sim \text{Gamma}(\alpha, \beta)$ , for  $\alpha, \beta > 0$ . Again referring to Ornthanalai (2014), we obtain via conditioning on x:

$$\mathbb{E}_t \left[ \exp \left\{ u \cdot y_{t+1} \right\} \right] = \left( \frac{\beta}{\beta - \frac{1}{2}u^2 - \theta u} \right)^{\alpha} \\ = \exp \left\{ -\alpha \cdot \log \left( 1 - \frac{\theta u + \frac{1}{2}u^2}{\beta} \right) \right\}.$$

Thus, choosing  $h_{y,t+1}$  to take the role of  $\alpha$ , we arrive at

$$\xi_y(u) = \log\left(\frac{\beta}{\beta - \frac{1}{2}u^2 - \theta u}\right).$$
(6.22)

Using this in (6.10) yields the optimality equation

$$\left[\lambda_z - \pi_t + \frac{\gamma \pi_t - 2\left(E_{t+1,T}^{z,*} - \Lambda E_{t+1,T}^{y,*}\right) a_z c_z}{1 - 2a_z \left(E_{t+1,T}^{z,*} + \Lambda E_{t+1,T}^{y,*}\right)}\right] + \Lambda \cdot \left[\frac{\beta + \theta^2}{\beta^2} \left(\frac{1}{2} - \pi_t\right) + \lambda_y - \log\left(\frac{\beta}{\beta - \theta - \frac{1}{2}}\right) + \frac{\theta + \gamma \pi_t}{\beta - \theta \gamma \pi_t - \frac{1}{2}\gamma^2 \pi_t^2}\right] = 0.$$
(6.23)

Multiplying both sides with the denominator of the last term leads to a third-order polynomial.

**Remark 6.5.** Although Equations (6.20), (6.21) and (6.23) seem to be tedious to solve for  $\pi_t$ , we can check numerically that the second-order condition is satisfied for the entire real line at all time points (for reasonable parameter values), i.e., that the objective function is concave in  $\pi_t$  and that a solution will be a unique maximum.

## 6.4 Numerical Analysis

This section is devoted to the numerical study of the prevailing Lévy GARCH model. In particular, the sensitivity of the optimal portfolio allocation to the central parameters of the model is investigated, as well as the WEL occurring when the investor follows suboptimal strategies. We are interested specifically in the impact of jumps on the optimal strategy and on potential losses and want to compare the Lévy GARCH models to the nested HN-GARCH (which does not incorporate jumps) and a homoscedastic variant. All analyses are going to be presented for the MJ model as well as for NIG and VG innovations.

If not specified otherwise, the choices for the model's investment parameters are given in Table 6.1.

Table 6.1: Standard values for the investment parameters used for the plots in Section 6.4.

Parameter	Т	r	$\gamma$
Value	252	0.01/252	-1

### 6.4.1 Parameter Estimates

Our analysis is based on the MLEs provided by Ornthanalai (2014). Since in the estimation process for these MLEs the linear dependence between the two time-homogeneous parameters,  $h_{y,t} = \Lambda h_{z,t}$  for  $t \in \{1, \ldots, T\}$ , was not assumed, we need to modify the original parameter set. The central question in this regard is how to derive the value for the parameter  $\Lambda$ , since this, by construction, will govern the behavior of the process  $\{h_{y,t}\}_t$ . The approach presented here uses the unconditional mean of the process  $\{h_{y,t}\}_t$  and derives a value for  $\Lambda$  such that the unconditional means of  $\Lambda h_{z,t}$  and  $h_{y,t}$  are matched. Assuming that both time-homogeneous parameters  $h_z$  and  $h_y$  satisfy the stationarity conditions  $b_z + a_z c_z^2 < 1$  and  $b_y < 1$ , the unconditional first moments of  $h_z$  and  $h_y$ , respectively, are given by (Ornthanalai, 2014, Equation (9))

$$\bar{h}_{z}^{(1)} = \frac{\omega_{z} + a_{z}}{1 - b_{z} - a_{z}c_{z}^{2}}, \qquad \bar{h}_{y}^{(1)} = \frac{\omega_{y} + a_{y} + a_{y}c_{y}^{2}h_{z}^{(1)}}{1 - b_{y}}.$$
(6.24)

- - (1)

We replace the parameters  $a_y$ ,  $b_y$ ,  $c_y$  and  $\omega_y$  in (6.24) according to the relations in (6.16) to obtain the  $\Lambda$ -dependent version  $\hat{h}_y^{(1)}$ . Subsequently, we choose the value for  $\Lambda$  such that  $\hat{h}_y^{(1)}$ and  $\bar{h}_y^{(1)}$  are equal, leading to the equation

$$\frac{\Lambda\omega_z + \Lambda a_z + \Lambda a_z c_z^2 \bar{h}_z^{(1)}}{1 - b_z} = \frac{\omega_y + a_y + a_y c_y^2 \bar{h}_z^{(1)}}{1 - b_y},$$
(6.25)

which is equivalent to

$$\Lambda = \frac{(1 - b_z) \left(\omega_y + a_y + a_y c_y^2 \bar{h}_z^{(1)}\right)}{(1 - b_y) \left(\omega_z + a_z + a_z c_z^2 \bar{h}_z^{(1)}\right)}.$$
(6.26)

Note that one could also include higher unconditional moments in the above approach and then derive the best value for  $\Lambda$ , e.g., based on the method of least squares. This, however, would sacrifice an exact match of the long-term mean of  $h_y$ , forming the reason why we remain with the exact choice in (6.26).

### 6.4.2 Methodology for Comparisons and WEL

In order to correctly investigate the impact of jumps on the solution to our portfolio optimization problem, we want to compare the prevailing jump model to the closest HN-GARCH model, which is nested in our setup as described in Corollary 6.3 and does not incorporate jumps. In order to avoid estimation errors from small samples, for every jump model, we simulate 100 paths with a long sample of  $N = 2 \times 10^4$  steps of log returns for the prevailing jump model and then derive MLEs for the HN-GARCH parameters based on this dataset. To this end, we follow the lines of the returns-only estimation of Escobar-Anel et al. (2022b) that we described in Section 4.4.2, exploiting the normal distribution of the HN-GARCH innovations to obtain the log-likelihood function l. For one path of log returns  $\{x_n\}_{n=1,...,N}$ , the estimates  $\hat{\theta}$  for  $\theta = (\lambda_z, \omega_z, a_z, b_z, c_z)$  are given by

$$\hat{\theta} = \arg \max_{\theta} l(x_1, \dots, x_N \mid \theta).$$
(6.27)

After finding estimates  $\hat{\theta}_1, \ldots, \hat{\theta}_{100}$  for all paths, we take the median as the best proxy for the HN parametrization<sup>4</sup>, which can be understood as the best HN-GARCH approximating a given Lévy GARCH. The same dataset is used to obtain the corresponding homoscedastic Gaussian model (HS), i.e., an HN-GARCH model with  $a_z = b_z = c_z = 0$ . In the following, the jump models – we use the MJ, the NIG and the VG model as examples – are always compared to their re-calibrated versions of the HN-GARCH and the homoscedastic variant.

We use the exponentially affine value function from Theorem 6.2 to measure investors' losses when following suboptimal strategies. In particular, we compare the optimal value function at time t,  $\phi_t$ , which is obtained via using the optimal strategy  $\{\pi_t^s\}_t$  in the recursive parameters, to the expected utility corresponding to the suboptimal strategy  $\{\pi_t^s\}_t$ , denoted by  $\phi_t^s$ . The WEL  $L_t^s$  is the percentage of wealth at time t that an investor following the optimal strategy can forgo in order to obtain the same expected utility as another investor following the suboptimal strategy  $\{\pi_t^s\}_t$ . Thus,  $L_t^s$  satisfies the equation

$$\phi_t^s \left( \log V_t, h_{z,t+1}, h_{y,t+1} \right) = \phi_t \left( \log \left( V_t \left( 1 - L_t^s \right) \right), h_{z,t+1}, h_{y,t+1} \right).$$
(6.28)

Using (6.8) and exploiting  $h_{y,t+1} = \Lambda h_{z,t+1}$  yields the following closed-form representation of the WEL:

$$L_{t}^{s} = 1 - \exp\left\{\frac{1}{\gamma} \cdot \left[D_{t,T}^{s}\left(\pi_{t}^{s}\right) - D_{t,T}\left(\pi_{t}^{*}\right) + \left(E_{t,T}^{z,s}\left(\pi_{t}^{s}\right) - E_{t,T}^{z}\left(\pi_{t}^{*}\right) + \Lambda E_{t,T}^{y,s}\left(\pi_{t}^{s}\right) - \Lambda E_{t,T}^{y}\left(\pi_{t}^{*}\right)\right)h_{z,t+1}\right]\right\}.$$
(6.29)

### 6.4.3 Results for the Merton Jump Model

This subsection presents an analysis of the sensitivity of the optimal solution to the most relevant investment parameters and aims at comparing the Merton jump model to the closest HN-GARCH and homoscedastic solution. To this end, we derived MLEs for the latter two jump-free models according to the methodology presented in Section 6.4.2, based on the MJ parameters from Ornthanalai (2014), adapted to our setting as described in Section 6.4.1. We add that, compared to the sample statistics for the original S&P 500 dataset used to estimate the MJ parameters in Table 6.2, namely a sample skewness of -0.194 and a kurtosis of 11.15 (Ornthanalai, 2014), the first 15 years of observations in our artificial dataset created with these MJ parameters yield a sample skewness of 0.010 and a kurtosis of 3.47.<sup>5</sup> For all three investigated models, similar sample statistics are obtained when simulating with or without modifying the parameter set according to Section 6.4.1 to achieve the dependence in (6.16).

Comparing the solutions to the portfolio optimization problem for MJ to its no-jump analogues as of Table 6.2, Figure 6.2 shows that the re-calibrated HN-GARCH model without jumps replicates the MJ very well, with the difference in the optimal initial allocation decreasing as shorter time horizons are considered. In contrast to this, the homoscedastic solution is not sensitive to the length of the time horizon and stays constant, also causing an increasing difference to the MJ as the time horizon increases.

Figure 6.3a addresses the sensitivity of the optimal solution with respect to the investor's level of relative risk aversion, which in the prevailing setting of a power utility function with parameter

<sup>&</sup>lt;sup>4</sup>We use the median rather than the average due to outliers in the estimates (i.e., potential local minima on some samples). We also report the standard error from the 100 samples to confirm the estimate's accuracy.

<sup>&</sup>lt;sup>5</sup>While these numbers for the artificial set of MJ returns are less extreme than the actual sample statistics of the original dataset, this picture changes when looking at different Lévy GARCH models, see Section 6.4.4 for the NIG model.

Table 6.2: MLEs for the MJ model and the recalibrated HN-GARCH and homoscedastic model, respectively. Standard errors for the recalibrated models are in parentheses. The MJ parameters are based on Ornthanalai (2014) and the methodology presented in Section 6.4.1, the two jump-free models have been estimated according to Section 6.4.2.

Par.	MJ	HN		HS	
$\lambda_z$	1.47	2.67	$(2.70\times10^{-1})$	2.66	$(1.85 \times 10^{-1})$
$\omega_z$	$-2.39\times10^{-6}$	$-1.23\times10^{-6}$	$(5.67\times 10^{-7})$	$6.88\times10^{-4}$	$(3.50 \times 10^{-5})$
$a_z$	$2.98\times 10^{-6}$	$4.45\times10^{-6}$	$(5.80\times10^{-7})$		
$b_z$	$9.41 \times 10^{-1}$	$9.59\times10^{-1}$	$(5.59\times10^{-3})$		
$c_z$	$1.35\times 10^2$	$9.04\times10^{1}$	$(1.19\times 10^1)$		
$\lambda_y$	$6.18\times 10^{-2}$				
Λ	$2.31\times 10^2$				
θ	$-2.39\times10^{-3}$				
δ	$2.85 \times 10^{-2}$				



Figure 6.2: Values of the optimal strategy  $\pi_0^*$  in the MJ model and the corresponding re-calibrated HN-GARCH and homoscedastic model, dependent on the length of the time horizon, for T up to five years. The parameters are given in Table 6.1 (investment parameters) and in Table 6.2 (MJ, HN, HS).

 $\gamma < 1$  is given via  $\mathcal{R} = 1 - \gamma$ . Clearly, the fraction of wealth invested in the risky asset increases in all three models as the level of relative risk aversion decreases. This can lead to major differences between investors using the same model, but having different risk preferences. Following the MJ, for instance, with a relative risk aversion of  $\mathcal{R} \approx 1.3$  yields an initial risky allocation of around 200%, whereas  $\mathcal{R} \approx 4$  leads to an allocation below 100%. For the entire plotted range,



(a) Values of  $\pi_0^*$  dependent on  $\mathcal{R}$ . The lines for the MJ and the HN are overlapping.

(b) WEL dependent on  $\mathcal{R}$ .

Figure 6.3: Values of the optimal initial allocation  $\pi_0^*$  and of wealth-equivalent losses, dependent on the investor's level of relative risk aversion  $\mathcal{R} = 1 - \gamma \in [1.1, 5]$ . The parameters are given in Table 6.1 (investment parameters) and in Table 6.2 (MJ, HN, HS).

the HN-GARCH strategy remains very close to the optimal strategy in the MJ, while investors following the homoscedastic variant have a slightly lower risky allocation. Figure 6.3b uses the same parameter range and considers the WEL that investors incur when following the HN-GARCH or the homoscedastic strategy, while log returns actually evolve according to an MJ. In general, the losses remain at a very low level for a one-year time horizon, yielding values below 5 basis points for the homoscedastic model and even less for the HN-GARCH. This confirms the impression from Figures 6.2 and 6.3a that the jump-free models lead to very similar investment strategies.



(a) WEL dependent on  $\theta$ , up to an absolute increase (b) WEL dependent on  $\delta$ , up to an increase by a factor of 10. of 10.

Figure 6.4: Values of the wealth-equivalent losses, dependent on the jump parameters in the MJ model. For each grid point, both the HN-GARCH and the homoscedastic model are re-calibrated using the methodology in Section 6.4.2. The time horizon is one year with 252 trading days, and we consider an investor with  $\gamma = -1$ .

In Figure 6.4, we consider more extreme jump settings, which are obtained via increasing the absolute mean of the normally distributed jump size in the MJ up to a factor of 10 and via increasing the standard deviation of the jump size by the same factor. For each grid point between the original parameter estimate as of Table 6.2 and the most extreme setting, we recalibrate the jump-free models to a set of simulated log return according to Section 6.4.2.<sup>6</sup> Both

 $<sup>^{6}</sup>$ The recalibration exercise leads to natural inaccuracies in the estimates, which explains the minor variation in the calculation of WEL among grid points.

plots suggest that the capability of the HN-GARCH and the homoscedastic model to imitate the MJ is not significantly weakened, i.e., losses remain negligible for all considered parameter settings.

Besides modifications concerning the jump size in the MJ, the jump intensity can of course play a huge role too, bearing in mind, e.g., the analysis by Ascheberg et al. (2016), suggesting that jumps matter more if the intensity is large. In the Merton jump model, the intensity can easily be modified via the parameter  $\Lambda$ , governing the jump intensity given by  $\Lambda h_{z,t}$  in the interval (t, t + 1]. One has to take into account, however, that the long-run equity premium, obtained via

$$\bar{\lambda} \coloneqq \lambda_z \bar{h}_z^{(1)} + \lambda_y \bar{h}_y^{(1)} = \lambda_z \bar{h}_z^{(1)} + \lambda_y \Lambda \bar{h}_z^{(1)}, \tag{6.30}$$

changes accordingly, which will have a major impact on the optimal solution. Since we wish to isolate the effect of modifications to the jump size, we compensate for changes is the long-run equity premium by adapting  $\lambda_z$  such that  $\bar{\lambda}$  remains constant. Figure 6.5 uses this approach



Figure 6.5: Values of the wealth-equivalent losses for the HN-GARCH and the homoscedastic model, dependent on the jump intensity in the MJ model. For each grid point, both the HN-GARCH and the homoscedastic model are re-calibrated using the methodology in Section 6.4.2. The time horizon is one year with 252 trading days, and we consider an investor with  $\gamma = -1$ .

and increases  $\Lambda$  up to one order of magnitude from the original value in the MJ model (see Table 6.2). Again, both jump-free models are re-calibrated for every grid point. As seen for the parameters affecting the jump size, the parameter  $\Lambda$  does not have a significant impact on the WEL occurring for the closest HN-GARCH and homoscedastic models, as all values stay below or close to 2 basis points, with the HN-GARCH outperforming the constant solution and yielding even lower values.

To conclude our numerical investigation of the Merton jump model, we modify both the mean jump size and the intensity at the same time. In particular, we increase the parameters  $\theta$  and  $\Lambda$  by a factor of 10. Concerning changes in the jump intensity, implied by an increase in  $\Lambda$ , we again use the approach described above to avoid changes in the long-run equity premium. Figure 6.6a shows the values of the initial optimal risky allocation for investors with different risk aversion levels, for the MJ as well as for the two jump-free models. The plot once more suggests that the strategies obtained from the HN-GARCH and the homoscedastic model are similar to the one in the MJ. Concerning the WEL investors incur when following the former



(a) Values of  $\pi_0^*$  dependent on  $\mathcal{R}$ . (b) WEL dependent on  $\mathcal{R}$ .

Figure 6.6: Values of the optimal initial allocation  $\pi_0^*$  and of wealth-equivalent losses, dependent on the investor's level of relative risk aversion  $\mathcal{R} = 1 - \gamma \in [1.1, 5]$ . These plots are based on a modified, extreme parameter setting for the MJ with  $\theta$  and  $\Lambda$  one order of magnitude larger than the original estimates in Table 6.2. The HN-GARCH and the homoscedastic model have been re-calibrated according to Section 6.4.2. The time horizon is one year with 252 trading days.

two suboptimal strategies, Figure 6.6b suggests a very good performance of the re-calibrated models, yielding values below one basis point for the entire range of relative risk aversion.

#### 6.4.4 Results for the Normal Inverse Gaussian Model

In this subsection, we address the Normal Inverse Gaussian model in terms of the sensitivity of the optimal solution to the main model parameters and compare the model to the Gaussian HN-GARCH model without jumps and to a homoscedastic variant. The parameters for these three models are displayed in Table 6.3. The estimates for the NIG model are based on Ornthanalai (2014), adapted to our setting according to Section 6.4.1. As described in 6.4.2, the HN-GARCH and the homoscedastic model parameters are re-estimated based on a set of simulated log returns from the NIG model. The sample skewness for the first 15 years of the simulated log returns is -2.99, the kurtosis amounts to  $73.90.^7$ 

Concerning the evolution over time, Figure 6.7 shows that an investor with a time horizon of more than 1.5 years will invest 2.5 percentage points more than an investor with a very short time horizon. While the HN-GARCH solution only shows slight differences in the initial allocation for very small values of T, the homoscedastic solution does not change for different time horizons.

Plotting the values of the initial optimal portfolio allocation for a one-year time horizon for investors with different levels of relative risk aversion, Figure 6.8a shows that the small differences observed in Figure 6.7 for the special case of  $\mathcal{R} = 2$  are confirmed for other reasonable values, too. This is also reflected in Figure 6.8b, suggesting extremely low values for the WEL occurring when following the HN-GARCH or the homoscedastic solution, while returns actually evolve according to the NIG model.

The interpretation of the NIG innovation as a normal variance-mean mixture  $y_t \sim \mathcal{N}(\beta x, x)$ , with  $x \sim \text{IG}\left(h_{y,t}, \sqrt{\alpha^2 - \beta^2}\right)$ , helps to get an intuition about the role of the different parameters

<sup>&</sup>lt;sup>7</sup>These values are much more extreme than the numbers observed in the original dataset from the S&P 500 (see Section 6.4.3), similar to the picture for implied average conditional skewness and kurtosis observed by Ornthanalai (2014).

Table 6.3: MLEs for the NIG model and the recalibrated HN-GARCH and homoscedastic model, respectively. Standard errors for the recalibrated models are in parentheses. The NIG parameters are based on Ornthanalai (2014) and the methodology presented in Section 6.4.1, the two jump-free models have been estimated according to Section 6.4.2.

Par.	NIG	HN		HS	
$\lambda_z$	$8.00 \times 10^{-1}$	1.83	$(4.30\times10^{-1})$	1.83	$(3.97 \times 10^{-1})$
$\omega_z$	$-1.65 \times 10^{-6}$	$3.40 \times 10^{-6}$	$(8.81 \times 10^{-8})$	$3.21 \times 10^{-4}$	$(2.96 \times 10^{-5})$
$a_z$	$2.41\times 10^{-6}$	$4.21\times 10^{-7}$	$(4.62 \times 10^{-8})$		
$b_z$	$9.40 \times 10^{-1}$	$7.04 \times 10^{-1}$	$(2.25 \times 10^{-2})$		
$c_z$	$1.43\times 10^2$	$6.70  imes 10^2$	$(6.14\times 10^1)$		
$\lambda_y$	$6.88 \times 10^{-1}$				
Λ	2.62				
$\alpha$	$1.16 \times 10^1$				
β	-6.86				



Figure 6.7: Values of the optimal strategy  $\pi_0^*$  in the NIG model and the corresponding re-calibrated HN-GARCH and homoscedastic model, dependent on the length of the time horizon, for T up to five years. The parameters are given in Table 6.3.

in this setting. As for the MJ model in Section 6.4.3, we intend to increase the average jump size, which can be achieved via modifications to  $\beta$ . As we change  $\beta$ , we correct the inverse Gaussian distribution to remain the same by adapting  $\alpha$  accordingly. Increasing the absolute value of  $\beta$  up to one order of magnitude in this manner, and then re-calibrating the HN-GARCH and the homoscedastic model leads to the WEL plotted in Figure 6.9a. The loss increases towards higher absolute jump sizes for both suboptimal strategies, but the numbers remain on a very



(a) Values of  $\pi_0^*$  dependent on  $\mathcal{R}$ . (b) WEL dependent on  $\mathcal{R}$ .

Figure 6.8: Values of the optimal initial allocation  $\pi_0^*$  and of wealth-equivalent losses in the NIG model, dependent on the investor's level of relative risk aversion  $\mathcal{R} = 1 - \gamma \in [1.1, 5]$ . The time horizon is one year with 252 trading days, the parameters are given in Table 6.3.



(a) WEL dependent on β, up to an absolute increase
 (b) WEL dependent on α, up to an increase by a factor of 10.

Figure 6.9: Values of the wealth-equivalent losses, dependent on the jump parameters in the NIG model. For each grid point, both the HN-GARCH and the homoscedastic model are re-calibrated using the methodology in Section 6.4.2. The time horizon is one year with 252 trading days, and we consider an investor with  $\gamma = -1$ .

low level in general. It is remarkable that in this case, the homoscedastic model could produce a lower WEL than the heteroscedastic HN-GARCH for some very extreme settings.

We add that the third parameter that can influence the jumps,  $\Lambda$ , plays a different role in the NIG as compared to the MJ model. While we were looking at an increase of said parameter in the MJ to raise the jump intensity, we may now increase  $\Lambda$  to obtain a larger mean of the inverse Gaussian random variable in the normal variance-mean mixture. Looking at an increase of up to one order of magnitude and re-calibrating the HN-GARCH and the homoscedastic model at each of the grid points also shows insignificant values for the WEL. The WEL evaluated for an investor with  $\mathcal{R} = 2$  does not exceed one basis point.

### 6.4.5 Results for the Variance Gamma Model

This subsection is devoted to the numerical results for the special case of the Variance Gamma model, see Section 6.3.2. We follow the lines of Sections 6.4.3 and 6.4.4 and investigate the impact of more extreme jump settings in the VG model on the capability of the HN-GARCH



Figure 6.10: Values of wealth-equivalent losses for HN-GARCH and homoscedastic model, compared to the NIG model and dependent on the parameter  $\Lambda$ . For each grid point, both the HN-GARCH and the homoscedastic model are re-calibrated using the methodology in Section 6.4.2. The time horizon is one year with 252 trading days, and we consider an investor with  $\gamma = -1$ .

Table 6.4: MLEs for the VG model and the recalibrated HN-GARCH and homoscedastic model, respectively. Standard errors for the recalibrated models are in parentheses. The VG parameters are based on Ornthanalai (2014) and the methodology presented in Section 6.4.1, the two jump-free models have been estimated according to Section 6.4.2.

Par.	VG	HN		HS	
$\lambda_z$	$6.82\times10^{-1}$	2.33	$(3.34 \times 10^{-1})$	2.35	$(2.76 \times 10^{-1})$
$\omega_z$	$-1.75\times10^{-6}$	$1.25\times 10^{-7}$	$(3.18 \times 10^{-7})$	$4.49 \times 10^{-4}$	$(2.13 \times 10^{-5})$
$a_z$	$2.57\times 10^{-6}$	$2.85\times10^{-6}$	$(3.23 \times 10^{-7})$		
$b_z$	$9.38  imes 10^{-1}$	$9.44 \times 10^{-1}$	$(6.50 \times 10^{-3})$		
$c_z$	$1.47 \times 10^2$	$1.31  imes 10^2$	$(1.50\times 10^1)$		
$\lambda_y$	$1.68\times 10^{-2}$				
Λ	$1.17  imes 10^2$				
θ	-8.19				
β	$9.49  imes 10^2$				

and the homoscedastic model when it comes to dynamic investment strategies. Table 6.4 shows the estimates for the VG model, derived from Ornthanalai (2014) and adapted according to Section 6.4.1. The HN-GARCH and the homoscedastic parameter estimates are obtained according to the methodology described in Section 6.4.2. In this case, the sample skewness for the first 15 years of simulated log returns is -0.13, the sample kurtosis is 4.29.



Figure 6.11: Values of the optimal strategy  $\pi_0^*$  in the VG model and the corresponding re-calibrated HN-GARCH and homoscedastic model, dependent on the length of the time horizon, for T up to five years. The parameters are given in Table 6.4.

Figure 6.11 shows that the re-calibrated HN-GARCH model, based on a simulated dataset of VG returns, produces an initial value for the risky allocation which is quite close to the one obtained directly from the VG model for all time horizons up to five years. The initial allocation resulting from the homoscedastic model is constant over different time horizons and almost identical to the HN-GARCH and the VG strategy for very small T.



(a) Values of  $\pi_0^*$  dependent on  $\mathcal{R}$ .

(b) WEL dependent on  $\mathcal{R}$ .

Figure 6.12: Values of the optimal initial allocation  $\pi_0^*$  and of wealth-equivalent losses in the VG model, dependent on the investor's level of relative risk aversion  $\mathcal{R} = 1 - \gamma \in [1.1, 5]$ . The time horizon is one year with 252 trading days, the parameters are given in Table 6.4.

This is also reflected in the plots in Figure 6.12, showing the difference of the initial risky allocations for a one-year time horizon across different levels of relative risk aversion. The HN-GARCH yields a value almost identical to the one from the VG model, the homoscedastic model seems to be quite close, too. Concerning the WEL, the picture observed for the MJ and the NIG models in Sections 6.4.3 and 6.4.4 is confirmed once more, since the values for the losses when following the HN-GARCH are very low, exceeded by the losses produced with

the homoscedastic solution, which, however, also remain small – around one basis point for  $\mathcal{R}$  between 2 and 4.

Concerning the jump parameters in the VG model, we again look at the normal variance-mean mixture interpretation of the underlying distribution. In particular, considering  $y_t \sim \mathcal{N}(\theta x, x)$ , based on  $x \sim \text{Gamma}(h_{y,t},\beta)$ , we quickly identify  $\theta$  as the key to modifications in the jump size. Since the mean of x is given by  $h_{y,t}/\beta$ , and this again is reflected in both the mean and the variance of  $y_t$ , we can modify these two measures via decreasing  $\beta$  and leaving the behavior of the process  $h_{y,t}$  unchanged. Repeating the same procedure as for the other WEL evaluations



(a) WEL dependent on θ, up to an absolute increase
 (b) WEL dependent on β, up to a decrease by a factor of 10.

Figure 6.13: Values of the wealth-equivalent losses, dependent on the jump parameters in the VG model. For each grid point, both the HN-GARCH and the homoscedastic model are recalibrated using the methodology in Section 6.4.2. The time horizon is one year with 252 trading days, and we consider an investor with  $\gamma = -1$ .

here yields a very familiar picture, as shown in Figure 6.13: Re-calibrated using simulations from the jump model for different parameter values, the losses for the HN-GARCH and the homoscedastic model remain very low, barely exceeding one basis point for a one-year time horizon in the VG setting.

The evidence in this section indicates that Gaussian models excel at replicating the simulated data of jump processes across different jump models and parameter scenarios. In this context, we note that our Gaussian models are fitted based on a relatively large data set, see Section 6.4.2. Other results, e.g., in the context of regime switches (Ang and Bekaert, 2002; Guidolin and Timmermann, 2007; Campani et al., 2021, among others), are based on comparatively short time series. This suggests that the large amount of data points we use helps to eliminate the estimation error and significantly improves the performance of the Gaussian models.

## 6.5 Conclusion

This chapter establishes an approximate closed-form solution for a discrete-time portfolio optimization problem in a Lévy GARCH framework and therefore extends the results presented in Chapter 3 in a different direction than Chapters 4 and 5 do. We provide optimality criteria for different special cases of the framework, including both finite-activity and infinite-activity jump components. Our numerical study focuses on the impact of jumps on the optimal strategy. It compares three different jump models to the jump-free Gaussian HN-GARCH model and a homoscedastic Gaussian model. Across all considered jump models, the results demonstrate that investors following the closest possible jump-free variants obtain similar strategies and incur only insignificant wealth-equivalent losses. Our finding remains true for modified parameter settings indicating extreme jump behavior in the Lévy GARCH models. This surprising performance of simple models might be explained by two factors. One is the connection between expected utility maximization and mean-variance (MV) theory, stating that expected utility maximization can be well approximated using MV, i.e., by the first two moments (Markowitz, 2012). This highlights the importance of matching the first two moments of the wealth distribution. The quality of this approximation in other settings, e.g., with a disappointment-averse investor (Dahlquist et al., 2017) could be subject to further research. Secondly, the frequency of re-balancing in a discrete-time setting leaves little room to take advantage of sudden jumps. Overall, our analysis shows that correctly calibrated jump-free models in the discrete-time GARCH framework could diminish the necessity of modeling jumps for investors.

# 7 Conclusion

After the first closed-form solution to the EUT problem in a general affine GARCH environment presented in Chapter 3, we tackled the extension towards consumption at intermediate time points in Chapter 4. Our construction with a HARA type utility function at all time points  $t = 0, \ldots, T - 1$  and a change of control for the investor's consumption avoid the consumption of the entire available wealth before the end of the investment horizon. We furthermore proved that the optimal log-wealth process again follows an affine GARCH process and that the simpler case without consumption at intermediate time points is still nested in this setup. In Chapter 5, we established two generalizations concerning the utility derived from terminal wealth. The first allows for the more general class of HARA utilities instead of the embedded CRRA family. For a risk-averse investor, we learned that the HARA function of terminal wealth can be interpreted as a CRRA utility based on a surplus beyond some fixed threshold that is achieved with certainty, which provides us with an intuitive interpretation as a CPPI approach. The second generalization relaxes an assumption on the investor's risk aversion, allowing in particular also for a quadratic utility and a connection to the prominent field of MV optimization. In this generalized framework, one may even include higher moments of the portfolio return and consider investors' preferences on skewness and excess kurtosis. While Chapters 3, 4 and 5 extended the problem formulation in several directions, the theory was based on the log asset price evolving according to a general affine GARCH model. Chapter 6 finally broke through these barriers and considered the Lévy GARCH setting, where an additional jump component besides a Gaussian one drives asset price evolution. Potential choices for the distribution of the jump innovation include the Merton jump (MJ) with finite activity and more involved constructions such as Normal-inverse Gaussian (NIG) and Variance Gamma (VG) processes. By disregarding the jump innovation, the setup still nests the Gaussian HN-GARCH model.

We have seen various numeric examples throughout this dissertation. In the model with consumption in Chapter 4, we studied the impact of including consumption at intermediate time points on optimal portfolio strategies and noted that this additional possibility of deriving utility disrupts optimal allocation levels. Since the model setup allows for Gaussian and non-Gaussian innovations, we also investigated the effect of non-Gaussianity, i.e., of conditional skewness and excess kurtosis in asset returns, on the optimal strategy. This led to the eve-opening conclusion that, among conditional non-Gaussianity and heteroscedasticity, the latter is the more important feature in asset return models. In a study based on S&P 500 index returns, we estimated the Gaussian HN-GARCH model and a homoscedastic variant. The five-year period from 2013 through 2017 revealed that following a homoscedastic strategy can lead to 10% less total consumption and, at the same time, 8% less terminal wealth as compared to the Gaussian GARCH model. Focusing on utility derived from terminal wealth again in Chapter 5, we chose the newly established connection to MV theory and studied the efficient frontiers for a Gaussian GARCH and a homoscedastic model. The results indicated that, given the same expected portfolio return, a GARCH investor faces significantly less variance, with the difference to a homoscedastic investor increasing with the length of the investment horizon. This confirmed the general numerical finding of Chapter 4 and can be summarized in the brief statement that heteroscedasticity matters. The numerical study in Chapter 6 took a different point of view and considered the question of whether jumps in the asset price process make an impact. Instead of calibrating all involved models based on the same dataset, which in real-world applications is

usually limited to small time series, we wanted to know if the best possible jump-free Gaussian GARCH model is capable of imitating a Lévy GARCH model with jumps. To this end, we sampled a large number of returns with jumps and calibrated the HN-GARCH model as well as a homoscedastic variant based on this artificial dataset. Although the results are hypothetical because of the limited availability of market data in real applications, the results indicate that correctly calibrated GARCH models produce only negligible losses in various extreme jump environments. This finding is extremely relevant for practitioners and connects very well to the numerical studies from Chapters 4 and 5, as the most significant feature to consider in the underlying asset return model among heteroscedasticity, non-Gaussianity, and jumps, seems to be the first one.

The results presented in this dissertation leave room for open questions. The most natural ones target the unfilled space in Figure 6.1, i.e., the connection between HARA utilities for terminal wealth, consumption at intermediate time points, and more general asset price models. Since all the extensions presented in this dissertation preserve the exponentially affine structure of the MGF, a bold conjecture could be that the combination works in a similar fashion. In the projects related to this dissertation, we chose to deal with these issues separately for brevity and lucidity. In addition to these opportunities, a new class of affine multi-factor models was introduced very recently by Augustyniak et al. (2023), nesting all models of the present dissertation. In particular, while the IG-GARCH model is not included in the Lévy GARCH setup, both are contained in the new family, paving the way to further generalizations of our existing results. Another natural extension of our environment concerns the presence of more than one risky asset, i.e., the transition to a multivariate setting, which in the world of affine GARCH models was introduced by Escobar-Anel et al. (2020). Note, however, that the general multi-factor model of Augustyniak et al. (2023) is formulated for the univariate case. Last but not least, it would be interesting to study many of the numerical results in this dissertation with respect to their dependence on the length of the time step. For simplicity, we assume this length to be exactly one day throughout this dissertation, and so it remains an open question whether more intermediate time points change the effect that heteroscedasticity, non-Gaussianity, and jumps have on optimal portfolio strategies and strategy performance. For clarity and in the spirit of addressing challenges methodically, we chose not to delve into the abovementioned extensions in this dissertation.

# **Bibliography**

- Ang, A. and G. Bekaert (2002). "Regime Switches in Interest Rates." In: Journal of Business & Economic Statistics 20.2, pp. 163–182. DOI: 10.1198/073500102317351930.
- Ascheberg, M., N. Branger, H. Kraft, and F. T. Seifried (2016). "When do jumps matter for portfolio optimization?" In: *Quantitative Finance* 16.8, pp. 1297–1311. DOI: 10.1080/ 14697688.2015.1131844.
- Augustyniak, M. and A. Badescu (2021). "On the computation of hedging strategies in affine GARCH models." In: Journal of Futures Markets 41.5, pp. 710–735. DOI: 10.1002/fut. 22187.
- Augustyniak, M., A. Badescu, and J.-F. Bégin (2023). "A discrete-time hedging framework with multiple factors and fat tails: On what matters." In: *Journal of Econometrics* 232.2, pp. 416–444. DOI: 10.1016/j.jeconom.2021.08.002.
- Babaoğlu, K., P. Christoffersen, S. Heston, and K. Jacobs (2018). "Option Valuation with Volatility Components, Fat Tails, and Nonmonotonic Pricing Kernels." In: *The Review of* Asset Pricing Studies 8.2, pp. 183–231. DOI: 10.1093/rapstu/rax021.
- Badescu, A., Z. Cui, and J.-P. Ortega (2019). "Closed-form variance swap prices under general affine GARCH models and their continuous-time limits." In: Annals of Operations Research 282.1-2, pp. 27–57. DOI: 10.1007/s10479-018-2941-9.
- Barndorff-Nielsen, O. E. (1997). "Processes of normal inverse Gaussian type." In: *Finance and Stochastics* 2.1, pp. 41–68. DOI: 10.1007/s007800050032.
- Basak, S. and G. Chabakauri (2010). "Dynamic Mean-Variance Asset Allocation." In: *Review of Financial Studies* 23.8, pp. 2970–3016. DOI: 10.1093/rfs/hhq028.
- Battauz, A., M. D. Donno, and A. Sbuelz (2015). "Kim and Omberg Revisited: The Duality Approach." In: Journal of Probability and Statistics 2015, pp. 1–6. DOI: 10.1155/2015/ 581854.
- Bekaert, G., C. B. Erb, C. R. Harvey, and T. E. Viskanta (1998). "Distributional Characteristics of Emerging Market Returns and Asset Allocation." In: *The Journal of Portfolio Management* 24.2, pp. 102–116. DOI: 10.3905/jpm.24.2.102.
- Bickel, P. J. and K. A. Doksum (2006). *Mathematical Statistics*. Second edition. Prentice Hall. ISBN: 0-13-230637-9.
- Bingham, N. H. and R. Kiesel (2004). Risk-Neutral Valuation. Springer London. DOI: 10.1007/978-1-4471-3856-3.
- Bollerslev, T. (1986). "Generalized autoregressive conditional heteroskedasticity." In: Journal of Econometrics 31.3, pp. 307–327. DOI: 10.1016/0304-4076(86)90063-1.
- Campani, C. H., R. Garcia, and M. Lewin (2021). "Optimal portfolio strategies in the presence of regimes in asset returns." In: *Journal of Banking & Finance* 123, p. 106030. DOI: 10. 1016/j.jbankfin.2020.106030.
- Campbell, J. Y. and L. M. Viceira (1999). "Consumption and Portfolio Decisions when Expected Returns are Time Varying." In: *The Quarterly Journal of Economics* 114.2, pp. 433–495. DOI: 10.1162/003355399556043.

- Cao, H., A. Badescu, Z. Cui, and S. K. Jayaraman (2020). "Valuation of VIX and target volatility options with affine GARCH models." In: *Journal of Futures Markets* 40.12, pp. 1880–1917. DOI: 10.1002/fut.22157.
- Chacko, G. and L. M. Viceira (2005). "Dynamic Consumption and Portfolio Choice with Stochastic Volatility in Incomplete Markets." In: *Review of Financial Studies* 18.4, pp. 1369–1402. DOI: 10.1093/rfs/hhi035.
- Chen, Z. (2005). "Multiperiod consumption and portfolio decisions under the multivariate GARCH model with transaction costs and CVaR-based risk control." In: *OR Spectrum* 27.4, pp. 603–632. DOI: 10.1007/s00291-005-0195-9.
- Christoffersen, P., S. Heston, and K. Jacobs (2006). "Option valuation with conditional skewness." In: *Journal of Econometrics* 131.1-2, pp. 253–284. DOI: 10.1016/j.jeconom.2005. 01.010.
- Christoffersen, P., K. Jacobs, and C. Ornthanalai (2012). "Dynamic jump intensities and risk premiums: Evidence from S&P500 returns and options." In: *Journal of Financial Economics* 106.3, pp. 447–472. DOI: 10.1016/j.jfineco.2012.05.017.
- Dahlquist, M., A. Farago, and R. Tédongap (2017). "Asymmetries and Portfolio Choice." In: *Review of Financial Studies* 30.2, pp. 667–702. DOI: 10.1093/rfs/hhw091.
- DeGroot, M. H. and M. J. Schervish (2012). Probability and Statistics. 4th ed. Pearson Education. ISBN: 978-0-321-50046-5.
- Duan, J.-C., P. Ritchken, and Z. Sun (2006a). "Approximating GARCH-jump models, jumpdiffusion processes, and option pricing." In: *Mathematical Finance* 16.1, pp. 21–52. DOI: 10.1111/j.1467-9965.2006.00259.x.
- Duan, J.-C., P. Ritchken, and Z. Sun (2006b). Jump Starting GARCH: Pricing and Hedging Options with Jumps in Returns and Volatilities. Working paper. DOI: 10.26509/frbc-wp-200619.
- Duffie, D. and L. G. Epstein (1992a). "Asset Pricing with Stochastic Differential Utility." In: *Review of Financial Studies* 5.3, pp. 411–436. DOI: 10.1093/rfs/5.3.411.
- (1992b). "Stochastic Differential Utility." In: *Econometrica* 60.2, p. 353. DOI: 10.2307/2951600.
- Engle, R. F. (1982). "Autoregressive Conditional Heteroscedasticity with Estimates of the Variance of United Kingdom Inflation." In: *Econometrica* 50.4, p. 987. DOI: 10.2307/1912773.
- Escobar-Anel, M., M. Gollart, and R. Zagst (2022a). "Closed-form portfolio optimization under GARCH models." In: *Operations Research Perspectives* 9, p. 100216. DOI: 10.1016/j.orp. 2021.100216.
- Escobar-Anel, M., J. Rastegari, and L. Stentoft (2020). "Affine multivariate GARCH models." In: Journal of Banking & Finance 118.105895. DOI: 10.1016/j.jbankfin.2020.105895.
- Escobar-Anel, M., B. Spies, and R. Zagst (2021). "Expected Utility Theory on General Affine GARCH Models." In: Applied Mathematical Finance 28.6, pp. 477–507. DOI: 10.1080/ 1350486x.2022.2101010.
- (2023). "Do Jumps Matter in Discrete-Time Portfolio Optimization?" Working paper submitted for publication.
- (2024a). "Mean-variance optimization under affine GARCH: A utility-based solution." In: Finance Research Letters 59.104749. DOI: 10.1016/j.frl.2023.104749.
- (2024b). "Optimal consumption and investment in general affine GARCH models." In: OR Spectrum. DOI: 10.1007/s00291-024-00749-z.

- Escobar-Anel, M., L. Stentoft, and X. Ye (2022b). "The benefits of returns and options in the estimation of GARCH models. A Heston-Nandi GARCH insight." In: *Econometrics and Statistics*. DOI: 10.1016/j.ecosta.2022.12.001.
- Guidolin, M. and A. Timmermann (2007). "Asset allocation under multivariate regime switching." In: Journal of Economic Dynamics and Control 31.11, pp. 3503–3544. DOI: 10.1016/ j.jedc.2006.12.004.
- (2008). "International asset allocation under regime switching, skew, and kurtosis preferences." In: *Review of Financial Studies* 21.2, pp. 889–935. DOI: 10.1093/rfs/hhn006.
- Harvey, C. R., J. C. Liechty, M. W. Liechty, and P. Müller (2010). "Portfolio selection with higher moments." In: *Quantitative Finance* 10.5, pp. 469–485. DOI: 10.1080/ 14697681003756877.
- Heston, S. L. (1993). "A Closed-Form Solution for Options with Stochastic Volatility with Applications to Bond and Currency Options." In: *Review of Financial Studies* 6.2, pp. 327– 343. DOI: 10.1093/rfs/6.2.327.
- Heston, S. L. and S. Nandi (2000). "A Closed-Form GARCH Option Valuation Model." In: *Review of Financial Studies* 13.3, pp. 585–625. DOI: 10.1093/rfs/13.3.585.
- Hinderer, K., U. Rieder, and M. Stieglitz (2016). Dynamic Optimization. Springer International Publishing. DOI: 10.1007/978-3-319-48814-1.
- Karpfinger, C. and K. Meyberg (2013). *Algebra*. Springer Berlin Heidelberg. DOI: 10.1007/978-3-8274-3012-0.
- Kim, T. S. and E. Omberg (1996). "Dynamic Nonmyopic Portfolio Behavior." In: Review of Financial Studies 9.1, pp. 141–161. DOI: 10.1093/rfs/9.1.141.
- Kraft, H. (2005). "Optimal portfolios and Heston's stochastic volatility model: an explicit solution for power utility." In: *Quantitative Finance* 5.3, pp. 303–313. DOI: 10.1080/14697680500149503.
- Lai, T.-Y. (1991). "Portfolio selection with skewness: A multiple-objective approach." In: *Review of Quantitative Finance and Accounting* 1.3, pp. 293–305. DOI: 10.1007/bf02408382.
- Liu, J. (2007). "Portfolio Selection in Stochastic Environments." In: *Review of Financial Studies* 20.1, pp. 1–39. DOI: 10.1093/rfs/hhl001.
- Liu, J., F. A. Longstaff, and J. Pan (2003). "Dynamic Asset Allocation with Event Risk." In: The Journal of Finance 58.1, pp. 231–259. DOI: 10.1111/1540-6261.00523.
- Liu, J. and J. Pan (2003). "Dynamic derivative strategies." In: *Journal of Financial Economics* 69.3, pp. 401–430. DOI: 10.1016/s0304-405x(03)00118-1.
- Madan, D. B. and E. Seneta (1990). "The Variance Gamma (V.G.) Model for Share Market Returns." In: *The Journal of Business* 63.4, p. 511. DOI: 10.1086/296519.
- Maheu, J. M. and T. H. McCurdy (2004). "News Arrival, Jump Dynamics, and Volatility Components for Individual Stock Returns." In: *The Journal of Finance* 59.2, pp. 755–793. DOI: 10.1111/j.1540-6261.2004.00648.x.
- Markowitz, H. M. (1952). "Portfolio Selection." In: *The Journal of Finance* 7.1, p. 77. DOI: 10.2307/2975974.
- (1959). Portfolio Selection: Efficient Diversification of Investments. Yale University Press. ISBN: 9780300013726.
- (2012). "Mean-variance approximations to expected utility." In: European Journal of Operational Research 234.2, pp. 346-355. DOI: 10.1016/j.ejor.2012.08.023.
- Merton, R. C. (1969). "Lifetime Portfolio Selection under Uncertainty: The Continuous-Time Case." In: *The Review of Economics and Statistics* 51.3, p. 247. DOI: 10.2307/1926560.

- Merton, R. C. (1971). "Optimum consumption and portfolio rules in a continuous-time model." In: Journal of Economic Theory 3.4, pp. 373–413. DOI: 10.1016/0022-0531(71)90038-x.
- (1973). "An Intertemporal Capital Asset Pricing Model." In: *Econometrica* 41.5, p. 867.
   DOI: 10.2307/1913811.
- Mossin, J. (1968). "Optimal Multiperiod Portfolio Policies." In: *The Journal of Business* 41.2, p. 215. DOI: 10.1086/295078.
- Ornthanalai, C. (2014). "Lévy jump risk: Evidence from options and returns." In: *Journal of Financial Economics* 112.1, pp. 69–90. DOI: 10.1016/j.jfineco.2013.11.009.
- Roberts, A. W. and D. E. Varberg (1973). *Convex functions*. Academic Press, p. 300. ISBN: 0125897405.
- Samuelson, P. A. (1969). "Lifetime Portfolio Selection By Dynamic Stochastic Programming." In: The Review of Economics and Statistics 51.3, p. 239. DOI: 10.2307/1926559.
- Seshadri, V. (1999). The Inverse Gaussian Distribution. Springer New York. DOI: 10.1007/978-1-4612-1456-4.
- Zhou, X. and D. Li (2000). "Continuous-Time Mean-Variance Portfolio Selection: A Stochastic LQ Framework." In: Applied Mathematics and Optimization 42.1, pp. 19–33. DOI: 10.1007/ s002450010003.
- Zhu, Y. and M. Escobar-Anel (2022). "Polynomial affine approach to HARA utility maximization with applications to OrnsteinUhlenbeck 4/2 models." In: *Applied Mathematics and Computation* 418, p. 126836. DOI: 10.1016/j.amc.2021.126836.

# A Appendix for Chapter 2

## A.1 Proof of Theorem 2.15

The following proof is based on Hinderer et al. (2016, Theorem 16.1.12), relying on further results that can be found in earlier chapters of the same book.

Proof of Theorem 2.15. We verify the result by induction on  $n \ge 1$ , using the assertion  $(I_n)$  that the three following properties hold:

- $\phi_{T-n+1} \in \mathbb{M}$ ,
- $\pi^*_{T-n}$  is a maximizer at time T-n,
- $\phi_{T-n} = \mathbb{U}\phi_{T-n+1}$ .

Note that  $(I_1)$  holds since  $\phi_T \in \mathbb{M}$  by assumption and  $\pi_{T-1}^*$  is a maximizer by Condition (i). Furthermore, by (2.4b) and Definition 2.14, we deduce  $\phi_{T-1} = \mathbb{U}\phi_T$ . Assuming that for some  $n \geq 1$ , the hypothesis  $(I_{\nu})$  is true for all  $\nu = 1, \ldots, n$ , Condition (ii) immediately yields  $\phi_{T-n} \in \mathbb{M}$ . We know by Condition (i) that  $\pi_{T-n-1}^*$  is a maximizer at time T - n - 1, since  $\phi_{T-n} \in \mathbb{M}$ . The set of true hypotheses  $(I_n), \ldots, (I_1)$  shows that  $\{\pi_t^*\}_{t=T-n}^{T-1}$  is an optimal policy for the problem SCM<sub>n</sub>. Thus, there remains to show that  $\phi_{T-n-1} = \mathbb{U}\phi_{T-n}$ . Note that for  $s \in \mathbb{S}$ , denoting  $\pi' \coloneqq \{\pi_t\}_{T-n}^{T-1}$ ,

$$\phi_{T-n-1}(s) = \sup_{(\pi_{T-n-1},\pi')} \phi_{T-n-1,(\pi_{T-n-1},\pi')}(s)$$
  
= 
$$\sup_{\pi_{T-n-1}} \sup_{\pi'} \left\{ r_w(s,\pi_{T-n-1}(s)) + \beta \cdot \mathbb{E} \left[ \phi_{T-n,\pi'} \left( \mathbb{T} \left( s,\pi_{T-n-1}(s),Y \right) \right) \right] \right\}$$
  
= 
$$\sup_{\pi_{T-n-1}} \left\{ r_w(s,\pi_{T-n-1}(s)) + \beta \sup_{\pi'} \mathbb{E} \left[ \phi_{T-n,\pi'} \left( \mathbb{T} \left( s,\pi_{T-n-1}(s),Y \right) \right) \right] \right\}.$$
 (A.1)

If we can change the order of taking the supremum and the expectation in the last line, then we are done. For all  $s \in S$ , by Definition 2.14, we have that  $\phi_{T-n,\pi'}(s) \leq \phi_{T-n}(s)$ . Since this inequality also holds after taking the conditional expectation and the supremum over  $\pi'$ , we arrive at

$$\sup_{\pi'} \mathbb{E}\left[\phi_{T-n,\pi'}\left(\mathbb{T}\left(s,\pi_{T-n-1}(s),Y\right)\right)\right] \le \mathbb{E}\left[\phi_{T-n}\left(\mathbb{T}\left(s,\pi_{T-n-1}(s),Y\right)\right)\right].$$
 (A.2)

On the other hand, since there exists an optimal policy  $\pi^* = \{\pi_t^*\}_{t=T-n}^{T-1}$ , we have

$$\sup_{\pi'} \mathbb{E}\left[\phi_{T-n,\pi'}\left(\mathbb{T}\left(s,\pi_{T-n-1}(s),Y\right)\right)\right] \ge \mathbb{E}\left[\phi_{T-n,\pi^*}\left(\mathbb{T}\left(s,\pi_{T-n-1}(s),Y\right)\right)\right]$$
$$= \mathbb{E}\left[\phi_{T-n}\left(\mathbb{T}\left(s,\pi_{T-n-1}(s),Y\right)\right)\right].$$
(A.3)

This shows that we can change the order of expectation and supremum in (A.1) and with the maximizer  $\pi^*_{T-n-1}$  at time T-n-1 we obtain

$$\phi_{T-n-1} = \mathbb{U}\phi_{T-n},\tag{A.4}$$

also implying that  $\{\pi_t^*\}_{t=T-n-1}^{T-1}$  is an optimal policy for  $SCM_{n+1}$ .

# **B** Appendix for Affine GARCH Models

# B.1 MGF for the HN-GARCH Model

This section contains the proof for the exponentially affine nature of the moment generating function (MGF) of the log-asset return in an HN-GARCH model as presented and used throughout the main body of this dissertation. In the setting of Heston and Nandi (2000), in general denoting the length of the time step by  $\Delta$ , we let  $\theta = (\lambda, \omega, \beta, \alpha, \rho)$  be a vector of parameters, and r be the continuously compounded riskless rate for the bank account  $P^{(0)}$ . Assume the following dynamics for the log-asset return and its conditional variance, respectively:

$$X_{t+\Delta} - X_t \coloneqq \log\left(\frac{P_{t+\Delta}^{(1)}}{P_t^{(1)}}\right) = r + \lambda h_{t+\Delta} + \sqrt{h_{t+\Delta}}\epsilon_{t+\Delta}, \tag{B.1a}$$

$$h_{t+\Delta} = \omega + \beta h_t + \alpha \left(\epsilon_t - \rho \sqrt{h_t}\right)^2,$$
 (B.1b)

with  $\{\epsilon_t\}_t$  a sequence of i.i.d. standard normally distributed random variables. Note that we consider only the single-lag version here, where  $P^{(1)}$  and h are allowed to be autoregressive and dependent only on the previous realization, and that in the main body, we work with the length of the time step set to one day, i.e.,  $\Delta = 1$ . Along the lines of Heston and Nandi (2000, Appendix A), we prove the following result:

**Proposition B.1** (Heston and Nandi, 2000). Assuming the dynamics described in (B.1), the generating function takes the form

$$\Psi_{X_T}\left(u;t,T\right) \coloneqq \mathbb{E}_t\left[\left(P_T^{(1)}\right)^u\right] = \left(P_t^{(1)}\right)^u \exp\left\{A(u;t,T) + B(u;t,T)\,h_{t+\Delta}\right\},\tag{B.2}$$

with the coefficients satisfying

$$A(u;t,T) = A(u;t+\Delta,T) + ur + \omega B\left(u;t+\Delta,T\right) - \frac{1}{2}\log\left(1 - 2\alpha B(u;t+\Delta;T)\right), \quad (B.3a)$$

$$B(u;t,T) = u(\lambda + \rho) - \frac{1}{2}\rho^2 + \beta B(u,t+\Delta,T) + \frac{\frac{1}{2}(u-\rho)^2}{1-2\alpha B(u;t+\Delta,T)},$$
(B.3b)

and the terminal conditions A(u;T,T) = B(u;T,T) = 0.

*Proof.* We prove Proposition B.1 via induction on  $n := \frac{T-t}{\Delta}$ . That is, we start at the end of the time horizon and move backwards step by step. The terminal conditions follow directly from evaluating Equation (B.2) at t = T, corresponding to n = 0. Now, assuming that the induction hypothesis is true up to some  $n - 1 \ge 0$  corresponding to  $t + \Delta$ , we start by using the tower property and apply the induction hypothesis in the second equation. This yields:

$$\Psi_{X_T}(u;t,T) = \mathbb{E}_t \left[ \Psi_{X_T}(u;t+\Delta,T) \right]$$
  
=  $\mathbb{E}_t \left[ \exp \left\{ u X_{t+\Delta} + A(u;t+\Delta,T) + B(u;t+\Delta,T) \cdot h_{t+2\Delta} \right\} \right]$  (B.4)

B Appendix for Affine GARCH Models

$$= \mathbb{E}_{t} \left[ \exp \left\{ uX_{t} + ur + u\lambda h_{t+\Delta} + u\sqrt{h_{t+\Delta}}\epsilon_{t+\Delta} + A(u; t+\Delta, T) + B(t+\Delta, T, u) \left( \omega + \beta h_{t+\Delta} + \alpha \left( \epsilon_{t+\Delta} - \rho \sqrt{h_{t+\Delta}} \right)^{2} \right) \right\} \right]$$
  
$$= \mathbb{E}_{t} \left[ \exp \left\{ u \left( X_{t} + r \right) + A(u; t+\Delta, T) + B(u; t+\Delta, T) \omega + B(u; t+\Delta, T) \alpha \left( \epsilon_{t+\Delta}^{2} - 2\epsilon_{t+\Delta}\rho \sqrt{h_{t+\Delta}} + \rho^{2} h_{t+\Delta} \right) + u\epsilon_{t+\Delta} \sqrt{h_{t+\Delta}} + (u\lambda + B(u; t+\Delta, T)\beta) h_{t+\Delta} \right\} \right].$$
(B.5)

Via completing the square and using that everything except for  $\epsilon_{t+\Delta}$  is known at time t, we obtain, continuing from (B.5):

$$\Psi_{X_T}(u;t,T) = \mathbb{E}_t \left[ \exp \left\{ u \left( X_t + r \right) + A(u;t+\Delta,T) + B(u;t+\Delta,T) \omega \right. \\ \left. + B(u;t+\Delta,T) \alpha \left( \epsilon_{t+\Delta} - \left( \rho - \frac{u}{2\alpha B(u;t+\Delta,T)} \right) \sqrt{h_{t+\Delta}} \right)^2 \right. \\ \left. + \left( u\lambda + B(u;t+\Delta,T) \beta + \rho u - \frac{u^2}{4\alpha B(u;t+\Delta,T)} \right) h_{t+\Delta} \right\} \right] \\ = \exp \left\{ u \left( X_t + r \right) + A(u;t+\Delta,T) + B(u;t+\Delta,T) \omega \right. \\ \left. + \left( u\lambda + B(u;t+\Delta,T) \beta + \rho u - \frac{u^2}{4\alpha B(u;t+\Delta,T)} \right) h_{t+\Delta} \right\} \\ \left. + \mathbb{E}_t \left[ \exp \left( a \left( \epsilon_{t+\Delta} + b \right)^2 \right) \right],$$
(B.6)

setting  $a = \alpha \cdot B(u; t + \Delta, T)$  and  $b = -\left(\rho - \frac{u}{2\alpha B(u; t + \Delta, T)}\right)\sqrt{h_{t+\Delta}}$  in (B.6). For a standard normally distributed variable z, an integration by substitution and completing the square shows that (Heston and Nandi, 2000, p. 619; Badescu et al., 2019, p. 33)

$$\mathbb{E}\left[\exp\left(a(z+b)^2\right)\right] = \exp\left(-\frac{1}{2}\log(1-2a) + \frac{ab^2}{1-2a}\right).$$
(B.7)

We use (B.7) in (B.6) in order to derive:

$$\begin{split} \Psi_{X_T}(u;t,T) &= \exp\left\{ u\left(X_t + r\right) + A(u;t+\Delta,T) + B(u;t+\Delta,T)\omega \right. \\ &\quad \left. - \frac{1}{2}\log\left(1 - 2\alpha B(u;t+\Delta,T)\right) + \left(u\lambda + B(u;t+\Delta,T)\beta + \rho u\right)h_{t+\Delta} \right. \\ &\quad \left. + \left[\frac{1}{1 - 2\alpha B(u;t+\Delta,T)}\left(\alpha B(u;t+\Delta,T)\rho^2 - \rho u\right. \right. \\ &\quad \left. + \frac{u^2}{4\alpha B(u;t+\Delta,T)}\right) - \frac{u^2}{4\alpha B(u;t+\Delta,T)}\right]h_{t+\Delta} \right\} \\ &= \exp\left\{ u\left(X_t + r\right) + A(u;t+\Delta,T) + B(u;t+\Delta,T)\omega \right. \\ &\quad \left. - \frac{1}{2}\log\left(1 - 2\alpha B(u;t+\Delta,T)\right) \right. \\ &\quad \left. + \left(u(\lambda + \rho) - \frac{1}{2}\rho^2 + B(u;t+\Delta,T)\beta + \frac{\frac{1}{2}(u-\rho)^2}{1 - 2\alpha B(u;t+\Delta,T)}\right)h_{t+\Delta} \right\}, \end{split}$$

which gives exactly the coefficients in (B.3) and thus proves the hypothesis for also for n and t, respectively. It is easy to check that the following representation for B is equivalent to (B.3b):

$$B(u;t,T) = u\lambda + B(u,t+\Delta,T)\left(\alpha\rho^2 + \beta\right) + \frac{\left(u - 2\alpha\rho B(u,t+\Delta,T)\right)^2}{2\left(1 - 2\alpha B(u;t+\Delta,T)\right)}.$$
 (B.8)

**Remark B.2.** Instead of the general form in (B.2) with the coefficients (B.3), we will use the conditional bivariate one-step MGF of the log return  $Y_{t+\Delta} = X_{t+\Delta} - X_t$  and the conditional variance  $h_{t+2\Delta}$ , which is defined as

$$\Psi_{(Y_{t+\Delta},h_{t+2\Delta})}\left(u,v\mid\mathcal{F}_{t}\right) \coloneqq \mathbb{E}_{t}\left[\exp\left\{u\cdot Y_{t+\Delta}+v\cdot h_{t+2\Delta}\right\}\right],\tag{B.9}$$

for  $t \leq T - 2\Delta$ . Analogously to the calculations in the proof of Proposition B.1, we can show that:

$$\Psi_{(Y_{t+\Delta},h_{t+2\Delta})}\left(u,v\mid\mathcal{F}_{t}\right) = \exp\left\{A(u,v;t,t+\Delta) + B(u,v;t,t+\Delta)\cdot h_{t+\Delta}\right\},\tag{B.10}$$

with the coefficients A and B given by

$$A(u, v; t, t + \Delta) = ur + \omega v - \frac{1}{2} \log (1 - 2\alpha v),$$
 (B.11a)

$$B(u, v; t, t + \Delta) = u \left(\lambda + \rho\right) - \frac{1}{2}\rho^2 + \beta v + \frac{\frac{1}{2}(u - \rho)^2}{1 - 2\alpha v}.$$
 (B.11b)

In this dissertation, we omit the last two time arguments of A and B in the one-step MGF where they are obsolete. Furthermore, note that the representations in (B.11) assume that  $1-2\alpha v > 0$ .

## B.2 MGF for the IG-GARCH Model

In the setting of Christoffersen et al. (2006), let the vector of parameters be  $\theta = (\nu, \eta, w, a, b, c)$ . We assume the following dynamics for the log asset price and its conditional variance, respectively:

$$\log\left(\frac{P_{t+\Delta}^{(1)}}{P_t^{(1)}}\right) = r\Delta + \nu h_{t+\Delta} + \eta y_{t+\Delta}, \tag{B.12a}$$

$$h_{t+\Delta} = w + bh_t + cy_t + a\frac{h_t^2}{y_t},$$
(B.12b)

where  $\{y_t\}_t$  is a sequence of random variables with inverse Gaussian distribution and single parameter  $\delta(t) = \delta_t = \frac{h_t}{\eta^2}$ . Christoffersen et al. (2006, Appendix A) proved the following result:

**Proposition B.3** (Christoffersen et al., 2006). Assuming the dynamics described in Equation B.12, the generating function takes the form

$$\Psi_{X_T}(u;t,T) = \mathbb{E}_t \left[ \left( P_T^{(1)} \right)^u \right] = \left( P_t^{(1)} \right)^u \exp \left\{ A(u;t,T) + B(u;t,T) \, h_{t+\Delta} \right\}, \tag{B.13}$$

#### **B** Appendix for Affine GARCH Models

with the coefficients satisfying

$$\begin{aligned} A(u;t,T) &= A\left(u;t+\Delta,T\right) + ur\Delta + wB\left(u;t+\Delta,T\right) \\ &- \frac{1}{2}\log\left(1 - 2\eta^4 aB\left(u;t+\Delta,T\right)\right), \end{aligned} \tag{B.14a} \\ B(u;t,T) &= bB\left(u;t+\Delta,T\right) + u\nu + \frac{1}{\eta^2} \\ &- \frac{1}{\eta^2}\sqrt{\left(1 - 2a\eta^4 B\left(u;t+\Delta,T\right)\right)\left(1 - 2u\eta - 2B\left(u;t+\Delta,T\right)c\right)}, \end{aligned} \tag{B.14b}$$

and the terminal conditions A(u;T,T) = B(u;T,T) = 0.

*Proof.* We again use induction on  $n := \frac{T-t}{\Delta}$ . The terminal conditions follow directly from evaluating  $\Psi_{X_T}(u; T, T)$ , proving the hypothesis for n = 0. Assuming that this hypothesis holds true up to some  $n - 1 \ge 0$ , corresponding to  $t + \Delta$ , we apply the tower property to find:

$$\begin{split} \Psi_{X_T}(u;t,T) &= \mathbb{E}_t \left[ \Psi_{X_T}(u;t+\Delta,T) \right] \\ &= \mathbb{E}_t \left[ \left( P_t^{(1)} \right)^u \exp\left\{ u \log\left( \frac{P_{t+\Delta}^{(1)}}{P_t^{(1)}} \right) + A\left(u;t+\Delta,T\right) + B\left(u;t+\Delta,T\right) h_{t+2\Delta} \right\} \right] \\ &= \left( P_t^{(1)} \right)^u \cdot \mathbb{E}_t \left[ \exp\left\{ u \left( r\Delta + \nu h_{t+\Delta} + \eta y_{t+\Delta} \right) + A\left(u;t+\Delta,T\right) \right. \\ &+ B\left(u;t+\Delta,T\right) \left( w + b h_{t+\Delta} + c y_{t+\Delta} + a \frac{h_{t+\Delta}^2}{y_{t+\Delta}} \right) \right\} \right] \\ &= \left( P_t^{(1)} \right)^u \cdot \mathbb{E}_t \left[ \exp\left\{ ur\Delta + A\left(u;t+\Delta,T\right) + B\left(u;t+\Delta,T\right)w \right. \\ &+ \left( u\nu + B\left(u;t+\Delta,T\right)b \right) \cdot h_{t+\Delta} \right. \\ &+ \left( u\eta + B\left(u;t+\Delta,T\right)c \right) \cdot y_{t+\Delta} \right. \\ &+ B\left(u;t+\Delta,T\right)a \cdot \frac{h_{t+\Delta}^2}{y_{t+\Delta}} \right\} \right]. \end{split}$$
(B.15)

Note that the only parameter that is not known by time t is  $y_{t+\Delta}$ , so the first to lines in (B.15) can just be taken out of the expectation. For what is left in the expectation we use the generalization of the MGF of the inverse Gaussian distribution, Formula (2) in Christoffersen et al. (2006). If y inversely Gaussian distributed with single parameter  $\delta$ , and  $\alpha, \beta \in \mathbb{R}$ , then (Christoffersen et al., 2006, Equation (2))

$$\mathbb{E}\left[\exp\left(\alpha y + \frac{\beta}{y}\right)\right] = \frac{\delta}{\sqrt{\delta^2 - 2\beta}} \exp\left\{\delta - \sqrt{\left(\delta^2 - 2\beta\right)\left(1 - 2\alpha\right)}\right\}.$$
 (B.16)

Continuing from (B.15), we set  $\alpha = u\eta + B(u; t + \Delta, T) c$  and  $\beta = B(u; t + \Delta, T) \cdot a \cdot h_{t+\Delta}^2$ , and  $\delta = h_{t+\Delta}/\eta^2$ . Writing  $A_{t+\Delta}$  and  $B_{t+\Delta}$  short for  $A(u; t + \Delta, T)$  and  $B(u; t + \Delta, T)$ , respectively, we obtain:

$$\Psi_{X_T}(u;t,T) = \exp\left\{ur\Delta + A_{t+\Delta} + B_{t+\Delta}w + (u\nu + B_{t+\Delta}b) \cdot h_{t+\Delta}\right\}$$

$$\times \left(1 - 2\eta^4 a B_{t+\Delta}\right)^{-\frac{1}{2}}$$

$$\times \exp\left\{\left(\frac{1}{\eta^2} - \sqrt{\left(\frac{1}{\eta^4} - 2aB_{t+\Delta}\right)\left(1 - 2u\eta - 2B_{t+\Delta}c\right)}\right)h_{t+\Delta}\right\}.$$
(B.17)

Comparing coefficients gives the recursions in (B.14), proving the hypothesis also for n.  $\Box$ 

**Remark B.4.** In the spirit of Remark B.2, we can derive the conditional bivariate one-step MGF of the log return and the conditional variance as

$$\Psi_{(Y_{t+\Delta},h_{t+2\Delta})}(u,v \mid \mathcal{F}_t) = \mathbb{E}_t \left[ \exp\left\{ u \cdot Y_{t+\Delta} + v \cdot h_{t+2\Delta} \right\} \right]$$
  
=  $\exp\left\{ A(u,v;t,t+\Delta) + B(u,v;t,t+\Delta) \cdot h_{t+\Delta} \right\},$  (B.18)

with the coefficients

$$A(u, v; t, t + \Delta) = ur + vw - \frac{1}{2} \log \left(1 - 2va\eta^4\right),$$
(B.19a)

$$B(u,v;t,t+\Delta) = vb + u\nu + \frac{1}{\eta^2} \left( 1 - \sqrt{(1 - 2va\eta^4)(1 - 2u\eta - 2vc)} \right).$$
(B.19b)

Note that the above calculations assume that  $1 - 2va\eta^4 > 0$  and  $1 - 2u\eta - 2vc > 0$ .

# C Appendix for Chapter 4

## C.1 Proofs from Section 4.3

Proof of Theorem 4.4. We start by guessing the correct form of the value function and using it as an ansatz. For lack of better terms, the factors in (4.11) are denoted by P, Q, and R. The recursive parameters P, R, D, and  $E^h$  depend on the optimal solution, but we omit the dependencies in the following lines for readability. Furthermore, referring to the coefficients Aand B of the moment generating function of the affine GARCH model in (4.5), let us introduce the following short notation:

$$\bar{A} = A\left(\left(E_{t+1,T}^{W} + E_{t+1,T}^{\hat{C}}Q_{t+1,T}\right)\pi_{t}, E_{t+1,T}^{h} + E_{t+1,T}^{\hat{C}}R_{t+1,T}\right),$$
(C.1a)

$$\bar{B} = B\left(\left(E_{t+1,T}^{W} + E_{t+1,T}^{\hat{C}}Q_{t+1,T}\right)\pi_{t}, E_{t+1,T}^{h} + E_{t+1,T}^{\hat{C}}R_{t+1,T}\right).$$
(C.1b)

Using the above-mentioned ansatz and the affine form of the optimal control on consumption, we arrive at:

$$\begin{split} & \phi_t \left( W_t, h_{t+1} \right) \\ &= \max_{\pi_t, \hat{C}_t} \left\{ -\bar{a}^{\bar{\gamma}} \exp\left\{ -\bar{\gamma} \hat{C}_t \right\} + \bar{\beta} \cdot \mathbb{E}_t \left[ \phi_{t+1} \left( \mathbb{T} \left( W_t, h_{t+1} \right) \right) \right] \right\} \\ &= \max_{\pi_t, \hat{C}_t} \left\{ -\bar{a}^{\bar{\gamma}} \exp\left\{ -\bar{\gamma} \hat{C}_t \right\} - \bar{\beta} \cdot \mathbb{E}_t \left[ \sum_{s=t+1}^{T-1} \bar{\beta}^{s-t-1} \bar{a}^{\bar{\gamma}} \exp\left\{ -\bar{\gamma} \hat{C}_s^* \right\} \right] \\ &\quad + \frac{\bar{\beta}}{\gamma} \cdot \mathbb{E}_t \left[ \bar{\beta}^{T-t-1} \exp\left\{ D_{t+1,T} + E_{t+1,T}^{\hat{C}} \cdot \hat{C}_{t+1}^* \right. \\ &\quad + E_{t+1,T}^W \cdot W_{t+1} + E_{t+1,T}^h \cdot h_{t+2} \right\} \right] \right\} \\ &= \max_{\pi_t, \hat{C}_t} \left\{ -\bar{a}^{\bar{\gamma}} \exp\left\{ -\bar{\gamma} \hat{C}_t \right\} - \mathbb{E}_t \left[ \sum_{s=t+1}^{T-1} \bar{\beta}^{s-t} \bar{a}^{\bar{\gamma}} \exp\left\{ -\bar{\gamma} \hat{C}_s^* \right\} \right] \\ &\quad + \frac{\bar{\beta}^{T-t}}{\gamma} \cdot \mathbb{E}_t \left[ \exp\left\{ D_{t+1,T} + E_{t+1,T}^{\hat{C}} \cdot P_{t+1,T} \right. \\ &\quad + \left( E_{t+1,T}^W + E_{t+1,T}^{\hat{C}} Q_{t+1,T} \right) \cdot W_{t+1} \right. \\ &\quad + \left( E_{t+1,T}^h + E_{t+1,T}^{\hat{C}} R_{t+1,T} \right) \cdot h_{t+2} \right\} \right] \right\}. \end{split}$$

Together with the approximation of the self-financing condition (4.3) and the bivariate generating function of the log return and the conditional variance in the underlying affine GARCH model (4.5), this leads to:

$$\phi_t\left(W_t, h_{t+1}\right)$$

C Appendix for Chapter 4

$$\begin{split} &= \max_{\pi_{t},\hat{C}_{t}} \left\{ -\bar{a}^{\tilde{\gamma}} \exp\left\{-\bar{\gamma}\hat{C}_{t}^{t}\right\} - \mathbb{E}_{t} \left[ \sum_{s=t+1}^{T-1} \tilde{\beta}^{s-t} \bar{a}^{\tilde{\gamma}} \exp\left\{-\bar{\gamma}\hat{C}_{s}^{t}\right\} \right] \\ &+ \frac{\bar{\beta}^{T-t}}{\gamma} \cdot \mathbb{E}_{t} \left[ \exp\left\{ D_{t+1,T} + E_{t+1,T}^{\hat{C}} \cdot P_{t+1,T} \right. \\ &+ \left( E_{t+1,T}^{W} + E_{t+1,T}^{\hat{C}} Q_{t+1,T} \right) \right. \\ &\times \left( W_{t} - \hat{C}_{t} + (1 - \pi_{t}) r + \pi_{t} \left( X_{t+1} - X_{t} \right) + \frac{1}{2} \left( \pi_{t} - \pi_{t}^{2} \right) h_{t+1} \right) \\ &+ \left( E_{t+1,T}^{h} + E_{t+1,T}^{\hat{C}} R_{t+1,T} \right) \cdot h_{t+2} \right\} \right] \right\} \\ &= \max_{\pi_{t},\hat{C}_{t}} \left\{ -\bar{a}^{\tilde{\gamma}} \exp\left\{ -\bar{\gamma}\hat{C}_{t}^{\hat{c}} \right\} - \mathbb{E}_{t} \left[ \sum_{s=t+1}^{T-1} \bar{\beta}^{s-t} \bar{a}^{\tilde{\gamma}} \exp\left\{ -\bar{\gamma}\hat{C}_{s}^{*} \right\} \right] \\ &+ \frac{\bar{\beta}^{T-t}}{\gamma} \exp\left\{ D_{t+1,T} + E_{t+1,T}^{\hat{C}} P_{t+1,T} \right. \\ &+ \left( E_{t+1,T}^{W} + E_{t+1,T}^{\hat{C}} Q_{t+1,T} \right) \left( 1 - \pi_{t} \right) r \\ &+ \left( E_{t+1,T}^{W} + E_{t+1,T}^{\hat{C}} Q_{t+1,T} \right) \left( W_{t} - \hat{C}_{t} \right) \\ &+ \frac{1}{2} \left( E_{t+1,T}^{W} + E_{t+1,T}^{\hat{C}} Q_{t+1,T} \right) \pi_{t} \left( X_{t+1} - X_{t} \right) \\ &+ \left( E_{t+1,T}^{k} + E_{t+1,T}^{\hat{C}} Q_{t+1,T} \right) \pi_{t} \left( X_{t+1} - X_{t} \right) \\ &+ \left( E_{t+1,T}^{k} + E_{t+1,T}^{\hat{C}} Q_{t+1,T} \right) h_{t+2} \right\} \right] \right\} \\ &= \max_{\pi_{t},\hat{C}_{t}} \left\{ -\bar{a}^{\tilde{\gamma}} \exp\left\{ -\bar{\gamma}\hat{C}_{t}^{\hat{c}} \right\} - \mathbb{E}_{t} \left[ \sum_{s=t+1}^{T-1} \bar{\beta}^{s-t} \bar{a}^{\tilde{\gamma}} \exp\left\{ -\bar{\gamma}\hat{C}_{s}^{*} \right\} \right] \\ &+ \frac{\bar{\beta}^{T-t}}{\gamma} \exp\left\{ D_{t+1,T} + E_{t+1,T}^{\hat{C}} P_{t+1,T} \right) h_{t+2} \right\} \right] \right\} \\ &= \max_{\pi_{t},\hat{C}_{t}} \left\{ -\bar{a}^{\tilde{\gamma}} \exp\left\{ -\bar{\gamma}\hat{C}_{t} \right\} - \mathbb{E}_{t} \left[ \sum_{s=t+1}^{T-1} \bar{\beta}^{s-t} \bar{a}^{\tilde{\gamma}} \exp\left\{ -\bar{\gamma}\hat{C}_{s}^{*} \right\} \right] \\ &+ \frac{\bar{\beta}^{T-t}}{\gamma} \exp\left\{ D_{t+1,T} + E_{t+1,T}^{\hat{C}} P_{t+1,T} \right\} h_{t+2} \right\} \right\} \right\}, \end{aligned}$$

using the coefficients A and B from the generating function of the affine GARCH model. This justifies (4.10) with the recursive definitions<sup>1</sup>

$$D_{t,T}(\pi_t) = D_{t+1,T}(\pi_{t+1}^*) + E_{t+1,T}^{\hat{C}} \cdot P_{t+1,T}(\pi_{t+1}^*) + \bar{A} + \left(E_{t+1,T}^W + E_{t+1,T}^{\hat{C}} Q_{t+1,T}\right) \cdot (1 - \pi_t) r,$$
(C.2a)

<sup>&</sup>lt;sup>1</sup>For completeness, dependencies on the optimal solution are displayed in (C.2), but omitted again afterwards.

$$E_{t,T}^{\hat{C}} = -\left(E_{t+1,T}^{W} + E_{t+1,T}^{\hat{C}}Q_{t+1,T}\right),\tag{C.2b}$$

$$E_{t,T}^{W} = \left( E_{t+1,T}^{W} + E_{t+1,T}^{\hat{C}} Q_{t+1,T} \right),$$
(C.2c)

$$E_{t,T}^{h}(\pi_{t}) = \frac{1}{2} \left( E_{t+1,T}^{W} + E_{t+1,T}^{\hat{C}} Q_{t+1,T} \right) \cdot \left( \pi_{t} - \pi_{t}^{2} \right) + \bar{B},$$
(C.2d)

for all t < T, where in the terminal case  $D_{T,T} = E_{T,T}^{\hat{C}} = E_{T,T}^{h} = 0$  and  $E_{T,T}^{W} = \gamma$ . Note that in particular, for all t < T - 1,

$$E_{t,T}^{W} = \left(E_{t+1,T}^{W} + E_{t+1,T}^{\hat{C}}Q_{t+1,T}\right) = E_{t+1,T}^{W}\left(1 - Q_{t+1,T}\right) = -E_{t+1,T}^{\hat{C}}\left(1 - Q_{t+1,T}\right).$$

Concerning the solution to the optimization problem, we apply the first-order conditions w.r.t.  $\pi_t$  to find the optimality equations from

$$\frac{\partial}{\partial \pi_t} D_{t,T}(\pi_t) = 0 \qquad \Leftrightarrow \qquad \frac{\partial}{\partial u} \bar{A} = r, \tag{C.3a}$$

and

$$\frac{\partial}{\partial \pi_t} E^h_{t,T}(\pi_t) = 0 \qquad \Leftrightarrow \qquad \frac{\partial}{\partial u} \bar{B} = \pi_t - \frac{1}{2}.$$
 (C.3b)

Taking derivatives w.r.t.  $\hat{C}_t$  yields the following first-order condition:

$$\bar{\gamma}\bar{a}^{\bar{\gamma}}\exp\left\{-\bar{\gamma}\hat{C}_{t}\right\} = -\frac{\bar{\beta}^{T-t}E_{t,T}^{\hat{C}}}{\gamma}\exp\left\{D_{t,T} + E_{t,T}^{\hat{C}}\hat{C}_{t} + E_{t,T}^{W}W_{t} + E_{t,T}^{h}h_{t+1}\right\},\tag{C.4}$$

which is equivalent to

$$\hat{C}_{t}^{*} = -\frac{1}{\bar{\gamma} + E_{t,T}^{\hat{C}}} \left( \log \left( -\frac{\bar{\beta}^{T-t} E_{t,T}^{\hat{C}}}{\gamma \bar{\gamma} \bar{a}^{\bar{\gamma}}} \right) + D_{t,T} + E_{t,T}^{W} W_{t} + E_{t,T}^{h} h_{t+1} \right).$$
(C.5)

Requiring that  $E_{t,T}^{\hat{C}} > 0$  for all t < T, this justifies (4.11) with<sup>2</sup>

$$P_{t,T}\left(\pi_{t}^{*}\right) = -\frac{1}{\bar{\gamma} + E_{t,T}^{\hat{C}}} \left( \log\left(-\frac{\bar{\beta}^{T-t}E_{t,T}^{\hat{C}}}{\gamma\bar{\gamma}\bar{a}^{\bar{\gamma}}}\right) + D_{t,T}\left(\pi_{t}^{*}\right) \right), \tag{C.6a}$$

$$Q_{t,T} = -\frac{E_{t,T}^W}{\bar{\gamma} + E_{t,T}^{\hat{C}}},\tag{C.6b}$$

$$R_{t,T}(\pi_t^*) = -\frac{E_{t,T}^h(\pi_t^*)}{\bar{\gamma} + E_{t,T}^{\hat{C}}}.$$
(C.6c)

Since  $C_T = 0$ , we start the recursion with  $P_{T,T} = Q_{T,T} = R_{T,T} = 0$ . Plugging (C.6b) into (C.2b) and using that  $\bar{\gamma} > 0$  yields for t < T - 1:

$$E_{t,T}^{\hat{C}} = E_{t+1,T}^{\hat{C}} - E_{t+1,T}^{\hat{C}} Q_{t+1,T} = E_{t+1,T}^{\hat{C}} + \frac{E_{t+1,T}^{\hat{C}} E_{t+1,T}^{W}}{\bar{\gamma} + E_{t+1,T}^{\hat{C}}}$$

 $<sup>^{2}</sup>$ Again, we display dependencies on the optimal solution in (C.6) for completeness.

C Appendix for Chapter 4

$$= E_{t+1,T}^{\hat{C}} - \frac{\left(E_{t+1,T}^{\hat{C}}\right)^2}{\bar{\gamma} + E_{t+1,T}^{\hat{C}}} > E_{t+1,T}^{\hat{C}} - E_{t+1,T}^{\hat{C}} = 0,$$

Now since  $E_{t,T}^{\hat{C}} > 0$  and  $E_{t,T}^{W} = -E_{t,T}^{\hat{C}}$  for all t < T, we also find that  $E_{t,T}^{W} < 0$  for all  $t \in \{0, \ldots, T\}$ . With all these ingredients, we can use Proposition 2.21 to show that the objective function is concave on  $\mathbb{A}_{\pi} \times \mathbb{A}_{\hat{C}}$ . To see this, note that in the notation and terminology of stochastic control models, our state space is  $\mathbb{W} \times \mathbb{H}$ , containing elements (W, h). The action space is  $\mathbb{A}_{\pi} \times \mathbb{A}_{\hat{C}} = \mathbb{R} \times [0, \infty)$ . We emphasize that the set of admissible actions stays the same for all states, implying in particular that the constraint set of states and actions is given by  $\mathbb{W} \times \mathbb{H} \times \mathbb{A}_{\pi} \times \mathbb{A}_{\hat{C}}$ . The space of disturbances is  $\mathbb{Y}$ , containing all possible log returns  $Y_t = X_t - X_{t-1}, t \in \{1, \ldots, T\}$ . The transition function is given by (4.6). Note that the first component of the transition function, denoted by  $\mathbb{T}_1$ , is increasing in W and concave in  $(W, \pi, \hat{C})$  on  $\mathbb{W} \times \mathbb{A}_{\pi} \times \mathbb{A}_{\hat{C}}$  for all  $h \in \mathbb{H}$  and  $Y \in \mathbb{Y}$ , since for  $\lambda \in (0, 1)$  and  $W_1, W_2 \in \mathbb{W}, \pi_1, \pi_2 \in \mathbb{A}_{\pi}$  and  $\hat{C}_1, \hat{C}_2 \in \mathbb{A}_{\hat{C}}$ , we find, using that  $x \mapsto -x^2$  is concave,

$$\begin{aligned} \mathbb{T}_{1} \left( \lambda W_{1} + (1-\lambda)W_{2}, h, \lambda\pi_{1} + (1-\lambda)\pi_{2}, \lambda \hat{C}_{1} + (1-\lambda)\hat{C}_{2}, Y \right) \\ &= \lambda W_{1} + (1-\lambda)W_{2} - \lambda \hat{C}_{1} - (1-\lambda)\hat{C}_{2} + (\lambda\pi_{1} + (1-\lambda)\pi_{2})Y \\ &+ (1-\lambda\pi_{1} - (1-\lambda)\pi_{2}))r \\ &+ \left( \lambda\pi_{1} + (1-\lambda)W_{2} - (\lambda\pi_{1} + (1-\lambda)\pi_{2})^{2} \right)h \\ &\geq \lambda W_{1} + (1-\lambda)W_{2} - \lambda \hat{C}_{1} - (1-\lambda)\hat{C}_{2} + (\lambda\pi_{1} + (1-\lambda)\pi_{2})Y \\ &+ (1-\lambda\pi_{1} - (1-\lambda)\pi_{2}))r \\ &+ (\lambda\pi_{1} + (1-\lambda)\pi_{2} - \lambda\pi_{1}^{2} - (1-\lambda)\pi_{2}^{2})h \\ &= \lambda \cdot \left[ W_{1} - \hat{C}_{1} + \pi_{1}Y + (1-\pi_{1})r + (\pi_{1} - \pi_{1}^{2})h \right] \\ &+ (1-\lambda) \cdot \left[ W_{2} - \hat{C}_{2} + \pi_{2}Y + (1-\pi_{2})r + (\pi_{2} - \pi_{2}^{2})h \right] \\ &= \lambda \mathbb{T}_{1} \left( W_{1}, h, \pi_{1}, \hat{C}_{1}, Y \right) + (1-\lambda) \mathbb{T}_{1} \left( W_{2}, h, \pi_{2}, \hat{C}_{2}, Y \right). \end{aligned}$$

The second component  $\mathbb{T}_2$  of the transition function is independent of  $(W, \pi, \hat{C})$ . The "one-step" return of the present stochastic control model is given via the utility derived from immediate consumption in the current time step. For  $t \in \{0, \ldots, T-1\}$ , we defined this utility to be

$$U_t\left(\hat{C}_t\right) = -\bar{a}^{\bar{\gamma}}\exp\left\{-\bar{\gamma}\hat{C}_t\right\},\,$$

which is an increasing and concave function in  $\hat{C}_t$  on  $\mathbb{A}_{\hat{C}}$  if  $\bar{\gamma} > 0$ . Finally, note that for  $\gamma < 0$ , the value function is bounded from above by zero, and the terminal value function, defined as

$$\phi_T\left(W_T\right) = rac{1}{\gamma} \exp\left\{\gamma W_T
ight\}$$

is increasing and concave in  $W_T$ . All in all, this leads to the requirements (i)-(v) in Proposition 2.21 being satisfied, implying in particular that the objective function at time t, denoted
by

$$f_t(W_t, h_{t+1}, \pi_t, \hat{C}_t) = -\bar{a}^{\bar{\gamma}} \exp\left\{-\bar{\gamma}\hat{C}_t\right\} - \mathbb{E}_t \left[\sum_{s=t+1}^{T-1} \bar{\beta}^{s-t} \bar{a}^{\bar{\gamma}} \exp\left\{-\bar{\gamma}\hat{C}_s^*\right\}\right] \\ + \frac{\bar{\beta}^{T-t}}{\gamma} \exp\left\{D_{t,T} + E_{t,T}^{\hat{C}}\hat{C}_t + E_{t,T}^W W_t + E_{t,T}^h h_{t+1}\right\}$$

for  $t \in \{0, ..., T-1\}$ , is concave on  $\mathbb{W} \times \mathbb{A}_{\pi} \times \mathbb{A}_{\hat{C}}$  for all  $t \in \{0, ..., T-1\}$ , and thus that a solution  $\{\pi_t^*, \hat{C}_t^*\}$  to (C.3) and (C.5) is actually a maximum.

Proof of Corollary 4.6. From (4.3), we deduce that

$$Var_{t-1}\left[W_{t}\right] = \left(\pi_{t-1}^{*}\right)^{2} \cdot h_{t}$$

Concerning the required conditional bivariate moment generating function of  $W_t$  and its conditional variance, we note that the optimal strategy  $\{\pi_t^*\}_t$  is non-random. For the sake of readability, we drop the asterisk for the optimal risky allocation and calculate:

$$\begin{split} \Psi_{(W_{t},\pi_{t}^{2}h_{t+1})}\left(u,v\mid\mathcal{F}_{t-1}\right) \\ &= \mathbb{E}_{t-1}\left[\exp\left\{u\cdot W_{t}+v\cdot\pi_{t}^{2}h_{t+1}\right\}\right] \\ &= \mathbb{E}_{t-1}\left[\exp\left\{u\cdot\left(W_{t-1}-\hat{C}_{t-1}+\pi_{t-1}Y_{t}+(1-\pi_{t-1})\,r+\frac{1}{2}\left(\pi_{t-1}-\pi_{t-1}^{2}\right)h_{t}\right)+v\cdot\pi_{t}^{2}h_{t+1}\right\}\right] \\ &= \exp\left\{u\left(W_{t-1}-\hat{C}_{t-1}\right)+(1-\pi_{t-1})\,ur+\frac{u}{2}\left(\pi_{t-1}-\pi_{t-1}^{2}\right)h_{t}\right\} \\ &\times \mathbb{E}_{t-1}\left[\exp\left\{u\pi_{t-1}Y_{t}+v\pi_{t}^{2}h_{t+1}\right\}\right] \\ &= \exp\left\{A\left(u\pi_{t-1},v\pi_{t}^{2}\right)+(1-\pi_{t-1})\,ur+u\left(W_{t-1}-\hat{C}_{t-1}\right)\right. \\ &\left.+\frac{1}{\pi_{t-1}^{2}}\left[B\left(u\pi_{t-1},v\pi_{t}^{2}\right)+\frac{u}{2}\left(\pi_{t-1}-\pi_{t-1}^{2}\right)\right]\cdot\pi_{t-1}^{2}h_{t}\right\} \\ &= \exp\left\{A\left(u\pi_{t-1},v\pi_{t}^{2}\right)-uP_{t-1}+(1-\pi_{t-1})\,ur+u\left(1-Q_{t-1}\right)W_{t-1}\right. \\ &\left.+\frac{1}{\pi_{t-1}^{2}}\left[B\left(u\pi_{t-1},v\pi_{t}^{2}\right)-uR_{t-1}+\frac{u}{2}\left(\pi_{t-1}-\pi_{t-1}^{2}\right)\right]\cdot\pi_{t-1}^{2}h_{t}\right\}. \end{split}$$

This shows that the conditional bivariate moment generating function is affine in the conditional variance, implying that  $\{W_t\}_t$  follows an affine GARCH process (cf. Badescu et al., 2019).

Checking the optimality equations for the optimal risky allocation, we note that there is no dependence on the conditional variance or on the level of wealth.  $\Box$ 

Proof of Corollary 4.8. We start by deriving the closed-form expression

$$\phi_t \left( W_t, h_{t+1} \right) = -\sum_{\tau=t}^T \exp\left\{ \tilde{D}_{t,\tau,T} \left( \{\pi_k^*\}_t^\tau \right) + \tilde{E}_{t,\tau,T}^W \cdot W_t + \tilde{E}_{t,\tau,T}^h \left( \{\pi_k^*\}_t^\tau \right) \cdot h_{t+1} \right\}.$$
 (C.7)

Again, for the sake of readability, we omit the dependencies of recursive parameters on the optimal solution except for definitions. Comparing (C.7) and (4.10), it is clear that

$$\tilde{D}_{t,t,T}\left(\pi_{t}^{*}\right) = \log\left(\bar{a}^{\bar{\gamma}}\right) - \bar{\gamma}P_{t,T}\left(\pi_{t}^{*}\right), \qquad (C.8a)$$

$$\tilde{E}_{t,t,T}^W = -\bar{\gamma}Q_{t,T},\tag{C.8b}$$

$$\tilde{E}_{t,t,T}^{h}\left(\pi_{t}^{*}\right) = -\bar{\gamma}R_{t,T}\left(\pi_{t}^{*}\right),\tag{C.8c}$$

for the first term corresponding to the utility derived from consumption at time t, and furthermore that

$$\tilde{D}_{t,T,T}(\pi_t^*) = \log\left(-\frac{\bar{\beta}^{T-t}}{\gamma}\right) + D_{t,T}(\pi_t^*) - E_{t,T}^W P_{t,T}(\pi_t^*), \qquad (C.9a)$$

$$\tilde{E}_{t,T,T}^{W} = E_{t,T}^{W} \cdot (1 - Q_{t,T}), \qquad (C.9b)$$

$$\tilde{E}_{t,T,T}^{h}(\pi_{t}^{*}) = E_{t,T}^{h}(\pi_{t}^{*}) - E_{t,T}^{W}R_{t,T}(\pi_{t}^{*}).$$
(C.9c)

For brevity, we now define, based on the coefficients A and B from (4.5):

$$A = A \left( -\bar{\gamma} Q_{\tau,T} \cdot \pi_{\tau-1}^*, -\bar{\gamma} R_{\tau,T}; \tau - 1, \tau \right),$$
  
$$\tilde{B} = B \left( -\bar{\gamma} Q_{\tau,T} \cdot \pi_{\tau-1}^*, -\bar{\gamma} R_{\tau,T}; \tau - 1, \tau \right).$$

With this notation, we now use the conditional bivariate MGF of  $(W_{\tau}, h_{\tau+1})$  from Corollary 4.6 to obtain the recursive parameters for general  $\tau \in \{t+1, \ldots, T-1\}$ :

$$\begin{split} \mathbb{E}_{\tau-1} \left[ \exp\left\{ -\bar{\gamma} \hat{C}_{\tau}^{*} \right\} \right] &= \mathbb{E}_{\tau-1} \left[ \exp\left\{ -\bar{\gamma} P_{\tau,T} - \bar{\gamma} Q_{\tau,T} W_{\tau} - \bar{\gamma} R_{\tau,T} h_{\tau+1} \right\} \right] \\ &= \exp\left\{ -\bar{\gamma} P_{\tau,T} \right\} \cdot \Psi_{(W_{\tau},h_{\tau+1})} \left( -\bar{\gamma} Q_{\tau,T} - \bar{\gamma} R_{\tau,T} \mid \mathcal{F}_{\tau-1} \right) \\ &= \exp\left\{ -\bar{\gamma} P_{\tau,T} + \tilde{A} - \left( 1 - \pi_{\tau-1}^{*} \right) \bar{\gamma} Q_{\tau,T} r - \bar{\gamma} Q_{\tau,T} \left( W_{\tau-1} - \hat{C}_{\tau-1}^{*} \right) \right. \\ &+ \left[ \tilde{B} - \frac{1}{2} \bar{\gamma} Q_{\tau,T} \left( \pi_{\tau-1}^{*} - \left( \pi_{\tau-1}^{*} \right)^{2} \right) \right] \cdot h_{\tau} \right\} \\ &= \exp\left\{ -\bar{\gamma} P_{\tau,T} + \tilde{A} - \left( 1 - \pi_{\tau-1}^{*} \right) \bar{\gamma} Q_{\tau,T} r + \bar{\gamma} Q_{\tau,T} P_{\tau-1,T} \right. \\ &- \bar{\gamma} Q_{\tau,T} \left( 1 - Q_{\tau-1,T} \right) W_{\tau-1} \\ &+ \left[ \tilde{B} - \frac{1}{2} \bar{\gamma} Q_{\tau,T} \left( \pi_{\tau-1}^{*} - \left( \pi_{\tau-1}^{*} \right)^{2} \right) + \bar{\gamma} Q_{\tau,T} R_{\tau-1,T} \right] \cdot h_{\tau} \right\}, \end{split}$$

leading to

$$\begin{split} \tilde{D}_{\tau-1,\tau,T}\left(\{\pi_{k}^{*}\}_{\tau-1}^{\tau}\right) &= \log\left(\bar{\beta}^{\tau-(\tau-1)}\bar{a}^{\bar{\gamma}}\right) - \bar{\gamma}P_{\tau,T}\left(\pi_{\tau}^{*}\right) + \tilde{A} \\ &- \left(1 - \pi_{\tau-1}^{*}\right)\bar{\gamma}Q_{\tau,T}r + \bar{\gamma}Q_{\tau,T}P_{\tau-1,T}\left(\pi_{\tau-1}^{*}\right), \\ \tilde{E}_{\tau-1,\tau,T}^{W} &= -\bar{\gamma}Q_{\tau,T}\left(1 - Q_{\tau-1,T}\right), \\ \tilde{E}_{\tau-1,\tau,T}^{h}\left(\{\pi_{k}^{*}\}_{\tau-1}^{\tau}\right) &= \tilde{B} - \frac{1}{2}\bar{\gamma}Q_{\tau,T}\left(\pi_{\tau-1}^{*} - \left(\pi_{\tau-1}^{*}\right)^{2}\right) + \bar{\gamma}Q_{\tau,T}R_{\tau-1,T}\left(\pi_{\tau-1}^{*}\right). \end{split}$$

In general, if  $t < \tau$ , we can use the conditional bivariate moment generating function of the log wealth and the conditional variance of the log-asset return from Corollary 4.6 again to

recursively calculate

$$\bar{\beta} \cdot \mathbb{E}_{t} \left[ \exp \left\{ \tilde{D}_{t+1,\tau,T} \left( \{\pi_{k}^{*}\}_{t+1}^{\tau} \right) + \tilde{E}_{t+1,\tau,T}^{W} W_{t+1} + \tilde{E}_{t+1,\tau,T}^{h} \left( \{\pi_{k}^{*}\}_{t+1}^{\tau} \right) h_{t+2} \right\} \right] \\ = \exp \left\{ \tilde{D}_{t,\tau,T} \left( \{\pi_{k}^{*}\}_{t}^{\tau} \right) + \tilde{E}_{t,\tau,T}^{W} \cdot W_{t} + \tilde{E}_{t,\tau,T}^{h} \left( \{\pi_{k}^{*}\}_{t}^{\tau} \right) \cdot h_{t+1} \right\},$$

with

$$\tilde{D}_{t,\tau,T}\left(\{\pi_{k}^{*}\}_{t}^{\tau}\right) = \log\left(\bar{\beta}\right) + \tilde{D}_{t+1,\tau,T}\left(\{\pi_{k}^{*}\}_{t+1}^{\tau}\right) + A\left(\tilde{E}_{t+1,\tau,T}^{W} \cdot \pi_{t}^{*}, \tilde{E}_{t+1,\tau,T}^{h}\left(\{\pi_{k}^{*}\}_{t}^{\tau}\right)\right) + (1 - \pi_{t}^{*})\tilde{E}_{t+1,\tau,T}^{W}r - \tilde{E}_{t+1,\tau,T}^{W}P_{t,T}\left(\pi_{t}^{*}\right),$$
(C.10a)

$$\tilde{E}_{t,\tau,T}^{W} = \tilde{E}_{t+1,\tau,T}^{W} \left( 1 - Q_{t,T} \right), \tag{C.10b}$$

$$\tilde{E}_{t,\tau,T}^{h}\left(\{\pi_{k}^{*}\}_{t}^{\tau}\right) = B\left(\tilde{E}_{t+1,\tau,T}^{W} \cdot \pi_{t}^{*}, \tilde{E}_{t+1,\tau,T}^{h}\left(\{\pi_{k}^{*}\}_{t+1}^{\tau}\right)\right) + \frac{1}{2}\tilde{E}_{t+1,\tau,T}^{W}\left(\pi_{t}^{*} - (\pi_{t}^{*})^{2}\right) - \tilde{E}_{t+1,\tau,T}^{W}R_{t,T}\left(\pi_{t}^{*}\right).$$
(C.10c)

We can continue repeating this last step until we have reached time point t to obtain an expression which is exponentially affine in  $W_t$  and  $h_{t+1}$ . Concerning the factors  $\tilde{E}^W_{t,\tau,T}$  for  $\tau \in \{t, \ldots, T\}$ , given by (C.8b), (C.9b) and (C.10b), we note that

$$\tilde{E}_{t,t,T}^{W} \stackrel{(C.8b)}{=} -\bar{\gamma}Q_{t,T} \stackrel{(C.6b)}{=} \bar{\gamma}\frac{E_{t,T}^{W}}{\bar{\gamma} + E_{t,T}^{\hat{C}}} = E_{t,T}^{W} \left(\frac{E_{t,T}^{\hat{C}} + \bar{\gamma} - E_{t,T}^{\hat{C}}}{\bar{\gamma} + E_{t,T}^{\hat{C}}}\right) \\
= E_{t,T}^{W} \left(1 - \frac{E_{t,T}^{\hat{C}}}{\bar{\gamma} + E_{t,T}^{\hat{C}}}\right) \stackrel{(C.6b)}{=} E_{t,T}^{W} \left(1 - Q_{t,T}\right) \stackrel{(C.9b)}{=} \tilde{E}_{t,T,T}^{W}.$$
(C.11)

This implies in particular that at time t = T - 1, both coefficients corresponding to the log wealth are equal. For t < T-1 and each  $\tau \in \{t+1, \ldots, T-1\}$ , we know that  $\tilde{E}^W_{t,\tau,T}$  is determined recursively, starting with  $\tilde{E}^W_{\tau,\tau,T}$  and then (repeatedly) using (C.10b). Thus, combining these steps leads to

$$\tilde{E}_{t,\tau,T}^{W} \stackrel{(C.10b)}{=} \tilde{E}_{t+1,\tau,T}^{W} (1 - Q_{t,T}) \\
\stackrel{(C.10b)}{=} \tilde{E}_{\tau,\tau,T}^{W} (1 - Q_{\tau-1,T}) \cdot \dots \cdot (1 - Q_{t+1,T}) (1 - Q_{t,T}) \\
\stackrel{(C.11)}{=} \tilde{E}_{\tau,T,T}^{W} \cdot (1 - Q_{\tau-1,T}) \cdot \dots \cdot (1 - Q_{t+1,T}) (1 - Q_{t,T}) \\
\stackrel{(C.9b)}{=} E_{\tau,T}^{W} (1 - Q_{\tau,T}) \cdot (1 - Q_{\tau-1,T}) \cdot \dots \cdot (1 - Q_{t+1,T}) (1 - Q_{t,T}) \\
\stackrel{(C.2c)}{=} E_{t,T}^{W} (1 - Q_{t,T}) \quad (1 - Q_{\tau-1,T}) \cdot \dots \cdot (1 - Q_{t+1,T}) (1 - Q_{t,T}) \\
\stackrel{(C.9b)}{=} \tilde{E}_{t,T}^{W} (1 - Q_{t,T}) \quad (C.12) \\
\stackrel{(C.9b)}{=} \tilde{E}_{t,T,T}^{W},$$

showing that all coefficients corresponding to the log wealth are equal. The same reasoning for  $\tilde{E}^{h}_{t,\tau,T}$  with  $\tau \in \{t, \ldots, T\}$ , given by (C.8c), (C.9c) and (C.10c) yields:

$$\tilde{E}_{t,t,T}^{h}(\pi_{t}^{*}) \stackrel{(C.8c)}{=} -\bar{\gamma}R_{t,T}(\pi_{t}^{*}) \stackrel{(C.6c)}{=} \bar{\gamma}\frac{E_{t,T}^{h}(\pi_{t}^{*})}{\bar{\gamma} + E_{t,T}^{\hat{C}}} = E_{t,T}^{h}(\pi_{t}^{*})\left(\frac{E_{t,T}^{\hat{C}} + \bar{\gamma} - E_{t,T}^{\hat{C}}}{\bar{\gamma} + E_{t,T}^{\hat{C}}}\right) \\
= E_{t,T}^{h}(\pi_{t}^{*})\left(1 - \frac{E_{t,T}^{\hat{C}}}{\bar{\gamma} + E_{t,T}^{\hat{C}}}\right) \stackrel{(C.6c)}{=} E_{t,T}^{h}(\pi_{t}^{*}) - E_{t,T}^{W}R_{t,T}(\pi_{t}^{*})$$

$$\stackrel{\text{(C.9c)}}{=} \tilde{E}^h_{t,T,T}\left(\pi^*_t\right). \tag{C.13}$$

Furthermore, using the next to last expression in (C.11) and (C.13), respectively, in (\*), we obtain

$$\tilde{E}_{\tau-1,\tau,T}^{h}\left(\{\pi_{k}^{*}\}_{\tau-1}^{\tau}\right) \stackrel{(C.10c)}{=} B\left(\tilde{E}_{\tau,\tau,T}^{W} \cdot \pi_{\tau-1}^{*}, \tilde{E}_{\tau,\tau,T}^{h}\left(\pi_{\tau}^{*}\right); \tau-1, \tau\right) \\
+ \frac{1}{2} \tilde{E}_{\tau,\tau,T}^{W}\left(\pi_{\tau-1}^{*} - \left(\pi_{\tau-1}^{*}\right)^{2}\right) - \tilde{E}_{\tau,\tau,T}^{W} R_{\tau-1,T}\left(\pi_{\tau-1}^{*}\right) \\
\stackrel{(*)}{=} B\left(E_{\tau,T}^{W}\left(1 - Q_{\tau,T}\right) \cdot \pi_{\tau-1}^{*}, E_{\tau,T}^{h}\left(\pi_{\tau}^{*}\right) - E_{\tau,T}^{W} R_{\tau,T}\left(\pi_{\tau}^{*}\right); \tau-1, \tau\right) \\
+ \frac{1}{2} E_{\tau,T}^{W}\left(1 - Q_{\tau,T}\right) \left(\pi_{\tau-1}^{*} - \left(\pi_{\tau-1}^{*}\right)^{2}\right) \\
- \tilde{E}_{\tau,\tau,T}^{W} R_{\tau-1,T}\left(\pi_{\tau-1}^{*}\right) \\
\stackrel{(C.2d)}{=} E_{\tau-1,T}^{h}\left(\pi_{\tau-1}^{*}\right) - \tilde{E}_{\tau,\tau,T}^{W} R_{\tau-1,T}\left(\pi_{\tau-1}^{*}\right) \\
\stackrel{(C.2c)}{=} E_{\tau-1,T}^{h}\left(\pi_{\tau-1}^{*}\right) - E_{\tau,T}^{W}\left(1 - Q_{\tau,T}\right) R_{\tau-1,T}\left(\pi_{\tau-1}^{*}\right) \\
\stackrel{(C.2c)}{=} \tilde{E}_{\tau-1,T}^{h}\left(\pi_{\tau-1}^{*}\right) - E_{\tau-1,T}^{W} R_{\tau-1,T}\left(\pi_{\tau-1}^{*}\right) \\
\stackrel{(C.2d)}{=} \tilde{E}_{\tau-1,T}^{h}\left(\pi_{\tau-1}^{*}\right) - E_{\tau-1,T}^{W} R_{\tau-1,T}\left(\pi_{\tau-1}^{*}\right) \\
\stackrel{(C.2d)}{=} \tilde{E}_{\tau-1,T}^{h}\left(\pi_{\tau-1}^{*}\right) - E_{\tau-1,T}^{W}\left(1 - Q_{\tau,T}\right) R_{\tau-1,T}\left(\pi_{\tau-1}^{*}\right) \\
\stackrel{(C.2d)}{=} \tilde{E}_{\tau-1,T}^{h}\left(\pi_{\tau-1}^{*}\right) - \tilde{E}_{\tau-1,T}^{W}\left(\pi_{\tau-1}^{*}\right) \\
\stackrel{(C.2d)}{=} \tilde{E}_{\tau-1,T}^{h}\left(\pi_{\tau-1}^{*}\right) - \tilde{E}_{\tau-1,T}^{W}\left(\pi_{\tau-1}^{*}\right) \\
\stackrel{(C.2d)}{=} \tilde{E}_{\tau-1,T}^{h}\left(\pi_{\tau-1}^{*}\right) - \tilde{E}_{\tau-1,T}^{W}\left(\pi_{\tau-1}^{*}\right) \\
\stackrel{(C.2d)}{=} \tilde{E}_{\tau-1,T}^{h}\left(\pi_{\tau-1}^{*}\right) \\
\stackrel{(C.2d)}{=} \tilde{E}_{\tau-1,\tau-1,T}^{h}\left(\pi_{\tau-1}^{*}\right) \\
\stackrel{(C.2d)}{=} \tilde{E}$$

Using this step (C.14), we can show by backward induction on t, for  $t \in \{1, \ldots, \tau\}$ , that  $\tilde{E}^{h}_{t,\tau,T}\left(\{\pi^*_k\}^{\tau}_t\right) = \tilde{E}^{h}_{t,t,T}\left(\pi^*_t\right)$ . In particular, using the induction hypothesis  $\tilde{E}^{h}_{t+1,\tau,T}\left(\{\pi^*_k\}^{\tau}_{t+1}\right) = \tilde{E}^{h}_{t+1,t+1,T}\left(\pi^*_{t+1}\right)$  in (\*') and proceeding as in (C.14), we arrive at

$$\begin{split} \tilde{E}_{t,\tau,T}^{h} \left( \left\{ \pi_{k}^{*} \right\}_{t}^{\tau} \right) \stackrel{(C.10c)}{=} B \left( \tilde{E}_{t+1,\tau,T}^{W} \cdot \pi_{t}^{*}, \tilde{E}_{t+1,\tau,T}^{h} \left( \left\{ \pi_{k}^{*} \right\}_{t+1}^{\tau} \right) \right) \\ &\quad + \frac{1}{2} \tilde{E}_{t+1,\tau,T}^{W} \left( \pi_{t}^{*} - \left( \pi_{t}^{*} \right)^{2} \right) - \tilde{E}_{t+1,\tau,T}^{W} R_{t,T} \left( \pi_{t}^{*} \right) \\ \stackrel{(*)}{=} B \left( \tilde{E}_{t+1,t+1,T}^{W} \cdot \pi_{t}^{*}, \tilde{E}_{t+1,t+1,T}^{h} \left( \pi_{t+1}^{*} \right) \right) \\ &\quad + \frac{1}{2} \tilde{E}_{t+1,t+1,T}^{W} \left( \pi_{t}^{*} - \left( \pi_{t}^{*} \right)^{2} \right) - \tilde{E}_{t+1,t+1,T}^{W} R_{t,T} \left( \pi_{t}^{*} \right) \\ \stackrel{(*)}{=} B \left( E_{t+1,t}^{W} \left( 1 - Q_{t+1,T} \right) \cdot \pi_{t}^{*}, E_{t+1,T}^{h} \left( \pi_{t+1}^{*} \right) - E_{t+1,t+1,T}^{W} R_{t+1,T} \left( \pi_{t}^{*} \right) \right) \\ &\quad + \frac{1}{2} E_{t+1,T}^{W} \left( 1 - Q_{t+1,T} \right) \left( \pi_{t}^{*} - \left( \pi_{t}^{*} \right)^{2} \right) - \tilde{E}_{t+1,t+1,T}^{W} R_{t,T} \left( \pi_{t}^{*} \right) \\ \stackrel{(C.2d)}{=} E_{t,T}^{h} \left( \pi_{t}^{*} \right) - \tilde{E}_{t+1,t+1,T}^{W} R_{t,T} \left( \pi_{t}^{*} \right) \\ \stackrel{(C.2d)}{=} E_{t,T}^{h} \left( \pi_{t}^{*} \right) - E_{t+1,T}^{W} \left( 1 - Q_{t+1,T} \right) R_{t,T} \left( \pi_{t}^{*} \right) \\ \stackrel{(C.2d)}{=} E_{t,T}^{h} \left( \pi_{t}^{*} \right) - E_{t+1,T}^{W} \left( 1 - Q_{t+1,T} \right) R_{t,T} \left( \pi_{t}^{*} \right) \\ \stackrel{(C.2d)}{=} E_{t,T}^{h} \left( \pi_{t}^{*} \right) - E_{t,T}^{W} R_{t,T} \left( \pi_{t}^{*} \right) \\ \stackrel{(C.2d)}{=} \tilde{E}_{t,T}^{h} \left( \pi_{t}^{*} \right) - \tilde{E}_{t,T}^{W} R_{t,T} \left( \pi_{t}^{*} \right) \\ \stackrel{(C.13)}{=} \tilde{E}_{t,t,T}^{h} \left( \pi_{t}^{*} \right) \right) \end{array}$$

implying that all coefficients corresponding to the conditional variance of the log-asset return are equal. Using (C.11), (C.12), (C.13) and (C.15) in (C.7) and re-ordering terms now yields (4.14).

Proof of Corollary 4.10. We show that in the limit for  $\bar{\gamma} \to \infty$ , all three coefficients  $P_{t,T}$ ,  $Q_{t,T}$ and  $R_{t,T}$  in (4.11) become zero for all  $t \in \{0, \ldots, T-1\}$  if  $\bar{a} = 1$ , resulting in  $\hat{C}_t = 0$ , which is equivalent to  $V_t = V_t - C_t$  and thus  $C_t = 0$ . In particular, take into account that  $P_{T,T} =$   $Q_{T,T} = R_{T,T} = 0$ . Now, fixing  $t \in \{0, ..., T-1\}$  and assuming that  $P_{s,T} = Q_{s,T} = R_{s,T} = 0$  for all  $s \in \{t + 1, ..., T\}$  in (C.2), we find that

$$D_{t,T}(\pi_t^*) = D_{t+1,T}(\pi_{t+1}^*) + E_{t+1,T}^W(1 - \pi_t^*)r + A\left(E_{t+1,T}^W\pi_t, E_{t+1,T}^h(\pi_{t+1}^*)\right), \quad (C.16a)$$

$$E_{t,T}^{U} = -E_{t+1,T}^{W} \cdot \mathbb{1}_{\{t < T\}},$$
(C.16b)

$$E_{t,T}^{w} = E_{t+1,T}^{w} = \gamma,$$
 (C.16c)

$$E_{t,T}^{h}\left(\pi_{t}^{*}\right) = \frac{1}{2} E_{t+1,T}^{W}\left(\pi_{t} - \pi_{t}^{2}\right) + B\left(E_{t+1,T}^{W}\pi_{t}^{*}, E_{t+1,T}^{h}\left(\pi_{t+1}^{*}\right)\right), \qquad (C.16d)$$

are independent of  $\bar{\gamma}$ . Checking (C.6) now reveals that  $Q_{t,T}$  and  $R_{t,T}$  approach zero as  $\bar{\gamma} \to \infty$ . Furthermore, we calculate for  $\gamma < 0$ , using the above independences and de l'Hôpital's rule in (\*):

$$\begin{split} \lim_{\bar{\gamma}\to\infty} P_{t,T}\left(\pi_t^*\right) &= \lim_{\bar{\gamma}\to\infty} \left\{ -\frac{1}{\bar{\gamma} + E_{t,T}^{\hat{C}}} \left( \log\left( -\frac{\bar{\beta}^{T-t} E_{t,T}^{\hat{C}}}{\gamma \bar{\gamma} \bar{a}^{\bar{\gamma}}} \right) + D_{t,T}\left(\pi_t^*\right) \right) \right\} \\ &= \lim_{\bar{\gamma}\to\infty} \left\{ -\frac{\log\left(\bar{\beta}^{T-t} E_{t,T}^{\hat{C}}\right) - \log\left(-\gamma \bar{\gamma} \bar{a}^{\bar{\gamma}}\right)}{\bar{\gamma} + E_{t,T}^{\hat{C}}} - \frac{D_{t,T}\left(\pi_t^*\right)}{\bar{\gamma} + E_{t,T}^{\hat{C}}} \right\} \\ &= \lim_{\bar{\gamma}\to\infty} \left\{ -\frac{\log\left(\bar{\beta}^{T-t} E_{t,T}^{\hat{C}}\right)}{\frac{\bar{\gamma} + E_{t,T}^{\hat{C}}}{-\gamma - 1}} + \frac{\log\left(-\gamma \bar{\gamma} \bar{a}^{\bar{\gamma}}\right)}{\bar{\gamma} + E_{t,T}^{\hat{C}}} - \frac{D_{t,T}\left(\pi_t^*\right)}{\frac{\bar{\gamma} + E_{t,T}^{\hat{C}}}{-\gamma - 1}} \right\} \\ &= 0 + \lim_{\bar{\gamma}\to\infty} \frac{\log\left(-\gamma \bar{\gamma} \bar{a}^{\bar{\gamma}}\right)}{\bar{\gamma} + E_{t,T}^{\hat{C}}} + 0 \\ &\stackrel{(*)}{=} \lim_{\bar{\gamma}\to\infty} \frac{\log\left(\bar{a}\right) + 1/\bar{\gamma}}{1} \\ &= \log\left(\bar{a}\right). \end{split}$$

This shows that the coefficient  $P_{t,T}$  approaches zero in the limit for  $\bar{\gamma} \to \infty$  if and only if  $\bar{a} = 1$  holds. Together with  $E_{T,T}^W = \gamma$ , having  $P_{t,T} = Q_{t,T} = R_{t,T} = 0$  for all  $t \in \{0, \ldots, T\}$  reproduces the formulas from Escobar-Anel et al. (2021) in (C.16) (i.e., in (C.2)) and (C.3).

Proof of Corollary 4.11. The formulas for the value function and optimal control on consumption follow directly from the fact that the HN-GARCH model is a special case of an affine GARCH model and that Theorem 4.4 can be applied. With A and B defined according to (4.19), we can easily check that (4.12a) is always satisfied. Using the abbreviation in (4.21), Equation (4.12b) requires

$$\begin{aligned} \lambda + \frac{2E_{t,T}^W \pi_t^* - 4v\alpha\rho}{2\left(1 - 2v\alpha\right)} &= \pi_t^* - \frac{1}{2} \\ \Leftrightarrow \qquad \frac{E_{t,T}^W \pi_t^*}{1 - 2v\alpha} - \pi_t^* &= \frac{2v\alpha\rho}{1 - 2v\alpha} - \lambda - \frac{1}{2} \\ \Leftrightarrow \qquad \frac{E_{t,T}^W - 1 + 2v\alpha}{1 - 2v\alpha} \cdot \pi_t^* &= \frac{2v\alpha\rho}{1 - 2v\alpha} - \lambda - \frac{1}{2} \\ \Leftrightarrow \qquad \pi_t^* &= \frac{1 - 2v\alpha}{E_{t,T}^W - 1 + 2v\alpha} \left(\frac{2v\alpha\rho}{1 - 2v\alpha} - \left(\lambda + \frac{1}{2}\right)\right) \end{aligned}$$

C Appendix for Chapter 4

$$\begin{aligned} \Leftrightarrow \qquad & \pi_t^* = \frac{2v\alpha\rho}{E_{t,T}^W - 1 + 2v\alpha} - \frac{\left(\lambda + \frac{1}{2}\right)\left(1 - 2v\alpha\right)}{E_{t,T}^W - 1 + 2v\alpha} \\ \Leftrightarrow \qquad & \pi_t^* = \frac{\left(\lambda + \frac{1}{2}\right)\left(1 - 2v\alpha\right) - 2v\alpha\rho}{1 - E_{t,T}^W - 2v\alpha}. \end{aligned}$$

Note that we already implicitly assumed that  $1 - 2v\alpha > 0$  for the representations of A and B in (4.19), see Appendix B.1. Furthermore, we showed in the proof of Theorem 4.4 that  $E_{t,T}^W < 0$  for all  $t = 0, \ldots, T$ . Together, this implies that the above representation of the optimal allocation is well-defined.

*Proof of Corollary 4.13.* Analogous to Corollary 4.11, we plug the expressions from (4.23) into (4.12). While (4.12a) is again trivially satisfied, (4.12b) with the abbreviation in (4.21) yields

$$\begin{split} \nu - \frac{1}{\eta^2} \left( \frac{\left(1 - 2va\eta^4\right) 2\eta}{2\sqrt{\left(1 - 2va\eta^4\right) \left(1 - 2E_{t,T}^W \pi_t^* \eta - 2vc\right)}} \right) &= \pi_t^* - \frac{1}{2} \\ \nu - \frac{\sqrt{1 - 2va\eta^4}}{\eta\sqrt{1 - 2E_{t,T}^W \pi_t^* \eta - 2vc}} &= \pi_t^* - \frac{1}{2}, \end{split}$$

where the conditions for the arguments of the squareroots have already been assumed to be positive for the representations of A and B in (4.23), see Appendix B.2.  $\Box$ 

### C.2 Complementary Material

 $\Leftrightarrow$ 

#### C.2.1 Additional Plots for Section 4.4.1: Sensitivity Analysis

In addition to the analysis in Section 4.4.1 regarding the sensitivity of the optimal solution to changes in parameter values, this section presents some additional insights, again using the set of maximum likelihood estimates (MLEs) from Table 4.1.



(a) Values of the optimal initial relative consumption dependent on the market price of risk.



Figure C.1: Values of the optimal initial relative consumption, dependent on the market price of risk (modified via  $\nu$ ) and on the parameter  $\bar{\beta}$  for intertemporal substitution. The time horizon is five years with 252 trading days each.

The first of these additional investigations targets changes in the market price of risk (MPR), which in the IG-GARCH model is given via  $\lambda = \nu + \eta^{-1}$ . Keeping the parameter  $\eta$  constant, we investigate sensitivity w.r.t. the market price of risk via changes in  $\nu$ . Figure C.1a shows that the initial relative consumption increases with the market price of risk, although the changes are relatively small with an order of  $1 \times 10^{-5}$  for the entire displayed range  $\lambda \in [0, 2.5]$  (with the original MPR for the set in Table 4.1 being 1.16). Another interesting analysis considers the factor  $\bar{\beta}$  for intertemporal substitution. As Figure C.1b shows, the initial consumption-wealth ratio increases as the value  $\bar{\beta}$  is decreased, i.e., as utilities at future time points are valued less.



(a) Values of the optimal initial relative consumption, dependent on w.

(b) Values of the optimal initial relative consumption, dependent on r.

Figure C.2: Values of the optimal initial relative consumption, dependent on the parameters w and r. The time horizon is five years with 252 trading days each.

Contrary to the optimal relative investment strategy  $\{\pi_t^*\}_t$ , the optimal consumption process does depend on the parameters w and r. In both cases, the level of relative consumption increases as the parameter increases. For w, which has an impact mainly on the conditional variance of the log-asset return and the long-term mean of this variance, we consider the positive part of the confidence interval for the MLE, since negative values of w may allow for negative values for the variance. Concerning the interest rate r, we consider the range [0.1%, 10%] of annualized interest rates.

We note that the initial level of relative consumption slightly increases as the parameters a, b and c of the underlying IG-GARCH model are increased within the corresponding confidence intervals reported with the MLEs presented in Table 4.1.

#### C.2.2 Alternative Parametric Choice

Complementary to the analysis in Section 4.4, we provide the central figures also for the MLEs by Christoffersen et al. (2006). As in Section 4.4, the analysis is based on the MLEs for the three different models – IG-GARCH, HN-GARCH and the homoscedastic variant – using the same underlying data set for all models. The parameter values for all three models can be found in Table C.1. Again, if not stated otherwise, we use the default investment parameters as described in Section 4.4.1.

Figure C.3a shows that the behavior of the optimal initial relative consumption  $\bar{C}_0^*$ , as the time horizon changes, is quite similar to what is observed for the first parameter set in Figure 4.2a. In particular, the value for initial relative consumption decreases as the length of the time horizon increases. Comparing Figure C.3b to 4.2b, however, shows a different picture in terms of the optimal risky allocation. While the special cases without consumption still lead to the smallest fractions of wealth invested in the risky asset, the IG-GARCH now yields the lowest

**Table C.1:** Values for the parameters used for the IG-GARCH, the HN-GARCH and the homoscedastic Gaussian model – consisting of the MLEs as of Christoffersen et al. (2006) for all parameters and models. The value of  $\nu$  for the IG-GARCH model implies that the market price of risk is  $\nu + \eta^{-1} \approx 2.150$ .

Panel A: IG-GARCH		Panel B: HN-GARCH		Panel C: Hom. Gauss.	
Param.	Values	Param.	Values	Param.	Values
ν	$1.625 \times 10^3$	$\lambda$	2.772	$\lambda$	3.106
η	$-6.162 \times 10^{-4}$	ω	$3.038\times10^{-9}$	ω	$9.520\times 10^{-5}$
w	$3.768 \times 10^{-10}$	eta	$9.026\times10^{-1}$		
a	$2.472\times 10^7$	α	$3.660\times 10^{-6}$		
b	$-1.933 \times 10^1$	ρ	$1.284 \times 10^2$		
С	$4.142 \times 10^{-6}$				

risky allocation among the strategies with consumption, followed by the HN-GARCH model. It is also interesting to observe that the numbers for  $\pi_0^*$  are significantly higher in general for this parameter set from Table C.1 than in the plot in Figure 4.2b created with the estimates from Table 4.1.

Given the similar behavior of the initial optimal relative consumption seen in Figure C.3a as compared to Figure 4.2a, it is not surprising to see that changes in risk preferences have a comparable impact on this value under the new parametric environment, too, see Figure C.4. As in Figure 4.3a, we use  $\gamma \in [-6, -0.1]$  and  $\bar{\gamma} \in [5 \times 10^2, 1 \times 10^4]$  to modify risk aversion for terminal wealth and consumption, respectively. If the risk aversion for utility derived from consumption is decreased, both relative consumption and the relative risky investment increase. Decreasing risk aversion regarding utility from terminal wealth leads to an increase in the risky allocation.

Figure C.5 once more shows that the impact of skewness and kurtosis on the optimal initial risky allocation and relative consumption, respectively, is relatively small. For this analysis, we again use the approach described in Section 4.4.1 and keep the first two moments of the IG-GARCH parameter set constant, while the parameter  $\eta$  is modified, to observe the isolated impact of skewness and excess kurtosis. In both cases, the optimal solution decreases as log asset-returns become more negatively skewed and leptokurtic, but the differences are very little. Again, as in Figure 4.4, we also plot dots for the optimal solutions to the original IG-GARCH parametrization as of Table C.1 as well as to the corresponding HN-GARCH and homoscedastic models with the same first two unconditional moments. Concerning the optimal initial risky allocation  $\pi_0^*$ , the homoscedastic solution suggests investing around 2% less in the risky asset than the heteroscedastic models, while the difference between HN-GARCH and IG-GARCH solution is much less. A similar pattern can be observed for the optimal initial relative consumption in the right plot.



(a) Optimal initial relative consumption, dependent on the time horizon.



(b) Optimal initial relative risky allocation, dependent on the time horizon.

Figure C.3: Values of the optimal initial strategy at time t = 0, dependent on the time horizon. The plots include an IG-GARCH, an HN-GARCH and a homoscedastic Gaussian, all fitted to the same data set. The solutions without consumption are obtained via setting  $\bar{\gamma} = 1 \times 10^8$  to approximate the case without consumption. One year is assumed to have 252 trading days. We used the parameter MLEs of Christoffersen et al. (2006).

#### C.2.3 Wealth-Equivalent Loss

This section presents an extra analysis concerning the wealth-equivalent loss (WEL) arising from subobtimal strategies. We follow the method described in Remark 4.9 to measure losses in closed form via the expected utility and iterated expectations. This study complements the real-world analysis presented in Section 4.4.2, where the outcome was calculated explicitly for a five-year time horizon based on S&P 500 index data.

For the following plots, we assume that log asset-returns actually evolve according to an IG-GARCH model, while the investor implements the strategy originating from an HN-GARCH model or a homoscedastic variant. The parameters are taken from Tables 4.1 (for Figures C.8, C.9, C.10) and C.1 (for Figures C.6 and C.7).

Figure C.6, based on the parameter set displayed in Table C.1, illustrates that losses are highest for investors with very low risk aversion. We observe this pattern both for HN-GARCH and homoscedastic investors, and for risk aversion concerning both terminal wealth and consumption. The heteroscedastic HN-GARCH model clearly outperforms the homoscedastic variant.



(a) Optimal initial relative consumption, dependent on the investor's risk aversion concerning utility from terminal wealth (left plot) and from consumption (right plot).



(b) Optimal initial relative risky allocation, dependent on the investor's risk aversion concerning utility from terminal wealth (left plot) and from consumption (right plot).

Figure C.4: Values of the optimal initial relative consumption and optimal initial risky allocation, dependent on relative risk aversion. Relative risk aversion for terminal wealth in the left plots is governed by the parameter  $\gamma$ , relative risk aversion for consumption in the right plots is modified via  $\bar{\gamma}$ , see Remark 4.2. The time horizon is five years.

Furthermore, a comparison of the loss levels in Figure C.6 shows that the attitude towards consumption has a higher impact than risk-aversion concerning the utility derived from terminal wealth. This results in losses up to 4% for a homoscedastic investor with low risk aversion for consumption over a five-year horizon.

Figure C.7a confirms that conditional skewness and excess kurtosis have minimal impact on the investment strategy's performance. For modifications in the parameter  $\eta$  of the IG-GARCH model, the remaining parameters are adjusted such that the first two moments are kept constant, and the effect of changes in skewness and kurtosis is isolated. As seen above, the HN-GARCH model outperforms the homoscedastic variant. For both models, however, even for rather extreme values for the third and fourth moments (see the corresponding scale in Figures 4.4 and C.5), the losses show little difference to the original values on the right end of the horizontal axis. For the plot in Figure C.7b, we adjust the market price of risk (MPR) in all models to the



Figure C.5: Values of the optimal initial risky allocation and relative consumption dependent on the parameter  $\eta$ . The time horizon is five years. The color bar on the right-hand side indicates the corresponding values for skewness and excess kurtosis of log asset-returns, the vertical dashed line marks the original MLE for  $\eta$  in the considered parameter set. For this plot, the first two moments are kept constant under changes in  $\eta$ .



(a) WEL dependent on risk aversion concerning utility from terminal wealth.

(b) WEL dependent on risk aversion concerning utility from consumption.

Figure C.6: WEL for the HN-GARCH and the homoscedastic model, dependent on risk aversion concerning utility from terminal wealth and from consumption. Relative risk aversion for terminal wealth in the left plot is governed by the parameter  $\gamma$ , and relative risk aversion for consumption in the right plot is modified via  $\bar{\gamma}$ , see Remark 4.2. The time horizon is five years. The parameters are given in Table C.1.

same value. This is implemented via changes in  $\nu$  for the IG-GARCH model and via changes in  $\lambda$  in the suboptimal models. The original value for the MPR in the IG-GARCH model with the parameters from Table C.1 is  $\lambda = \nu + \eta^{-1} \approx 2.15$ . The first finding resulting from this plot is that matching the MPR decreases the overall level of losses for both suboptimal models. Secondly, the losses for the homoscedastic model exceed the losses originating form the HN-GARCH model significantly, leading to a flat line for the heteroscedastic model in a joint plot (see Figure C.7b). The WEL increases as the MPR increases.

The observations from Figures C.6 and C.7 are confirmed when performing the same analysis with the parameter set presented in Table 4.1 (Christoffersen et al., 2006) instead of the one in Table C.1. In this case, however, the difference between the losses for the heteroscedastic HN-GARCH model and the homoscedastic variant is so large that plotting the losses for both models at once would result in a seemingly flat line close to zero for the heteroscedastic GARCH model.





(b) WEL dependent on the MPR. In this plot, all models are modified to share the same MPR.

Figure C.7: WEL for the HN-GARCH and the homoscedastic model, dependent on the parameters  $\eta$  and  $\nu$ . Modifications in  $\eta$  lead to different levels of conditional skewness and excess kurtosis, see Figures 4.4 and C.5. The parameter  $\nu$  is used to modify the MPR. The time horizon is five years. The original parameter set is given in Table C.1.

In general, Figure C.8 again verifies that losses increase as the investor's level of risk aversion decreases. This trend is observed for both models and for both types of risk aversion. Furthermore, the impact of risk aversion concerning consumption, governed via the parameter  $\bar{\gamma}$ , is larger than the effect changes in the level of risk aversion concerning utility from terminal wealth. Overall, however, the losses resulting from the parameter sets in Table 4.1 are smaller than the ones presented in Figures C.6 for the parameter sets in Table C.1. In particular, even for very low risk aversion concerning consumption, the WEL for the homoscedastic model does not exceed 1% for a five-year horizon.

The plots in Figure C.9 seem surprising at first since more negative skewness and more excess kurtosis, both induced by a decrease in  $\eta$ , lead to lower WEL levels for both the HN-GARCH and the homoscedastic model. A closer look at Figures 4.2 and 4.4, however, reveals that a decrease in  $\eta$  also decreases the optimal risky allocation (left plots in Figures 4.4 and C.5, respectively), and furthermore that the IG-GARCH yields the highest fraction of wealth invested in the risky assets (Figure 4.2b). That is, the changes induced by conditional skewness and excess kurtosis are eclipsed by the original pattern stemming from the differences in the first and second moments. As opposed to that, the parameter set from Table C.1 leads to the inverse pattern regarding the risky allocation (with the IG-GARCH yielding the lowest fraction invested in the risky asset, see Figure C.3b), and thus a decrease in  $\eta$  also leads to higher WEL levels for the HN-GARCH and the homoscedastic model. We also note that significant changes in the levels of conditional skewness and excess kurtosis do not lead to large differences in the corresponding WEL – the effect is practically negligible for both models.

As in Figure C.7b, we adjust the MPR in all three models to obtain the WEL values plotted in Figures C.10a and C.10b. Again, matching the market price of risk leads to lower WEL levels in general, with all values remaining far below one basis point. As observed previously, the level of losses increases as the MPR increases.

In summary, the key observation deduced from the plots in this section is that heteroscedasticity matters, which can be concluded from the fact that the HN-GARCH model outperforms the homoscedastic variant in all studies. An interpretation of the specific numbers quantifying the loss, and especially a comparison to the real-world study in Section 4.4.2, remains difficult since the approaches are fundamentally different.



(a) WEL for the HN-GARCH model, dependent on the investor's risk aversion concerning utility from terminal wealth.



(c) WEL for the HN-GARCH model, dependent on the investor's risk aversion concerning utility from consumption.



(b) WEL for the homoscedastic model, dependent on the investor's risk aversion concerning utility from terminal wealth.



(d) WEL for the homoscedastic model, dependent on the investor's risk aversion concerning utility from consumption.

Figure C.8: WEL for the HN-GARCH and the homoscedastic model, dependent on risk aversion concerning utility from terminal wealth and from consumption. Relative risk aversion for terminal wealth in the left plot is governed by the parameter  $\gamma$ , and relative risk aversion for consumption in the right plot is modified via  $\bar{\gamma}$ , see Remark 4.2. The time horizon is five years. The model parameters are given in Table 4.1.





(a) WEL for the HN-GARCH model, dependent on the parameter  $\eta$ .

(b) WEL for the homoscedastic model, dependent on the parameter  $\eta$ .

Figure C.9: WEL for the HN-GARCH and the homoscedastic model, dependent on the parameter  $\eta$ . Modifications in  $\eta$  lead to different levels of conditional skewness and excess kurtosis, see Figures 4.4 and C.5. The time horizon is five years. The original parameter set is given in Table 4.1.

#### C Appendix for Chapter 4





(b) WEL dependent on the MPR.

Figure C.10: WEL for the HN-GARCH and the homoscedastic model, dependent on the MPR. In this plot, all models are modified to share the same MPR. The time horizon is five years. The original parameter set is given in Table 4.1. The original MPR for the IG-GARCH model with the parameters as of Table 4.1 is 1.16.

# **D** Appendix for Chapter 6

## D.1 Proofs from Sections 6.2 and 6.3

Proof of Prop 6.1. According to the model setup presented in Section 6.2, the innovations  $z_{t+1}$  and  $y_{t+1}$  are contemporaneously independent. Furthermore, for  $z_{t+1} \sim \mathcal{N}(0, h_{z,t+1})$ , integrating by substitution and completing the square shows that (Badescu et al., 2019, p. 33)

$$\mathbb{E}_t \left[ \exp\left\{ \alpha z_{t+1} + \beta z_{t+1}^2 \right\} \right] = \exp\left\{ \frac{\alpha^2 h_{z,t+1}}{2\left(1 - 2\beta h_{z,t+1}\right)} - \frac{1}{2} \log\left(1 - 2\beta h_{z,t+1}\right) \right\}.$$
 (D.1)

Using (D.1) in the last step, we prove Proposition 6.1:

$$\begin{split} \Psi_{(R_{t+1},h_{z,t+2},h_{y,t+2})} &(u, v_z, v_y \mid \mathcal{F}_t) \\ &= \mathbb{E}_t \left[ \exp \left\{ u \cdot R_{t+1} + v_z \cdot h_{z,t+2} + v_y \cdot h_{y,t+2} \right\} \right] \\ &= \mathbb{E}_t \left[ \exp \left\{ u \left[ r_{t+1} + (\lambda_z - \xi_z(1)) h_{z,t+1} + (\lambda_y - \xi_y(1)) h_{y,t+1} + z_{t+1} + y_{t+1} + v_z \cdot h_{z,t+2} + v_y \cdot h_{y,t+2} \right\} \right] \\ &= \exp \left\{ u \left[ r_{t+1} + (\lambda_z - \xi_z(1)) h_{z,t+1} + (\lambda_y - \xi_y(1)) h_{y,t+1} \right] \right\} \\ &\times \mathbb{E}_t \left[ \exp \left\{ u \cdot (z_{t+1} + y_{t+1}) + v_z \left[ \omega_z + b_z h_{z,t+1} + \frac{a_z}{h_{z,t+1}} (z_{t+1} - c_z h_{z,t+1})^2 \right] + v_y \left[ \omega_y + b_y h_{y,t+1} + \frac{a_y}{h_{z,t+1}} (z_{t+1} - c_y h_{z,t+1})^2 \right] \right\} \right] \\ &= \exp \left\{ u \cdot r_{t+1} + v_z \cdot \omega_z + v_y \cdot \omega_y + \left[ u \cdot (\lambda_z - \xi_z(1)) + v_z \cdot (b_z + a_z c_z^2) + v_y a_y c_y^2 \right] \cdot h_{z,t+1} + \left[ u \cdot (\lambda_y - \xi_y(1)) + v_y \cdot b_y \right] \cdot h_{y,t+1} \right\} \\ &\times \mathbb{E}_t \left[ \exp \left\{ u \cdot y_{t+1} + [u - 2v_z a_z c_z - 2v_y a_y c_y] \cdot z_{t+1} + \left[ v_z \cdot \frac{a_z}{h_{z,t+1}} + v_y \cdot \frac{a_y}{h_{z,t+1}} \right] \cdot z_{t+1}^2 \right\} \right] \end{split}$$

$$= \exp\left\{ ur_{t+1} + v_z \omega_z + v_y \omega_y + \left[ u \cdot (\lambda_z - \xi_z(1)) + v_z \cdot (b_z + a_z c_z^2) + v_y a_y c_y^2 \right] \cdot h_{z,t+1} + \left[ u \cdot (\lambda_y - \xi_y(1)) + v_y b_y + \xi_y(u) \right] \cdot h_{y,t+1} + \frac{(u - 2v_z a_z c_z - 2v_y a_y c_y)^2}{2 \left[ 1 - 2 \left( v_z a_z + v_y a_y \right) \right]} \cdot h_{z,t+1} - \frac{1}{2} \log \left( 1 - 2 \left( v_z a_z + v_y a_y \right) \right) \right\},$$

justifying (6.6) together with the definitions in (6.7). This completes the proof of Proposition 6.1.  $\hfill \Box$ 

Proof of Theorem 6.2. We use Bellman's principle in order to maximize via backwards induction and use the optimal solution at time t + 1 for our approach at time t. This results in the calculations below, where for the terminal step at time t = T - 1 we just work with

$$\phi_T(W_T) = U(V_T) = \frac{1}{\gamma} \exp\{\gamma W_T\},\$$

resulting in  $D_{T,T} = E_{T,T}^z = E_{T,T}^y = 0$ . We obtain:

$$\begin{split} \max_{\pi_{t}} & \mathbb{E}_{t} \left[ \phi_{t+1} \left( W_{t+1}, h_{z,t+2}, h_{y,t+2} \right) \right] \\ = & \max_{\pi_{t}} \mathbb{E}_{t} \left[ \frac{1}{\gamma} \exp \left\{ D_{t+1,T}(\pi_{t+1}^{*}) + \gamma W_{t+1} \right. \\ & + E_{t+1,T}^{z}(\pi_{t+1}^{*}) + h_{z,t+2} + E_{t+1,T}^{y}(\pi_{t+1}^{*}) + h_{y,t+2} \right\} \right] \\ = & \max_{\pi_{t}} \frac{1}{\gamma} \exp \left\{ D_{t+1,T}(\pi_{t+1}^{*}) \right\} \\ & \times \mathbb{E}_{t} \left[ \exp \left\{ \gamma \cdot \left[ W_{t} + \pi_{t} \cdot R_{t+1} \right. \\ & + \frac{1}{2} \left( \pi_{t} - \pi_{t}^{2} \right) \cdot \left( h_{z,t+1} + h_{y,t+1} \cdot \xi_{y}^{y}(0) \right) + (1 - \pi_{t}) r_{t+1} \right] \right. \\ & \left. + E_{t+1,T}^{z}(\pi_{t+1}^{*}) + h_{z,t+2} + E_{t+1,T}^{y}(\pi_{t+1}^{*}) + h_{y,t+2} \right\} \right] \\ = & \max_{\pi_{t}} \frac{1}{\gamma} \exp \left\{ D_{t+1,T}(\pi_{t+1}^{*}) + (1 - \pi_{t}) \gamma r_{t+1} + \gamma W_{t} \right. \\ & \left. + \frac{\gamma}{2} \left( \pi_{t} - \pi_{t}^{2} \right) \cdot \left( h_{z,t+1} + h_{y,t+1} \cdot \xi_{y}^{y}(0) \right) \right\} \\ \times & \mathbb{E}_{t} \left[ \exp \left\{ \gamma \pi_{t} R_{t+1} + E_{t+1,T}^{z}(\pi_{t+1}^{*}) + h_{z,t+2} + E_{t+1,T}^{y}(\pi_{t+1}^{*}) + h_{y,t+2} \right\} \right] \\ = & \max_{\pi_{t}} \frac{1}{\gamma} \exp \left\{ D_{t+1,T}(\pi_{t+1}^{*}) + (1 - \pi_{t}) \gamma r_{t+1} + \gamma W_{t} \right. \\ & \left. + \frac{\gamma}{2} \left( \pi_{t} - \pi_{t}^{2} \right) \cdot \left( h_{z,t+1} + h_{y,t+1} \cdot \xi_{y}^{y}(0) \right) \right\} \\ \times & \mathbb{V}_{(R_{t+1}, h_{z,t+2}, h_{y,t+2})} \left( \gamma \pi_{t}; E_{t+1,T}^{z}(\pi_{t+1}^{*}), E_{t+1,T}^{y}(\pi_{t+1}^{*}) \right| \mathcal{F}_{t} \right) \\ = & \max_{\pi_{t}} \frac{1}{\gamma} \exp \left\{ D_{t,T}(\pi_{t}) + \gamma W_{t} + E_{t,T}^{z}(\pi_{t}) \cdot h_{z,t+1} + E_{t,T}^{y}(\pi_{t}) \cdot h_{y,t+1} \right\}, \quad (D.2)$$

where we used the multivariate generating function from the proof of Proposition 6.1 in the Lévy GARCH setting and the definitions from Equations (6.9) in the last step. We seek to

maximize (D.2) with respect to the single variable  $\pi_t$ . Thus, we approach this step by taking derivatives, setting the first derivative to zero while having a negative second derivative at this point (or globally in case of concave functions). Concerning the first of the above-mentioned steps, note that, given some  $\gamma \neq 0$ , the first derivative is equal to zero if and only if

$$D'_{t,T}(\pi_t) + \left(E^z_{t,T}\right)'(\pi_t) \cdot h_{z,t+1} + \left(E^y_{t,T}\right)'(\pi_t) \cdot h_{y,t+1} = 0, \tag{D.3}$$

which, according to (6.9), is equivalent to

$$-\gamma r_{t+1} + \frac{\partial}{\partial u} A\left(\gamma \pi_t, E_{t+1,T}^{z,*}, E_{t+1,T}^{y,*}; t, t+1\right) \cdot \gamma + \left[\frac{\gamma}{2} - \gamma \pi_t + \frac{\partial}{\partial u} B\left(\gamma \pi_t, E_{t+1,T}^{z,*}, E_{t+1,T}^{y,*}; t, t+1\right) \cdot \gamma\right] \cdot h_{z,t+1}$$
(D.4)
$$+ \left[\xi_y''(0) \cdot \left(\frac{\gamma}{2} - \gamma \pi_t\right) + \frac{\partial}{\partial u} C\left(\gamma \pi_t, E_{t+1,T}^{y,*}; t, t+1\right) \cdot \gamma\right] \cdot h_{y,t+1} = 0,$$

where  $E_{t+1,T}^{x,*}$  is short for  $E_{t+1,T}^x(\pi_{t+1}^*)$  with  $x \in \{y, z\}$ . Now, using Definitions (6.7), we note that the first line reduces to zero, and, also dividing by  $\gamma$ , we arrive at

$$\begin{bmatrix} \frac{1}{2} - \pi_t + \lambda_z - \xi_z(1) + \frac{\gamma \pi_t - 2E_{t+1,T}^{z,*} a_z c_z - 2E_{t+1,T}^{y,*} a_y c_y}{1 - 2E_{t+1,T}^{z,*} a_z - 2E_{t+1,T}^{y,*} a_y} \end{bmatrix} \cdot h_{z,t+1} + \begin{bmatrix} \xi_y''(0) \left(\frac{1}{2} - \pi_t\right) + (\lambda_y - \xi_y(1)) + \xi_y'(\gamma \pi_t) \end{bmatrix} \cdot h_{y,t+1} = 0$$

Exploiting the Gaussian distribution of z, we further use  $\xi_z(1) = 1/2$  to obtain

$$\begin{bmatrix} \lambda_z - \pi_t + \frac{\gamma \pi_t - 2E_{t+1,T}^{z,*} a_z c_z - 2E_{t+1,T}^{y,*} a_y c_y}{1 - 2E_{t+1,T}^{z,*} a_z - 2E_{t+1,T}^{y,*} a_y} \end{bmatrix} \cdot h_{z,t+1} + \begin{bmatrix} \xi_y''(0) \left(\frac{1}{2} - \pi_t\right) + (\lambda_y - \xi_y(1)) + \xi_y'(\gamma \pi_t) \end{bmatrix} \cdot h_{y,t+1} = 0.$$
(D.5)

Remember that the two time-homogeneous parameters are assumed to be linearly dependent, i.e.,

$$h_{y,t+1} = \Lambda \cdot h_{z,t+1} \tag{D.6}$$

holds for some  $\Lambda \in \mathbb{R}$ . This yields the single equation

$$\left[\lambda_{z} - \pi_{t} + \frac{\gamma \pi_{t} - 2E_{t+1,T}^{z,*} a_{z}c_{z} - 2E_{t+1,T}^{y,*} a_{y}c_{y}}{1 - 2E_{t+1,T}^{z,*} a_{z} - 2E_{t+1,T}^{y,*} a_{y}}\right] + \Lambda \cdot \left[\xi_{y}''(0)\left(\frac{1}{2} - \pi_{t}\right) + (\lambda_{y} - \xi_{y}(1)) + \xi_{y}'(\gamma \pi_{t})\right] = 0.$$
(D.7)

How easily the resulting equation can be solved for  $\pi_t$  now largely depends on the shape of  $\xi_y$ , i.e. on the explicit distribution of the Lévy jump innovation y. In order to ensure that the solution to (D.7) is actually a maximum, we require (6.11). This concludes the proof.

Proof of Corollary 6.4. If we require (6.18) with strict inequality, then the continuous function on the left-hand side in the optimality equation (6.20) is strictly decreasing in  $\pi_t$ . Furthermore, using that  $\gamma < 0$ , we obtain that the left-hand side goes to  $\pm \infty$  if  $\pi_t \to \mp \infty$ . Using the intermediate value theorem, the existence of a unique solution follows.